FRANCHISE DISCLOSURE DOCUMENT



ALL AMERICAN PET RESORTS, LLC a Michigan limited liability company 41850 West Eleven Mile Road Suite 202 Novi, MI, 48375 (248) 513-3006 AllAmericanPetResorts.com

As a franchisee, you will operate a business offering professional pet boarding, daycare, and grooming services.

The total investment necessary to begin operation of an All American Pet Resorts® Business is \$687,500-\$1,509,500. This includes \$69,500 that must be paid to the franchisor or its affiliate for the initial franchise fee and \$20,000 which must be paid to franchisor or its affiliate for the real estate and resort coordination fee.

The disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar days before you sign a binding agreement with, or make any payment to, us or our Affiliates in connection with the franchise sale or grant. **Note, however, that no governmental agency has verified the information contained in this document.**

You may wish to receive your disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact the Chief Operating Officer at All American Pet Resorts, LLC, at 41850 West Eleven Mile Road, Suite 202, Novi, MI, 48375, (248) 513-3006 or stephan@aaprllc.com.

The terms of your contract will govern your franchise relationship. Do not rely on the disclosure document alone to understand your contract. Read your entire contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or an accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: April 30, 2024

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibit C.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit A includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only All American Pet Resorts® business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be an All American Pet Resorts® franchisee?	Item 20 or Exhibit C lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees.</u> You may have to pay royalties and other fees even if you are losing money.

Business model can change. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions.</u> You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions.</u> The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor.</u> Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends.</u> The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit E.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About *This* Franchise

Certain states require that the following risk(s) be highlighted:

- 1. Out of State Dispute Resolution. The franchise agreement requires you to resolve disputes with the franchisor by arbitration and/or litigation only in Michigan. Out-of-state arbitration or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to arbitrate or litigate with the franchisor in Michigan than in your own state.
- 2. **Spousal Liability.** Your spouse must sign a document that makes your spouse liable for all financial obligations under the franchise agreement even though your spouse has no ownership interest in the franchise. The guarantee will place your and your spouse's marital and personal assets, perhaps including your house, at risk if your franchise fails.
- 3. **Estimated Initial Investment.** The franchisee will be required to make an estimated initial investment ranging from \$687,500-\$1,509,500. This amount exceeds the franchisor's stockholders' equity as of December 31, 2023, which is \$147,889.
- 4. <u>Mandatory Minimum Payments.</u> You must make minimum advertising payments regardless of your sales levels. Your inability to make the payments may result in termination of your franchise and loss of your investment.

ADDITIONAL DISCLOSURES FOR THE STATE OF MICHIGAN

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

Each of the following provisions is void and unenforceable if contained in any documents relating to a franchise:

- 1. A prohibition on the right to join an association of franchisees.
- 2. A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- 3. A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than thirty (30) days, to cure such failure.
- 4. A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if:

 (i) the term of the franchise is less than 5 years old and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least 6 months advance notice of the franchisor's intent to renew the franchise.
- 5. A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- 6. A provision requiring that arbitration or litigation is to be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of the arbitration, to conduct arbitration at a location outside this state.
- 7. A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchases the franchise. Good cause shall include, but is not limited to:

V

a) The failure of the proposed transferee to meet the franchisor's then current reasonable qualifications or standards.

- b) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.
- c) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- d) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- 8. A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in the subdivision (c).
- 9. A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

If the franchisor's most recent financial statements are unaudited, and show a net worth of less than \$100,000.00, the franchisee may request the franchisor to arrange for the escrow of initial investment and other funds paid by the franchisee until the obligations, if any, of the franchisor to provide real estate, improvements, equipment, inventory, training or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENFORCEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to:

State of Michigan
Consumer Protection Division
Attn: Franchise Administrator
670 G. Mennen Williams Building
525 West Ottawa Street
Lansing, Michigan 48909
Telephone Number: (517) 373-7117

Michigan Effective Date: August 11, 2023

TABLE OF CONTENTS

	PAGE
CHISOR AND ANY PARENTS, PREDECESSORS AND AFFI	LIATES . 8
EXPERIENCE	9
N	10
CY	11
ES	11
S	12
O INITIAL INVESTMENT	17
ONS ON SOURCES OF PRODUCTS AND SERVICES	20
EE'S OBLIGATIONS	23
G	24
SOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEM	
E BUSINESS	
IONS ON WHAT THE FRANCHISEE MAY SELL	41
, TERMINATION, TRANSFER AND DISPUTE RESOLUTIO	N 41
GURES	43
L PERFORMANCE REPRESENTATIONS	43
AND FRANCHISEE INFORMATION	57
L STATEMENTS	60
TS	60
	60
Financial Statements Franchise Agreement List of Franchisees Trademark-Specific Franchisee Organizations List of State Administrators and Agents for Service of Process State-Specific Addenda to Disclosure Document, Franchise Agrand Area Development Agreement Non-Disclosure and Non-Competition Agreement Non-Disclosure and Non-Competition Agreement (Designated Manager) Statement of Franchisee Operations Manual Table of Contents Application and Confidentiality Agreement Form Release of Claims	
	EXPERIENCE N

AAPR FDD 2024

ITEM 1 THE FRANCHISOR AND ANY PARENTS, PREDECESSORS AND AFFILIATES

To simplify the language in this Franchise Disclosure Document, "All American Pet Resorts®" or "we" means All American Pet Resorts, LLC, the "Franchisor" and "You" means the person, corporation, partnership or other business entity that buys the franchise, the "Franchisee". If you are a business entity, "You" includes your owners and partners.

The Franchisor, Predecessor and Affiliates

All American Pet Resorts, LLC is a Michigan limited liability company formed on November 4, 2005. We do not do business under any other name. Our principal business address is 41850 West Eleven Mile Road, Suite 202, Novi, MI, 48375. Our fiscal year ends on December 31. Our agents for service of process in Michigan is Arthur K. Rimbold. Our agents for service of process in other states are disclosed in Exhibit E. We have offered franchises since 2005. We do not do business under any name other than "All American Pet Resorts, LLC", nor have we offered franchises in other lines of business.

We do not have any predecessors or parents as of the date of this disclosure document, but we do have one affiliate. Our affiliate is Rimbold Enterprises, Inc. d/b/a All American Pet Resorts[®] Royal Oak, a Michigan corporation, formerly doing business as Pet Ritz America that was incorporated on May 4, 2005 ("All American Pet Resorts[®] Royal Oak"). The principal business address for All American Pet Resorts[®] Royal Oak is 4227 Delemere Blvd., Royal Oak, MI 48073. All American Pet Resorts[®] Royal Oak operates an All American Pet Resorts[®] Business similar to the one described in this FDD since 1983 and assists with providing training to our franchisees. Our affiliate does not do business under any other name.

The Business

We offer franchises for the use of our "All American Pet Resorts®" trademarks, trade names, service marks and logos ("All American Pet Resorts® Marks" or "Marks") for the operation of All American Pet Resorts® Businesses, which will provide services for boarding, daycare and grooming of pets and the sale of related products. We do not operate an All American Pet Resorts® Business. We do not operate any business other than offering franchises under the All American Pet Resorts® Marks. The franchise is operated under a business format using a unique system, including our valuable know-how, information, trade secrets, training methods, Operations Manual, standards, designs, methods of trademark usage, copyrights, sources and specifications, confidential electronic and other communications, methods of Internet usage, marketing programs, and research and development connected with the operation and promotion of All American Pet Resorts® Businesses ("System"). We reserve the right to change or otherwise modify the System at any time. Each All American Pet Resorts® Business will offer professional pet boarding, daycare and grooming services.

You must sign our standard franchise agreement ("**Franchise Agreement**") and must operate your All American Pet Resorts[®] Business using our standard business operating practices. Your All American Pet Resorts[®] Business must offer authorized services and products, including pet boarding and daycare. We reserve the right to add, modify, or delete any services or products that you must offer or sell at your All American Pet Resorts[®] Business at any time as we deem

appropriate. We will gather market research and solicit Franchisee owner input in the evaluation of changes in any services or products that you must offer or sell at your All American Pet Resorts[®] Business. You must also obtain all necessary permits, licenses and approvals to operate your All American Pet Resorts[®] Business.

We make our offer to prospective franchisees using a single-unit Franchise Agreement in the form attached as Exhibit B to this Disclosure Document. You must sign a separate Franchise Agreement for each All American Pet Resorts[®] Business that you operate. Unless otherwise stated, any reference in this Disclosure Document to "you" or "Franchisee" includes you as a franchisee as well as any guarantors of franchisee's obligations under a Franchise Agreement.

Regulations

Certain states and local jurisdictions may have enacted laws, rules, regulations, and ordinances that apply to animal care in general or specifically to animal boarding, daycare and/or grooming. These regulations may establish certain standards, specifications, and requirements that must be followed by you. You should investigate whether there are any regulations and requirements that may apply in the geographic area in which you are interested in locating your Franchise and you should consider both their effect and the cost of compliance. You are responsible for obtaining all required licenses and permits and ensuring that your employees and others providing All American Pet Resorts® Products and Services to customers on behalf of your All American Pet Resorts® Business have all required licenses and permits. You should investigate whether there are any specific employment or data privacy regulations and requirements that may apply in the geographic area in which you are interested in locating your Franchise as these laws may vary from state to state and may impose additional requirements regarding your relationships with your employees and/or customers.

Market Competition

The All American Pet Resorts® System presently focuses on serving pet owners in urban and suburban areas. You may have to compete with other businesses including franchised operations, national chains and independently owned companies offering pet boarding, daycare and grooming, including animal training companies, animal obedience companies, grooming and pet companies, animal boarding kennels, regional and national pet store chains, and some veterinary offices that provide pet boarding, daycare and grooming. The market for animal boarding, daycare and grooming services is competitive.

ITEM 2 BUSINESS EXPERIENCE

Managing Member & CEO: Arthur Rimbold Founder, Owner, Operator

Mr. Rimbold is our Managing Member since our formation in November 2005 and our CEO since December 2015 in Novi, Michigan. Mr. Rimbold also has served as our President since January 2013 and from May 2005 to April 2011. He has also served as the CEO and President of All American Pet Resorts® Royal Oak since its formation in May 2005 in Royal Oak, Michigan.

Member: Yolanda Rimbold. Founder, Owner, Operator

Mrs. Rimbold has been a Member since our formation in November 2005 in Novi, Michigan. Mrs. Rimbold has also been the Vice President and Owner/Operator of All American Pet Resorts® Royal Oak since its formation in May 2005 in Royal Oak, Michigan.

Chief Operating Officer: Stephan Dimitroff

Mr. Dimitroff has been our Chief Operating Officer (formerly titled Executive Vice President) since December 2017 in Novi, Michigan. Previously, he was a Senior Account Manager for Omron Automotive Electronics from August 2011 until December 2017 in Novi, Michigan.

Director of Operations: Kevin Casida

Mr. Casida has been our Director of Operations since September 2021 in Novi, MI. Previously, from May 2017 to May 2020, Mr. Casida worked with GP Strategies Corporation in Troy, Michigan as the Global Training and Operations Manager on the QuickLane Tire and Auto project. From June 2020 to September 2021, Mr. Casida worked with MSX Intl in Southfield, Michigan as the Quick Lane Express Training Manager.

Digital Marketing Manager-Consultant: Laura Mikulski

Ms. Mikulski has been our Digital Marketing Manager since April 2021. Ms. Mikulski is located in Ferndale, Michigan. From January 2023 to present, Ms. Mikulski has worked as a Digital Strategist for Practis in Charlotte, North Carolina. From October 2021 to January 2023, Ms. Mikulski worked as a Marketing Manager for Goldstein Group Communications in Solon, OH on accounts with budgets from 100-500K. From January 2011 to October 2021, Ms. Mikulski worked as a Marketing Manager for Spotlight Grabber in Troy, Michigan on Dogtopia accounts locates in the Midwest and on the East coast. Her efforts were focused on improving digital presence, creating referral strategies, implementing store pre-launch strategies, and increasing client volume and loyalty.

Franchise Brand Marketing & Operations - Consultant Skip Starr

Mr. Starr is President of SC Brand Consultants in Grosse Pointe Park, Michigan and has been since January 2018. Mr. Starr joined our team in 2023 and has over 30 years in all areas of franchising with a dedicated focus on marketing. Mr. Starr has worked with several national franchise brands and their franchisees both as a consultant and also on the corporate side as a franchisor.

ITEM 3 LITIGATION

Pending Prior Litigation

Kathleen and Richard Kresge v. All American Pet Resorts, LLC and Arthur K. Rimbold, Case No. 18-301109-CB, in the 44th Circuit Court of Michigan, Civil Division, filed on November 10, 2018. Plaintiffs asserted claims against us and our President for alleged failure to perform corporate

duties in good faith and engaging in unfair and oppressive conduct as it relates to Ms. Kresge's membership interest. Plaintiffs sought a declaration of their rights and remedies, dissolution of the company or purchase of Ms. Kresge's interest, more than \$25,000 in damages, and recovery of their attorney's fees. We filed counterclaims against the plaintiffs, asserting that they breached their fiduciary duties to the company, tortiously interfered with the Franchisor's business relationships, and improperly received money that should be paid to the Franchisor. The parties entered into a settlement agreement whereby Franchisor purchased Richard Kresge's ownership interest in the Franchisor, AAPR withdrew all default notices previously issued to the Plaintiffs' companies, the Plaintiffs withdrew the complaint they had filed with the Michigan Attorney General relating to same allegations in this litigation, the parties updated the franchise agreements for the Plaintiffs' respective locations, and the parties agreed to a dismissal of all claims with prejudice. The case has been dismissed with prejudice. The Plaintiffs remain All American Pet Resorts® franchisees and continue to operate both All American Pet Resorts® Business locations as part of the System.

Other than this action, no litigation is required to be disclosed in this Item.

ITEM 4 BANKRUPTCY

No bankruptcy actions are required to be disclosed in this Item.

ITEM 5 INITIAL FEES

<u>Initial Franchise Fee</u>

The Initial Franchise Fee for your first All American Pet Resorts[®] Business is \$69,500 and is paid to us on the date when you sign a Franchise Agreement. The Initial Franchise Fee is deemed fully earned by us once paid and is non-refundable.

The Initial Franchisee Fee for your second and any additional All American Pet Resorts[®] Businesses is \$50,000 for each All American Pet Resorts[®] Business and is paid to us on the date when you sign a second or additional Franchise Agreement. The Initial Franchise Fee is deemed fully earned by us once paid and is non-refundable.

If you are a qualified United States Veteran, or active military member, we will discount the initial franchise fee for your All American Pet Resorts[®] Business(es) by 10% on your first All American Pet Resorts Business.

During our last fiscal year ending December 31, 2023, we did not collect any non-uniform Initial Franchise Fees.

Real Estate & Resort Coordination Fee

The Real Estate & Resort Coordination Fee is \$20,000 and is paid to us fifteen (15) days after you sign a Franchise Agreement. This fee is due per each location you open. This fee for additional locations is due upon signing Franchise Agreement for additional location. This fee is

deemed fully earned by us once paid and is nonrefundable. During our last fiscal year ending December 31, 2022, we did not collect a Real Estate & Resort Coordination Fee.

ITEM 6 OTHER FEES

Type of Fee	Amount	Due Date	Remarks
Royalty ^{(1)†}	7% of Gross Revenues	Payable monthly via electronic funds transfer on or before the 10th day of each month during the Operating Term (as defined in Item 11)	Gross Revenues are all revenue generated by the franchise operation. Payments are due after the first month of operations.
Local Advertising Expense ^{(2)†}	The greater of 1% of Gross Revenues ⁽¹⁾ per month or \$1,000 per month	Required fees calculated on a monthly basis and payment due as required by third party vendors	Expense after the first 90 days of operation for local advertising.
Brand Development and/or Promotions Fee ^{(3)†}	2% of Gross Revenues ⁽¹⁾ per month	Payable monthly via electronic funds transfer on or before the 10 th day of each month during the Operating Term	We have the right to increase the Brand Development and/or Promotions Fee up to 3% if it is approved by the Franchisee Advisory Council ("FAC"), should an FAC be formed, or with 90 days' written notice in the absence of a FAC. Payments are due after the first month of operations.
Cooperative Advertising ⁽⁴⁾	As determined by us	As determined by us	We may determine the amount of contribution if an advertising cooperative is established and active. Your contributions to cooperative advertising promotional programs will be credited toward your Individual Advertising Expense, up to a maximum of 3% of your Gross Revenues. All operators in the area of the cooperative will participate in the fund in proportion to its sales during the preceding twelve (12) month period.

Type of Fee	Amount	Due Date	Remarks
Initial Training for Additional Persons †	\$100 per day per person in excess of 2 people plus the cost of travel, lodging, meals and personal expenses	Payable prior to the beginning of the Initial Training Program and as incurred	If you want additional people to attend the initial training program, we will charge a training fee of \$100 per day per person. You will need to pay for airfare, lodging, ground transportation, meals, salary and benefits, and other personal expenses for each person attending the initial and any recurring or additional training program. Training fees can be increased or decreased by us at any time.
Franchisee Requested Additional Assistance †	\$150 per hour with a minimum of 4 hours (\$600) plus travel expenses, lodging and meals if on-site assistance is requested	50% at time of scheduling; expenses billed separately and remaining due upon arrival	If you request additional assistance if you are not meeting our requirements or if we determine that it is necessary for us to provide additional assistance to keep your location compliant with our standards.
Advanced Training Program [†]	Per Operations Manual, between \$100 per day per person who receives the training as determined by us	As required	Advance training programs may be offered based on your needs and requirements. Such programs may include Red Cross sponsored pet emergency medical training, practical applications and pet wellness.
Franchise Agreement Renewal Fee	\$5,000	At time of renewal	Payable when you renew your Franchise.
Franchise Agreement Transfer Fee [†]	\$10,000, plus our actual costs and expenses in evaluating the transfer request not to exceed an additional \$25,000.	At time of transfer	Payable when you transfer your Franchise to a person or entity approved by us. You are also responsible for paying our costs and expenses in evaluating the transfer request.
Proposed Product or Supplier Approval Fee	Any costs incurred by us in evaluating a product or supplier that you have proposed	As incurred	We will approve or disapprove of your proposed product or supplier within 30 days from the date we receive your request.
Kennel Operations Software Support Contract	\$2,700	Yearly	You must pay a fee to the software vendor we designate. You may pay on a monthly basis based on our vendors terms, however, the monthly fee may be more than on an annual basis.
Financial Audit [†]	Cost of Audit	Immediately upon completion of audit	If an audit discloses an understatement in any report of 2% or more of Gross Revenues, you must pay us for all cost and expenses connected with the audit, plus any unpaid royalties and brand fund amounts.

Type of Fee	Amount	Due Date	Remarks
Insurance ⁽⁵⁾	An annual premium of approximately \$12,000 - \$20,000, plus administrative fee of 15% of cost of insurance	As Incurred	We have the right to acquire insurance on your behalf if you fail to do so and charge you an administrative fee.
Indemnification [†]	Will vary under circumstances	As incurred	You must reimburse us if we are held liable for claims arising from your All American Pet Resorts® Business.
Cost of Enforcement or Defense [†]	All costs including accounting and attorney's fees	Upon settlement or conclusion of claim or action	You must reimburse us if we are required to incur any expenses in enforcing our rights against you under the Franchise Agreement.
Interest [†]	The lesser of 1.5% per month or the highest legal rate permitted by applicable law	Monthly	Begins to accrue after any payments are due and unpaid.
Late Report Fee [†]	\$100 per violation, per day	As incurred	Payable only if a required report or financial statement is not delivered when due. Revenue Reports are due the first business day of a new month for the previous month's business, unless that day falls on a Saturday or Sunday in which case the report is due on the first Monday of the new month. This fee is applicable after third occurrence.
Seminars, Conventions or Programs [†]	The estimated range of cost is \$500-\$1,000 per person, which does not include the cost of travel, lodging, meals and personal expenses for each person to attend, plus materials estimated at \$250	As incurred	You must attend our biennial franchise convention. We may conduct other mandatory meetings of all Franchisees. You must pay your expenses, as well as the expenses your Designated Business Manager and employees incur in attending these meetings.
Management Fee ^{†(6)}	7% of Gross Revenues, plus Incentive Fee of 10% of profits or \$5,000 per month plus expenses, whichever is greater	As incurred	If you request or we are required to send a representative to your Pet Resort to operate your All American Pet Resorts® Business when you are unable to do so.

Type of Fee	Amount	Due Date	Remarks
Non-Compliance Fee (7)	 \$250 first offense \$500 for second offense Upon third offense, \$500 	As Incurred	Applicable if you do not comply with terms of Franchise Agreement.
	plus daily fee of \$100 until default cured		

Denotes fees which are imposed and payable to us. All fees paid to us are uniformly imposed and are non-refundable under any circumstances. Fees paid to vendors or other suppliers may or may not be refundable depending on your vendors and suppliers. We reserve the right to require you to pay fees and other amounts due to us via electronic funds transfer as direct debits from your business bank operating account as described in the Franchise Agreement. Under this procedure, you must authorize us to initiate debit entries and/or credit correction entries to a designated checking or savings account for payments of fees, interest and/or other amounts payable to us. You must make the funds available to us for withdrawal by electronic transfer no later than the payment due date.

Notes:

- Royalty Fee and Gross Revenue. "Gross Revenues" means the total of all invoiced charges derived from (i) all sales of products and services at your All American Pet Resorts® Business, (ii) labor, (iii) insurance claims for lost profits to the extent a claim is paid by the insurer, (iv) all business related pass-through charges, and (v) all other products and services sold or performed by or for you or your All American Pet Resorts® Business or by means of the business conducted under this Agreement, whether the receipts are evidenced by cash, credit, checks, gift certificates, scrip, coupons, services, property or other means of exchange. Gross Revenues do not include the amount of any tax imposed by any federal, state, municipal or other governmental authority directly on sales and collected from customers, provided that the amount of any such tax is shown separately and in fact paid by the Franchisee to the appropriate governmental authority; and all customer refunds, approved and valid discounts and coupons, and credits made by the All American Pet Resorts® Business (exclusions will not include any reductions for credit card user fees, returned checks or reserves for bad credit or doubtful accounts).
- (2) <u>Individual Advertising Expense</u>. After the Start-Up Advertising Period expires and during the remaining Term of the Franchise Agreement, your required Individual Advertising Expense is equal to a minimum of 1% of your Gross Revenues for the preceding month or \$1,000, whichever is greater, per month for advertising and promotion within the Territory.

- Brand Development and Promotions Fee. The Brand Development and Promotions Fee is 2% of Gross Revenue. We may increase the Brand Development and Promotion up to a maximum of 3% of Gross Revenue upon thirty (30) days' notice, in consultation with the Franchisee Advisory Council, should a FAC be formed. In the absence of a FAC, we may increase these fees with thirty (30) days' written notice to you.
- (4) <u>Cooperative Advertising</u>. We may require you to participate in and contribute your share to such cooperative advertising and promotional programs in your Advertising Coverage Area. If we establish a cooperative advertising and promotional program in your Advertising Coverage Area, the cost of the program will be allocated among All American Pet Resorts[®] businesses in the Advertising Coverage Area and each outlet's share (including company- or affiliate-owned outlets) will be in proportion to its sales during the preceding 12-month period, or portion of such period. The voting power of Company- or Affiliate-owned outlets within a cooperative is equal to that of any franchisee-operated outlets. We do not currently have any Cooperative Advertising programs in place.
- (5) <u>Insurance</u>. If you fail to procure and maintain the required insurance coverage, we have the right and authority to procure the insurance coverage and charge you, which charges, together with a reasonable fee for our expenses incurred in this procurement, not to exceed 15% of the cost of the insurance coverage. You will pay this amount immediately upon notice.
- Management Fee. If you request or if certain events that would entitle us to terminate the Franchise Agreement occur, we may, but are not obligated to, provide management services to assist you to operate your All American Pet Resorts® Business for such period of time as the reason for our managing the All American Pet Resorts® Business lasts or until we decide to terminate our management of the All American Pet Resorts® Business, whichever occurs first. If we provide management services, you must pay our out-of-pocket expenses of providing management services plus a management fee equal to 7% of the Gross Revenues of your All American Pet Resorts Business and an incentive fee of 10% of the cash flow after payment of all expenses of operation of your All American Pet Resorts® Business, including our out-of-pocket expenses and management fee, or \$5,000 per month plus expenses, whichever is greater. Management services include but are not limited to the services described in Section 8.7 of the Franchise Agreement.
- Non-Compliance Fee. If you violate or breach any term of the Franchise Agreement, including your failure to pay or to have adequate amounts available for electronic transfer of amounts you or your affiliates owe us or our affiliates, your failure to timely provide required reports and financial information, and/or your failure to attend in its entirety all mandatory meetings such as our franchisee convention, we may require you to pay such non-compliance charges. The daily charge will continue until we determine that you have cured all deficiencies and are compliant with all terms of the Franchise Agreement. These non-compliance charges are intended to compensate us for additional expenses and certain losses we will incur as a result of your non-compliance and are not a penalty or an expression of the total amount of such damages. We may periodically change or eliminate this charge. Imposition of this fee does not waive any rights we otherwise have under the Franchise Agreement.

ITEM 7 ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT						
Type of Expenditure ⁽¹⁾	Amount (Low – High)	Method of Payment	When Due	To Whom Payment is Made		
Initial Franchise Fee ⁽²⁾	\$69,500 - \$69,500	Lump Sum	As described in Note (2)	Us		
Travel and living expenses while training ⁽³⁾	\$0 - \$5,000 per person	As incurred	As incurred during training	Airlines, hotels, restaurants		
Real Estate Leasing ⁽⁴⁾	state Leasing ⁽⁴⁾ \$0 - \$20,000 As negotiated Monthly or according to lease terms		Us or an affiliate, if we or an affiliate own the Pet Resort, otherwise, as per terms of the lease.			
Leasehold Improvements ⁽⁵⁾	\$500,000- \$1,200,000	As agreed	Varied times	Building contractor		
Furniture, Fixtures, Equipment ⁽⁶⁾	\$3,000 - \$10,000	As agreed	Varied times	Suppliers, vendors		
Signage ⁽⁷⁾	\$5,000 - \$15,000	As billed	Terms of supplier	Supplier		
Computer hardware, software and cameras ⁽⁸⁾	\$15,000 - \$25,000	Lump sum	At delivery	Suppliers, vendors		
Utility Deposits & Licenses ⁽⁹⁾	\$2,000 - \$5,000	Lump Sum Negotiable	Negotiable/ As Incurred	Landlord/Utilities /Local Government Agencies		
Legal, Architectural and Accounting Fees ⁽¹⁰⁾	\$20,000 - \$40,000	As incurred	As Invoiced	Attorneys, Accountants & Architects		
Inventory and Supplies(11)	\$5,000 - \$10,000	Prior to opening and as needed	At delivery	Suppliers		
Insurance ⁽¹²⁾	\$3,000 - \$15,000	As billed	As billed	Provider		
Start-Up Advertising Expense ⁽¹³⁾	\$20,000 - \$25,000	As incurred	Varied times	Vendors		
Additional funds for first 3 months ⁽¹⁴⁾	\$25,000 - \$50,000	As agreed	As incurred	Vendors or third parties		
Real Estate & Resort Coordination Fee (15)	\$20,000	Lump Sum	(15) days after you sign a Franchise Agreement	Us		
TOTAL ⁽¹⁶⁾	\$687,500- \$1,509,500					

Notes:

- (1) <u>Expenditures</u>. All fees imposed by us are non-refundable unless otherwise noted. Fees and expenses paid to vendors or other third parties may or may not be refundable depending on the terms negotiated with them.
- (2) <u>Initial Franchise Fee</u>. The Initial Franchisee Fee is \$69,500 for your first franchise and is due when you sign the Franchise Agreement. The Initial Franchisee Fee for your second and any additional franchise is described in Item 5. The Initial Franchise Fee is non-refundable.
- Travel and Living Expenses While Training. We provide twenty (20) days of training in part at a location designated by us and in part at your Pet Resort. You must pay for airfare, meals, transportation costs, salaries, benefits, lodging and incidental expenses for all initial training program attendees. You must spend a portion of training at an existing All American Pet Resorts[®] Business working at the Business. You are responsible for paying all of your costs and expenses associated with these onsite visits.
- (4) Real Estate Leasing. These amounts may also include a security deposit equal to one month's rent. The actual amount of your real estate leasing expenses will vary depending on local landlord practices and other factors. In certain circumstances, we or an affiliate may own the facility used by your All American Pet Resorts[®] Business and lease it to you.
- Leasehold Improvements. You will need to make certain improvements to your premises to accommodate your All American Pet Resorts® Business or to comply with our standards and specifications. We estimate that the cost of leasehold improvements for a 100 suite All American Pet Resorts® Business is approximately \$10,000-\$15,000 per suite. Generally, an All American Pet Resorts® Business has 80-100 suites, but in certain instances, we may approve an All American Pet Resorts® Business with fewer than 80 suites. The lower estimate in the table above reflects an All American Pet Resorts® Business with 50 suites. The higher estimate in the table above reflects an All American Pet Resorts® Business with 80 suites. The cost of leasehold improvements will vary, however, based upon size, condition and location of the premises, local wage rates and material costs. This estimate is based on leasing a building. It does not include the costs to purchase a building. A building suitable for an All American Pet Resorts® Business that could service 100 dogs requires approximately 6,500 square feet and located on at least .75 acre of land.
- (6) <u>Furniture, Fixtures, Equipment</u>. You will be required to purchase or lease office furniture and any equipment necessary for providing the services offered by All American Pet Resorts[®] Businesses. Although some of these items may be leased, the range shown represents the actual purchase price.
- (7) <u>Signage</u>. These amounts are estimates of your costs for exterior signs as allowed by local ordinance. The cost of signs may vary significantly depending on the location your All American Pet Resorts[®] Business and local ordinance/zoning restrictions.
- (8) <u>Computer Hardware, Software, and Cameras.</u> The estimated initial investment includes costs related to the purchase of specified computer hardware, software, printer, and web-

based cameras. We require you to provide us with electronic access to certain daily information and access to all cameras 24 hours per day, 365 days per year. You will be required to use Software designated by us (See Item 8). The annual fee paid to our designated supplier for the Kennel Operations Software is \$2,000. You will also pay a set-up fee of \$300 for the Kennel Operations Software.

- (9) <u>Utility Deposits /Licenses</u>. These amounts include estimated utility deposits. The actual amount of these deposits will vary depending on local practices and other factors. These amounts also include the estimated cost of obtaining licenses or permits to operate a pet care and boarding Pet Resort. Other permit and license fees will also vary depending on location and on whether applicable laws require the payment of occupational or other taxes for pet care, boarding and daycare facilities. Franchise taxes are not included.
- (10) <u>Legal Architectural and Accounting Fees</u>. You will need to employ an attorney, architect, accountant and other consultants to assist you in establishing your All American Pet Resorts[®] Business. These fees may vary from location to location depending upon the prevailing rate of attorneys', architects', accountants', and consultants' fees.
- (11) <u>Inventory and Supplies</u>. Your initial inventory and supplies will typically include pet food, cleaning and janitorial supplies, food and water bowls, pet bedding, laundry supplies, uniforms, grooming supplies, temporary leashes, office supplies and two-way radios. We may change the inventory and supplies you are required to purchase at any time.
- (12)Insurance. You must procure and maintain, at your own expense, insurance policies protecting you, us, our designated Affiliates and each of our officers, directors and employees against any loss, liability, personal injury, death, property damage, business interruption or expense resulting from the operation of your All American Pet Resorts® Business and all services you provide in connection with the operation of your All American Pet Resorts[®] Business as we may require for your and our protection in amounts noted in the Operations Manual and Franchise Agreement (which may be adjusted periodically). You must procure and maintain business interruption insurance covering at least twelve (12) months of operations. You must also procure and maintain all other insurance required by state or federal law, including workers compensation insurance and unemployment insurance. The policies must also stipulate that we must receive a 30-day prior written notice of cancellation and must contain endorsements by the insurance companies waiving all rights of subrogation against us. Original or duplicate copies of all insurance policies, certificates of insurance or other proof of insurance acceptable to us, including original endorsements affecting the coverage required by us must be furnished to us together with proof of payment within 10 days of issuance. You must also furnish us with certificates and endorsements evidencing such insurance coverage within 10 days after each of the following events: (i) at all policy renewal periods, no less often than annually; and (ii) at all instances of any change to, addition to, or replacement of any insurance. The certificates and endorsements for each insurance policy are to be signed by a person authorized by that insurer to bind coverage on its behalf. All certificates and endorsements are subject to our approval. Your cost of insurance may vary based on the location of your All American Pet Resorts® Business, value of your leasehold improvements, amount of inventory, number of employees and other factors.

- (13) <u>Start-Up Advertising Expense</u>. For the first 3 months after you satisfactorily complete the initial training program, you must spend a total of at least \$20,000 on promotional advertising within the Territory. However, you may choose to spend more on promotional advertising to promote and advertise the opening of your All American Pet Resorts[®] Business.
- Additional Funds for First 3 Months. This amount includes estimated operating expenses you should expect to incur during the first 90 days of operations for the operation of a single unit, not including any revenue generated by your All American Pet Resorts® Business. It includes your estimated payroll costs, deposits, fees for city, state and local business licenses, business entity organization expenses, other prepaid expenses, accounting and professional fees, and other operational expenses. These figures do not include any taxes that you may pay or Royalty Fees. You should check with your local and state governmental agencies for any taxes that may be assessed.
- (15) Real Estate & Resort Coordination Fee. The Real Estate & Resort Coordination Fee is payable to us and is due (15) fifteen days after you sign the Franchise Agreement. The fee is due per location we approve you to open and is due and payable upon your signing of a Franchise Agreement to open additional location. We will advise and consult with you, your real estate provider, architect, and contractor regarding numerous aspects in the development of your resort. This includes real estate comparison review, site inspection, demographic analysis, lease review & negotiation, resort design and construction consultation. The Real Estate & Resort Coordination Fee is non-refundable.
- Total Estimated Initial Investment. These figures are estimates only and are based on our (16)estimate of nationwide costs and market conditions prevailing as of the date of this Disclosure Document. We have relied on over 31 years of combined experience in the pet care industry to compile these estimates. You must bear any deviation or escalation in costs from the estimates that we have given. You should review these figures carefully with a business advisor before making any decision to purchase an All American Pet Resorts® Business. Many factors that are unique to your location can make a dramatic difference in the estimates provided. You may incur additional expenses starting your All American Pet Resorts® Business. Your costs depend on several factors, including how well you follow our methods and procedures; your management skill, experience and business acumen; local economic conditions; the local market for our services; the prevailing wage rate; competition; and sales levels reached by your All American Pet Resorts® Business during the initial period. The availability and terms of financing depend on several factors, including the availability of financing generally, your creditworthiness, collateral you may have and lending policies of financial institutions.

ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

You must establish and operate your All American Pet Resorts[®] Business in compliance with your Franchise Agreement and the standards and specifications contained in the All American Pet Resorts[®] confidential Operations Manual ("**Operations Manual**"), which is a collective term

for the multiple publications we may provide you relating to the operation of the All American Pet Resorts® Business and the operation of franchises, and an All American Pet Resorts® Architectural Specification Document that we will loan to you.

We have established standards and specifications for your Pet Resort, equipment, uniforms, supplies, forms, Products, Services, advertising materials and most other services and products used in, sold or provided through your All American Pet Resorts® Business. The services include providing professional pet boarding, daycare and grooming services ("Services"). The products include products related to the Services, including branded kennels, sanitation, janitorial supplies and equipment, scoopers, hoses, pet food, bowls, leashes and collars, grooming tables and supplies, bedding, storage bins and similar items ("Products"). We reserve the right to require that you sell additional or different Services and Products in your All American Pet Resorts® Business on 30 days prior written notice to you. There is no limit on the amount of additional expense you may incur in connection with such Services and Products. You must provide the Services and sell the Products using our specifications and standards. We reserve the right to change our standards and specifications with 30 days prior written notice to you.

In order to maintain our standards of consistent, high quality Products, customer recognition, advertising support, value and uniformity in all American Pet Resorts[®] Businesses, you must purchase all of your required equipment, supplies, fixtures, inventory, goods, services and Products used in or sold through your All American Pet Resorts® Business, per our specifications and standards, only from us or our approved or designated suppliers and distributors. As of the date of this Disclosure Document neither we nor our affiliates are an approved supplier, but we reserve the right to become an approved supplier at any time. As of the date of this Disclosure Document none of our officers own an interest in any approved supplier. Upon request, we may provide specifications or standards to you; however, you may not purchase any product, material or supply from any non-approved or non-designated supplier, even if such product, material or supply complies with our standards and specifications. If you want us to approve any product, material or supply you want to sell or offer, or purchase any products from a supplier not currently approved, you must notify us and you must submit samples and other information to us so that we can make an informed decision as to whether your proposed product or supplier satisfies our standards. We may require that our representatives be permitted to inspect the supplier's facilities and that samples from the supplier be delivered to us or another company for testing. We will approve or disapprove of your proposed product or supplier within 60 days from the date we receive your request. You will be charged for any costs incurred in evaluating a product or supplier. We may re-inspect the facilities and retest the products of any such approved supplier periodically and may revoke our approval if the supplier fails to continuously meet the above criteria. It is a material breach of your Franchise Agreement if you buy Products, equipment, supplies, fixtures, inventory, goods or services from anyone other than our approved suppliers without our prior written approval.

We estimate that the purchase of these supplies, equipment, inventory, fixtures, goods, services and Products from us or our designated or approved sources, or those meeting our standards and specifications, will be approximately 25% to 50% of your total cost to establish an All American Pet Resorts[®] Business and 25% to 50% of your total cost of operating an All American Pet Resorts[®] Business (not including amortization, depreciation, or replacement of worn or obsolete improvements, equipment or fixtures).

In our last fiscal year ending December 31, 2023, we received \$5,000 as rebates on approved products and services from suppliers. Other than this amount, neither we nor our Affiliate collected any revenue, rebates or other consideration directly from a franchisee or from a supplier or distributor based upon franchisee required purchases or leases, but we reserve the right to derive revenue from your purchases or leases of goods, services, supplies, fixtures, equipment, inventory and Products from us, our Affiliate or our approved or designated suppliers and distributors in the future. For example, we or our Affiliate may collect revenue, rebates or other consideration by charging you more than the cost of the product or service or by receiving a rebate from a supplier or distributor that is based on the amount of products or services you purchase.

You must use the proprietary computer programs and related materials developed for use in the operation of the All American Pet Resorts[®] Business, including without limitation Kennel Operations Software and the use of QuickBooks, that we designate ("**Software**"). We require that you pay any license and support fees for the Software (See Item 6). You will also pay a set-up fee for the use of the Software (See Item 7). The purchase of the Software license may include technical support. You may use the Software on the computer equipment and hardware of your choice, so long as it meets specifications to run the required software ("**Computer Equipment**").

You must procure and maintain, at your own expense, insurance policies protecting you, us, our designated Affiliates and each of our officers, directors and employees against any loss, liability, personal injury, death, property damage, business interruption or expense resulting from the operation of your All American Pet Resorts® Business and all services you provide in connection with the operation of your All American Pet Resorts® Business as we may require for your and our protection in amounts noted in the Operations Manual and Franchise Agreement (which may be adjusted periodically). You must also procure and maintain all other insurance required by state or federal law, including workers compensation insurance and unemployment insurance. Specifically, you must maintain workers' compensation insurance with limits not less than: \$100,000 bodily injury/disease/employee, \$500,000 bodily injury/disease aggregate, and \$100,000 bodily injury by accident. You must also maintain comprehensive auto liability insurance with a minimum limit of \$500,000/occurrence combined with a single limit bodily injury and property damages for owned, non-owned, leased or hired automobile. You must procure and maintain business interruption insurance covering at least twelve (12) months of operations to protect against property damage and business losses due to fire, wind, and other acts of God, among others. You must also maintain comprehensive general liability insurance, which shall be site-specific, with minimum limits of \$2,000,000 combined single limits bodily injury and property damage/occurrence. Your cost of insurance may vary based on the location of your All American Pet Resorts[®] Business, value of your leasehold improvements, amount of inventory, number of employees and other factors. You must maintain Employment Related Practices insurance with minimum limits of \$2,000,000. All required insurance policies must list us and/or our designee as additional interests. Further, Franchisor, Franchisor's Affiliates and their respective officers, directors, agents, employees shall be designated as additional insureds under all required insurance.

We do not have any purchasing or distribution co-operatives as of the date of this Disclosure Document. We may negotiate purchase arrangements with suppliers and distributors of approved products for the benefit of our franchisees and we reserve the right to receive rebates on volume discounts from our purchase of products that we may re-sell to you. We do not provide

material benefits, such as renewing or granting additional franchises to franchisees, based on their use of designated or approved suppliers. There are no caps or limitations on the maximum amount of rebates we may receive from our suppliers as the result of franchisee purchases.

ITEM 9 FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this disclosure document.

		Section in	
		Franchise	Disclosure Document
	Obligation	Agreement	Item
a.	Site selection and acquisition/lease	Sections 8 &9	ITEM 11
b.	Pre-opening purchases/leases	Sections 9 and 10	ITEM 8 & ITEM 11
c.	Site development and other pre-opening requirements	Section 8	ITEM 5, ITEM 6, ITEM 7 & ITEM 11
d.	Initial and ongoing training	Sections 8 and 9	ITEM 11
e.	Opening	Section 9 and 15	ITEM 7, ITEM 11
f.	Fees	Sections 6, 7 and 12	ITEM 5 & ITEM 6
g.	Compliance with standards and policies / Operations Manual	Sections 8(e) and 9	ITEM 11
h.	Trademarks and proprietary information	Section 11	ITEM 13 & ITEM 14
i.	Restrictions on products/services offered	Sections 9 and 10	ITEM 8 & ITEM 16
j.	Warranty and customer service requirements	Section 9	ITEM 11
k.	Territorial development and sales quotas	Not Applicable	ITEM 12
1.	Ongoing product/service purchases	Sections 9 and 10	ITEM 16
m.	Maintenance, appearance and remodeling requirements	Sections 3 and 9	ITEM 7
n.	Insurance	Section 13	ITEM 6, ITEM 7, ITEM 8 & ITEM 11
0.	Advertising	Section 12	ITEM 11
p.	Indemnification	Section 13	ITEM 6
q.	Owners participation/Management/Staffing	Sections 8 and 9	ITEM 1, ITEM 11 & ITEM 15
r.	Records and reports	Section 7	ITEM 6 & ITEM 17
s.	Inspections and audits	Sections 7	ITEM 6
t.	Transfer	Section 16	ITEM 17
u.	Renewal	Section 4	ITEM 17
v.	Post-termination obligations	Sections 11, 15, and 18	ITEM 17
w.	Non-competition covenants	Section 15	ITEM 17
х.	Dispute resolution	Section 20	ITEM 17
y.	Other	Not Applicable	Not Applicable

ITEM 10 FINANCING

We do not offer direct or indirect financing. We do not guarantee your note, lease or obligation.

ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

Pre-opening Obligations

Before you open your All American Pet Resorts® Business, we (or our designee) will provide the following assistance and services to you.

- 1. Approve the site for your All American Pet Resorts® Business. We do not own or lease the Pet Resort for your All American Pet Resorts® Business to you. We will analyze a location by examining population density, census data, and proximity of the proposed location to other All American Pet Resorts® Businesses, or any other criteria noted in the Franchise Agreement. The Pet Resort site is subject to our approval, which may be granted or denied as we determine appropriate. We will approve or reject sites within 10 days after you have submitted all information necessary to determine if the location is appropriate. If we cannot agree on a proposed site, Franchisor may elect to terminate this Agreement. Notwithstanding anything herein to the contrary, Franchisor may extend the time periods set forth in the Franchise Agreement (Sections 8.4 and 9.2(b) of the Franchise Agreement).
- 2. Assist you with prototype plans and specifications to the extent available for the Pet Resort as well as recommendations for initial and replacement equipment, tools, inventory and supplies required for the operation of your All American Pet Resorts® Business (Sections 8.3(b) and 10 of the Franchise Agreement). We are not responsible for the cost and expense associated with developing any original plans for your location. During the design phase and development of your All American Pet Resorts® Business you or your designated representative will be required to spend at least 1 working day at your expense at an existing All American Pet Resorts® Business in order to further understand the architectural specifications and the implications of those specifications in the development of your All American Pet Resorts® Business.
- 3. Designate your Territory (Sections 5 and 8.3(a) of the Franchise Agreement and Attachment B to the Franchise Agreement).
- 4. Prior to your Opening Date, we will conduct a training course for you, which will be up to twenty (20) days, or if you are not an individual, your Designated Business Manager and 1 additional person. The training is conducted in phases and will be held at a location we designate. Other training will be added as services mature and as market demands. You must pay for airfare, lodging, meals, ground transportation, salaries and benefits, and any other travel expenses for yourself and any additional attendees that are incurred during this time. (Section 8.3(d) of the Franchise Agreement).

- 5. Loan one copy of our confidential and proprietary Operations Manual to you at the time of execution of the Franchise Agreement. The Operations Manual consists of 1 or more manuals, technical bulletins or other written materials. We may modify the Operations Manual periodically as we determine appropriate. The Operations Manual may be in an electronic format. We reserve the right to require you to use an electronic version of the Operations Manual and to require you to access the document using the Internet or an intranet created and supported by us. The Operations Manual contains approximately 229 pages that summarize operational procedures for kennel staff and common issues involved with managing and operating a franchised business. The table of contents for the Operations Manual can be found in Exhibit I to this Disclosure Document.
- 6. Assist you in developing an advertising and promotions plan for the Start-Up Advertising Period (Below and Section 12.1 of the Franchise Agreement.)
- 7. Provide on-site pre-opening and grand opening assistance during the first 3 months of operations of your All American Pet Resorts[®] Business. We may charge you a reasonable fee for on-site pre-opening services (Section 8.3(g) of the Franchise Agreement).

Schedule for Opening

We estimate that the typical length of time between signing the Franchise Agreement and opening your All American Pet Resorts[®] Business will be 12 to 18 months. Some factors which may affect this timing are your ability to acquire a desired site through lease negotiations or land acquisition, your ability to secure any necessary financing, your ability to comply with local zoning and other ordinances, your ability to obtain any necessary permits and certifications, the timing of the delivery of equipment, tools and inventory and the time to convert, renovate or build the boarding, daycare and grooming Pet Resort. The Franchise Agreement governing your All American Pet Resorts[®] Business will be effective and binding when signed by us and you and will continue through the Operating Term. The Operating Term will begin on the day you sign your All American Pet Resorts[®] Franchise Agreement and will continue for a period of 10 years (the "Operating Term").

You must comply with all applicable ordinances, building codes and permit requirements and with lease requirements and restrictions. You must apply for all required licenses and permits in sufficient time to open your All American Pet Resorts® Business. Specifically, we will have the option to terminate the Franchise Agreement and retain your Initial Franchise Fee if you have not: (a) obtained our approval for a location for your All American Pet Resorts® Business within 4 months of executing the Franchise Agreement (*see* Item 5) or (b) opened for business within 18 months of executing the Franchise Agreement after diligent pursuit (*see* Item 5).

You may not open your All American Pet Resorts® Business until: (1) we notify you in writing that all of your obligations have been fulfilled; (2) initial training is completed to our satisfaction; (3) all amounts due to us have been paid; (4) we have been furnished with copies of all insurance policies and certificates required by the Franchise Agreement, or other documentation of insurance coverage and payment of premiums that we request; (5) you notify us that all approvals and conditions stated in the Franchise Agreement have been met; (6) you have received all required permits and licenses; (7) you have provided us with a fully signed copy of the Lease for your Pet Resort negotiated per the terms of the Franchise Agreement; and (8) you have ordered,

received and installed your equipment, supplies, inventory, tools, products, uniforms and computer equipment and software required by us. You must begin operating your All American Pet Resorts[®] Business is ready for opening.

Continuing Obligations

During the term of the Franchise Agreement, we (or our designee) will:

- 8. Inform you of mandatory specifications, standards and procedures for the operations of your All American Pet Resorts® Business (Section 9 of the Franchise Agreement). There are no additional charges for these services.
- 9. Make a representative reasonably available to speak with you on the telephone during regular business hours to discuss your operational experiences and support needs (Section 8.5 (a) of the Franchise Agreement). There are no additional charges for these services.
- 10. Research new Products, Services and training methods and provide you with information concerning developments of this research (Section 8.5(d) of the Franchise Agreement). There are no additional charges for these services.
- 11. Maintain the Brand Development and Promotions Fund and use these funds to develop promotional and advertising materials and programs for All American Pet Resorts[®] Businesses (Section 8.5(e) of the Franchise Agreement). There are no additional charges for these services.
- 12. Assist you in developing a promotional advertising plan and provide advertising materials to you in the form of a graphics arts package, which is included in your Operations Manual (Section 8.5(g) of the Franchise Agreement). There are no additional charges for these services.
- 13. Our representative may provide additional assistance if we determine it is necessary, including visits to your All American Pet Resorts[®] Business (Section 8.5(g) of the Franchise Agreement). There may be additional charges for these services (\$150 an hour, 4-hour minimum), along with travel expenses, lodging and meals. If we provide additional assistance, we must agree in advance on the charges you will pay and the length of the visit.
- 14. Hold periodic conferences to discuss branding, sales techniques, new product developments, bookkeeping, training, accounting, inventory control, performance standards, advertising programs, merchandising procedures and other topics. You must attend these meetings and pay any attendance fee, including \$500 to \$1,000 per person and an estimated \$250 in materials, and all of your travel and living expenses and expenses for any of your personnel to attend. These conferences are held at a location we choose (Section 8.5(b) of the Franchise Agreement).
- 15. We may provide you with a quarterly newsletter (Section 8.5(h) of the Franchise Agreement).
- 16. We may provide management services to assist you in the operation of your All American Pet Resorts[®] Business if you are unable to do so because of your or your Designated Business Manager's death or disability or under certain circumstances where you or your Designated Business Manager are

unable to devote your full time and attention to the American Pet Resorts[®] Business (Section 8.7 of the Franchise Agreement).

We reserve the right to delegate some or all of our pre-opening and continuing obligations under the Franchise Agreement to an Area Director with regional responsibility over the geographic area in which you operate your All American Pet Resorts® Business. Except as listed above, we do not provide any additional assistance to you.

Advertising Programs

Start-up Advertising and Individual Advertising Expense

For the first 3 months after you satisfactorily complete the initial training program, you must spend a total of at least \$20,000 on promotional advertising within the Territory ("Start-Up Advertising Period"). After the Start-Up Advertising Period and during the remaining Operating Term of the Franchise Agreement, you must spend a minimum of 1% of your Gross Revenues for the preceding month or \$1,000, whichever is greater, per month ("Individual Advertising Expense") for advertising and promotion within the Territory. We will assist you in developing an advertising plan for both the Start-Up Advertising Period and thereafter. You may not advertise outside your Territory without our approval (Section 12.1 of the Franchise Agreement).

Brand Development Fund

In addition to your Local Advertising Expenses, beginning with the fourth month of the Operating Term you must pay us a brand development fee ("Brand Development Fee") of 2% of your monthly Gross Revenues. The Brand Development Fee is paid at the same time that you pay your Royalty. We may increase the Brand Development Fees to up to 3% of your monthly Gross Revenues. We will provide thirty (30) days advance notice before increasing the Brand Development Fee. We will deposit the Brand Development Fee in a separate bank account, commercial account or savings account ("Brand Development Fund"). We will administer the Brand Development Fund, as we determine appropriate, and we may use a professional advertising agency or media buyer to assist us. Your contribution to the Brand Development Fund will be in addition to all other advertising fees set out in this ITEM 11 (Section 12.4 of the Franchise Agreement). During the fiscal year ended December 31, 2023, a total of \$193,205 was collected in the Brand Development Fund, and we spent \$173,679 as follows: 32.95% was used for search engine optimization and social media, 6.04% was spent on marketing, 5.6% was spent on administrative expenses, 8.57% was spent on other activities and 46.83% was carried over to 2024. No advertising expenditures from the Brand Development Fund were devoted principally to the sale of new franchises. This fund is unaudited, and we will make available to you once a year, upon request by telephone or written correspondence, an unaudited representation for expenditures from the Brand Development Fund that shows how the fund proceeds have been raised and spent for the previous year within 120 days after our fiscal year end.

In the event that we choose to establish a franchisee advisory council (the "FAC"), we may consult with the FAC regarding the administration of the Brand Development Fund, changes in the Brand Development Fund and the format for marketing and promotional campaigns conducted by the Brand Development Fund or directly by franchisees. The FAC will be composed of

members of our senior management and Franchisee representatives. We intend that the Franchisee representatives will be elected by the Franchisees; however, until there are a sufficient number of franchisees in the System, we may appoint the Franchisee representatives to the FAC. We must approve the number of meetings to be held by the FAC and the procedures under which the FAC will operate (Section 12.4(f) of the Franchise Agreement).

We may reimburse ourselves, our authorized representatives or our Affiliates from the Brand Development Fund for administrative costs; independent audits; reasonable accounting, bookkeeping, reporting and legal expenses; taxes and all other reasonable direct or indirect expenses that may be incurred by us or our authorized representatives and associated with the programs funded by the Brand Development Fund. We assume no other direct or indirect liability or obligation to collect amounts due to the Brand Development Fund or to maintain, direct or administer the Brand Development Fund. Any unused funds in any calendar year will be applied to the following year's funds, and we reserve the right to contribute or loan additional funds to the Brand Development Fund on any terms we deem reasonable.

We may use the Brand Development Fund for the creation, production and placement of commercial advertising, including Internet advertising; rebranding of domain and associated electronic mail addresses for franchisees; agency costs and commissions; creation and production of video, audio and written advertisements; administering multi-regional advertising programs, direct mail and other media advertising; in-house staff assistance and related administrative costs; local and regional promotions; supporting public relations; market research; and other advertising and marketing activities, including participating at trade shows. Advertising may be placed in local, regional, or national media of our choice, including print, direct mail, radio or television, and the Internet. We do not guarantee that advertising expenditures from the Brand Development Fund will benefit you or any other franchisee directly, on a pro rata basis, or at all. Except as described above, we are not obligated to spend any amount on advertising in the geographical area where you are or will be located.

We will not use Brand Development Fund monies to solicit franchisees. In all instances the phrase "Franchises Available" will be included in all Franchisor participation. Neither our Affiliate nor we receive payments for providing goods or services to the Brand Development Fund, except for reimbursement of expenses as described above.

You must order sales and marketing materials from our designated supplier. It is a material breach of the Franchise Agreement to use other marketing materials without prior written approval. If you desire to use your own advertising materials, you must obtain our prior approval. We will review your request, and we will respond in writing within 48 hours from the date we receive all requested information. Our failure to notify you in the specified time frame will be deemed a disapproval of your request. All advertising must include the phrase "Franchises Available." Use of logos, Marks and other name identification materials must be consistent with our approved standards. You may not use our logos, Marks, and other name identification materials on items to be sold or services to be provided without our prior written approval. If we approve of promotional items or services that will be sold in your All American Pet Resorts® Business, those items or services must be included in your Gross Revenues and will be subject to Royalties, Individual Advertising Expense and the Brand Development Fee.

We retain the right to market on the Internet, including all use of websites, social media platforms, domain names, URL's, linking, advertising, and co-branding arrangements. You may not independently market on the Internet or through social media platforms, or use any domain name, address, locator, link, metatag, or search technique, with words or symbols similar to the Marks, without our prior written approval. We intend that any franchisee website be accessed only through our home page. You will provide us content for our Internet marketing. We retain the right to approve any linking or other use of our website.

Cooperative Advertising

From time to time we may designate a local, regional or national advertising coverage area in which your All American Pet Resorts[®] Business and at least 1 other All American Pet Resorts[®] Business is located for purposes of developing a cooperative local, regional or national advertising or promotional program ("Advertising Coverage Area"). You must participate in and contribute your share to such cooperative advertising and promotional programs in your Advertising Coverage Area. The cost of the program will be allocated among All American Pet Resorts[®] businesses in the Advertising Coverage Area and each outlet's share (including company- or affiliate-owned outlets) will be in proportion to its sales during the preceding 12-month period, or portion of such period. Your contributions to cooperative advertising promotional programs will be credited toward your Individual Advertising Expense, up to a maximum of 3% of your Gross Revenues. The Advertising Coverage Area will consist of the area covered by the particular advertising medium (television, radio or other medium) as recognized in the advertising industry.

We may establish regional advertising councils ("Council(s)") to administer the cooperative advertising program, which will be comprised of us, franchisees in the Advertising Coverage Area, and company-owned or affiliate-owned All American Pet Resorts[®] Businesses. At the time a program is designated, we will submit a list to you of all operating All American Pet Resorts[®] Businesses (including company-owned and affiliate-owned outlets) within your Advertising Coverage Area. We will determine in advance how each Council will be organized and governed, who will administer it, and when it must start operation. If a Council has been established for an Advertising Coverage Area where your All American Pet Resorts[®] Business is located when the Franchise Agreement is signed, or if any Council is established during the term of the Franchise Agreement, you must become a member of the Council and abide by the rules of the Council. We reserve the right to form, change, dissolve or merge any Council.

If we have established a Council for your Advertising Coverage Area, you must contribute to the Council the amounts required by its written governing documents. All contributions to the Council will be maintained and administered in accordance with the written documents governing the Council. If there are company-owned and affiliate-owned outlets in your Advertising Coverage Area offering products and services similar to the All American Pet Resorts® Business, they will make contributions to the Council equal to the contributions required of the All American Pet Resorts® Businesses within that area. The Council will be operated solely as a conduit for collecting and spending cooperative fees for the purposes outlined above. The Council may not use any advertising or promotional plans or materials without our prior approval.

The members of the Council, subject to our approval, will determine the amount of contribution, and whether other franchisees must contribute the same amount or at the same rate.

Each member will have one vote for each All American Pet Resorts® Business operated by the member within the Advertising Coverage Area subject to the Council, and a simple majority will carry each motion. You will be obligated by the Franchise Agreement to pay any increased contributions even if you vote against the increase. Each Council will have to prepare an annual financial statement reporting its expenditures for the previous year to its members.

We have not created any Council as of the date of this Franchise Disclosure Document.

Software, Computer, Printer and Camera Equipment

You must use the computer software required by us on the hardware of your choice, as long as that hardware meets our specifications. Currently, you are required to purchase a desktop personal computer ("Hardware") that runs on the Windows operating system. Your computer must also have Kennel Operations Software, Point of Sale ("POS") Software, and any other software specified by us(collectively, the "Software") and be web-based with access to the Internet. You also must purchase a printer that includes faxing and scanning capabilities and web-based cameras that fit our standards and specifications. The Hardware, Software, printer, and web-based cameras are referred to as the "Computer System," which costs between \$15,000 and \$30,000. You must update your Computer System, at your expense, as we may require from time to time to meet our specifications as they evolve. Upgrades, in some cases, may only be available through our suppliers. You are responsible for ongoing maintenance, repairs, upgrades or updates for your Computer System. The annual cost of optional or required maintenance, updating, upgrading or support contracts is approximately \$500.

You will be required to purchase a subscription to the Kennel Operations Software offered by our designated supplier. This Software will run your All American Pet Resorts[®] business and its day-to-day operations. The reports in the Software will be used to track all of your boarding, daycare, grooming, services, training and/or retail revenue. You must use these reports to communicate your gross revenue to us on the first day of every new month. We may inspect your POS data and other financial records, without requesting your approval, at any time. We may use the Software to access the cameras located in your All American Pet Resorts[®] business at any time. You will be required to purchase an annual support contract from the Software vendor and from time to time, perform all upgrades from the vendor at your own expense. The annual cost of the support contract is \$2,000. You will also pay a set-up fee of \$300 to our designated supplier.

We reserve the right to mandate that you use financial software we designate to manage the finances of your All American Pet Resorts Business (the "Financial Software"). Training on the Financial Software is your responsibility and at your expense. We will provide you with a standardized chart of accounts in order to maintain consistency with the System. All financial reports we require must be reported in the form, format and schedule we designate. Currently, all financial reports must be submitted to us electronically. Regardless of the software utilized in your Franchised Business and whether it is designated by us, you must cooperate in providing us direct access to said software and data you generate in the software platform.

You must have sufficient computer skills to be able to operate your Computer System and to access e-mail and the Internet. You must have access to the Internet and maintain an email account that allows us to communicate with you on a regular basis. You must check your email

account at least once every day. If we determine that you require additional computer training, you must take and pay for, at your own expense, a computer training course at a local computer training school. You must complete this training within three months of the day we advise you of this requirement, and you must present us with a certificate acceptable to us to show that you have satisfactorily completed the training.

We have the right to independently access your All American Pet Resorts® information and data electronically through our proprietary data management and internet system, and to collect and use your electronic information and data in any manner we choose to promote the development of the System, the Brand and the sale of franchises. There is no contractual limitation on our right to receive or use information through our proprietary data management and intranet system.

You are required to have at least one high resolution, hand-held digital camera and several permanently mounted interactive video cameras in your All American Pet Resorts® of the specific brand and type which meet our brand standards and requirements. Cameras must be accessible through the Internet and be viewable from any browser at any time. The quantity of interactive cameras will be determined by the size of your All American Pet Resorts®. Interactive video cameras will be used by customers, staff, management and franchisor and must be kept on-line at all times. The number of cameras viewable by the public will be determined by the size of your All American Pet Resorts®. The hand-held digital camera will be used for both marketing and staff management as outlined in the Operations Manual.

Training

Prior to opening your All American Pet Resorts® Business, we will provide an initial training program lasting a minimum of 20 days depending on your level of experience in the industry and as we determine in our sole discretion. Additionally, we require that you complete two to four weeks of onsite, hands-on training. The duration of this training will be based on your ability to pass any of our written testing so utilized as well as perform any tasks required of franchisees. Training may be split between our corporate office, your Pet Resort, and a Pet Resort of our choice, as explained in the chart below. We hold training programs on an as needed basis. Under the Franchise Agreement, before you begin operating your All American Pet Resorts® Business, you or your "Designated Business Manager" must attend and successfully complete to our satisfaction our initial training program. You may have 1 additional person attend the initial training program at no additional training fee. If you desire to have additional people attend the initial training program, there will be a \$100 per day per person training fee for such additional people. We do not pay any travel expenses, lodging, meals, ground transportation or other personal expenses for any person attending the initial training program. If the Designated Business Manager's employment with you terminates, you must designate a new Designated Business Manager, who must successfully complete our initial training program within 3 months after the termination of the initial Designated Business Manager. If we do not hold an initial training program during that 90-day period, the replacement Designated Business Manager must attend and successfully complete the first available initial training program held by us. You may be charged a training fee for a replacement Designated Business Manager and you are responsible for the costs for airfare, ground transportation, lodging, meals, personal expenses, and the Designated

Business Manager's salary and benefits. Our training program consists of approximately, 35 hours of classroom training and 205 - 285 hours of on the job training as follows:

TRAINING PROGRAM

	Hours of		
	Classroom	Hours Of On The	
Subject	Training	Job Training	Location
Pet Resort Design Development	4	3	Headquarters, All
			American Pet Resorts of
			our choice, your location,
			in person, or via
			telephone.
Resort Set Up, Work Flow Analysis &	2	4	Headquarters, All
Design			American Pet Resorts of
			our choice, your location,
			in person, or via
			telephone.
Marketing Training	2	2	Headquarters, All
			American Pet Resorts of
			our choice, and your
			location.
IT Training	2	4	Headquarters, All
			American Pet Resorts of
			our choice, and your
			location.
Staffing Training	2	4	Headquarters, All
			American Pet Resorts of
			our choice, and your
D ' DI '	2	2	location.
Business Planning	2	2	Headquarters, All
			American Pet Resorts of
			our choice, and your
Haalth & Cafaty	2	2	location. Headquarters, All
Health & Safety	2	Δ	American Pet Resorts of
			our choice, and your
			location.
POS Training	2	4	Headquarters, All
1 OS Training	2	4	American Pet Resorts of
			our choice, and your
			location.
Customer Service Training	2	6	Headquarters, All
Constituting Training			American Pet Resorts of
			our choice, and your
			location.
Sales Training	1	2	Headquarters, All
			American Pet Resorts of
			our choice, and your
			location.

	Hours of	H Of O Th	
Cubic at	Classroom	Hours Of On The	
Subject Variations	Training 4	Job Training	All American Pet Resorts
Kennel Operations	4	24	of our choice, and your
			location.
Basic Pet Care	2	24	All American Pet Resorts
Dasic Fet Care	2	24	of our choice, and your
			location.
Advanced Pet Care	2	16	All American Pet Resorts
Advanced Fet Care	2	10	of our choice, and your
			location.
Business Operations Training	2	16	All American Pet Resorts
Business Operations Training	2	10	of our choice, and your
			location.
Management Training	2	8	All American Pet Resorts
Wanagement Training	2	0	of our choice, and your
			location.
Basic Site Management	1	2	Headquarters, All
Basic Site Management	1	2	American Pet Resorts of
			our choice, and your
			location.
Financial Training – Revenue	1	2	Headquarters, All
Reporting & Tracking	1	_	American Pet Resorts of
The porting to Trucking			our choice, and your
			location.
Onsite Hands-On Training	0	80-160	Headquarters, All
			American Pet Resorts of
			our choice, and your
			location.
Total	35	205-285	

The training materials consist of our Operations Manual and other course books, certifications, classroom instruction, onsite training, tests, quizzes, written and practical examinations. and/or. All American Pet representatives and/or other employees of participating All American Pet Resorts® will conduct the initial training program and other on-going training. Art Rimbold has designed, developed, owned and operated kennels. He is also the past president of a Michigan Humane Society. Yolanda Rimbold has trained groomers and groomed dogs. She has owned and operated daycares and kennels. Together they have 80 years' experience in the industry. We may change or substitute training personnel as necessary, and we may delegate our duties and share our responsibilities with regard to training. For example, our team of All American Pet Representative may assist with training. This team consists of individuals each having at least 1 year of experience in the style of pet care we require of our franchisees. You and/or your Designated Business Manager must pass all training to our sole satisfaction.

In addition to the initial training program described above, if at any point we determine, in our sole discretion, that you could benefit from additional training, we will require you to complete such training on a mutually agreed upon timeline. You and/or your Designated Business Manager must complete all additional training to our sole satisfaction and at your cost.

We may hold an annual or biennial franchise convention, which you will be required to attend unless you receive express written consent from us not to attend the convention. We may also hold additional mandatory conferences to discuss sales techniques, new Service and Product developments, training, bookkeeping, accounting, performance standards, advertising programs, merchandising procedures and other topics. You must pay any convention or conference fee we designate and all personal travel and living expenses. The annual franchise convention and any additional mandatory annual conferences may be held at a location chosen by us.

We reserve the right to require you to comply with reasonable restrictions on maximum and minimum prices (to the extent permitted by applicable law) of specific goods, programs or services offered and sold by the Franchised Business as required in the Manual or as we otherwise reasonably direct in writing from time to time.

ITEM 12 TERRITORY

Franchise Agreement

The franchise grants you the right to operate your All American Pets Resort Business and sell the Products and Services you are authorized to sell by the Franchise Agreement from the approved location within the non-exclusive territory granted to you ("**Territory**"). Your Territory will be based on a radial parameter, geographical boundaries related to highways, zip codes, demographics and other characteristics including population density, pet MRI, consumer demographics, segmentation, average income and other characteristics of the surrounding area, natural boundaries, extent of competition, the size of your All American Pet Resorts® Business and the amount and size of urban, suburban and rural areas.

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

We will grant only one franchise per Territory. Your Territory will be determined after we approve the location for your All American Pet Resorts® Business. We will use the most recent population information available in the U.S. Census Data, or other population statistical sources of our choosing to determine populations. We have the exclusive right to determine the boundaries of your Territory. You may not change the location of your Business or your Territory without our written consent before making the change. We have the right to approve or deny the relocation of your Pet Resorts based on the following factors: whether the Business is located in your Territory; the condition of the premises, including the size of the building and the adjoining land; the number of suites; and whether it is located sufficiently far enough away from any outlet of another All American Pet Resorts® franchisee as we determine. Under your Franchise Agreement, you do not receive any options for additional franchises, any rights of first refusal to acquire additional franchises, or any similar rights to buy additional franchises. Continuation of your Territory does not depend on the achievement of a certain sales volume, market penetration, or any other contingency. There are no circumstances that permit us to modify your territorial rights.

Customers from your Territory may purchase Services and Products from us and our Affiliate or designees over the Internet or in other reserved channels of distribution. We are not required to pay you any compensation if customers inside your Territory purchase Services and Products from us or our Affiliate or designees. Unless otherwise approved by us, you do not have the right to solicit or accept orders from customers outside of your Territory (including through the Internet, social media or other forms of electronic communication). You may not solicit or accept orders from customers located in another All American Pet Resorts® franchisee's territory.

We reserve the right, among others:

- 1. to use, and license others to use, the Marks and System for the operation of All American Pet Resorts[®] Businesses at any location outside of the Territory, regardless of the proximity to your Territory;
- 2. to use the Marks and the System to sell any Products or Services, similar to those which you will sell, through any alternate channels of distribution within or outside of the Territory. This includes, but is not limited to, retail locations and other channels of distribution such as television, mail order, catalog sales, wholesalers, retail outlets or other distribution outlets, or over the Internet. We exclusively reserve the Internet as a channel of distribution for us, and you may not independently market on the Internet, use any domain name, address, locator, link, metatag or search technique with words or symbols similar to the Marks or conduct e-commerce. We exclusively reserve the right to use and own any websites utilizing a domain name incorporating one or more of the words "American," "Pet" and/or "Resorts" or similar derivatives thereof. If we use the Marks to sell the Products and Services that you are required to sell to a customer in your Territory over the Internet we or our supplier and manufacturers we are under no obligation to provide you with a credit;
- 3. to use and license the use of other proprietary and non-proprietary marks, service marks or methods which are not the same as or confusingly similar to the Marks, whether in alternate channels of distribution or in the operation of a business offering professional pet boarding, daycare and grooming services and related products, at any location, including within the Territory, which may be the same as, similar to or different from the business operated by you;
- 4. to use the Marks and System in connection with the provisions of other services and products or in alternate channels of distribution at any location including within the Territory;
- 5. to purchase or be purchased by, or merge or combine with, any business, including a business that competes directly with your All American Pet Resorts® Business, wherever located;
- 6. to acquire businesses that are the same as or similar to the All American Pet Resorts[®] Businesses and operate such businesses regardless of where such businesses are located, including inside the Territory;
- 7. convert to the System operated by us any businesses offering professional pet boarding, daycare and grooming services and related products, including such businesses operated by competitors or otherwise operated independently or as part of, or in association with, any other system or chain, whether franchised or corporately owned and whether located inside or outside of the Territory; and

8. to implement multi-area marketing programs which may allow us or others to solicit or sell to customers anywhere. We also reserve the right to issue mandatory policies to coordinate such multi-area marketing programs.

We are not required to pay you any compensation if we exercise any of the rights noted above.

The territorial rights granted to you under the Franchise Agreement are not dependent on your achievement of a certain sales volume, market penetration or other contingency and your Territory may not be altered, unless you fail to adhere to or comply with the Franchise Agreement, in which case we will have the right to terminate the Franchise Agreement. Also, we do not grant rights of first refusal.

To the extent that any other pet boarding, daycare and grooming businesses or Pet Resort we or our Affiliate(s) operate may be located within the same market as your All American Pet Resorts[®] Business, like all other similar businesses, they will compete for customers by offering pet boarding, daycare and grooming services.

ITEM 13 TRADEMARKS

The Franchise Agreement grants you the nonexclusive right to use our Marks, including the service mark "All American Pet Resorts®", and various designs and logos associated with our services. You may also use our other current or future Marks as we may designate to operate your All American Pet Resorts® Business.

We have a trademark and filed all required affidavits on our principal Mark with the United States Patent and Trademark Office ("**USPTO**") as follows:

Mark	Registration Date	Registration No.	Register
ALL AMERICAN PET RESORTS	April 14, 2009	3,607,712	Principal Register
	August 11, 2009	3,666,368	Principal Register
ALL AMERICAN PET RESORTSBECAUSE YOU CARE	April 19, 2011	3,948,259	Principal Register
	September 27, 2016	5,047,854	Principal Register

Mark	Registration Date	Registration No.	Register		
ALL AMERIC N PET RESORTS* = EST 2005 =	June 9, 2020	6076139	Principal Register		
ALL AMERIC#N PET RESORTS	June 9, 2020	6076138	Principal Register		
ALL AMERICAN PET RESORTS EST 2005	June 9, 2020	6076137	Principal Register		
ALL AMERIC N PET RESORTS = EST 2005 = "Because You Care"	Pending	97061970 (Serial Number)	Pending		
BECAUSE YOU CARE	Pending	97061760 (Serial Number)	Pending		
ALL AMERIC N PET RESORTS "Because You Care"	Pending	97061801 (Serial Number)	Pending		
BECAUSE YOU CARE WE C.A.R.E.	Pending	98321239 (Serial Number)	Pending		
BECAUSE YOU CAREWE CARE	Pending	98321218 (Serial Number)	Pending		
CARE = Creating A Remarkable Experience =	Pending	98321266 (Serial Number)	Pending		
ALL AMERICAN PET RESORTS	Pending	98531597 (Serial Number)	Pending		

You must follow our rules when you use our Marks. You may not use any of the Marks alone or with modifying words, designs or symbols as part of a corporate or business name or in

any form on the Internet or through social media, including but not limited to URLS, domain names, e-mail addresses, locators, links, metatags or search techniques. You must get our prior written approval of your company name before you file any registration documents. Your trade name must follow these guidelines: the words "All American Pet Resorts" followed by a geographic description such as city, state, area, county or similar descriptor as is legally permissible. You must indicate, as required in the Franchise Agreement and specified in the Operations Manual, that you are an independent operator. Guidelines regarding proper trademark use and notices are in the Operations Manual and will be updated annually as we deem appropriate. You may not use our Marks with an unauthorized product or service, or in a manner not authorized in writing by us.

There are no currently effective material determinations of the USPTO, the Trademark Trial and Appeal Board, the trademark administrator of any state or any court, any pending infringement, opposition or cancellation proceedings or any pending material litigation involving any of our Marks which are relevant to the use of these Marks. No currently effective litigation affects our use or ownership rights in any Mark. No currently effective agreement limits our right to use or license the use of our Marks.

We are not currently involved in any pending litigation regarding our trademarks.

You must notify us within three (3) days after receiving notice of any claim, demand or cause of action based upon or arising from any attempt by any other person, firm or corporation to use the Marks or any colorable imitation thereof or the Copyrighted Materials. Upon receipt of timely notice of an action, claim or demand against you relating to the Marks or Copyrighted Materials, we have the sole right, but not the duty, to defend any such action. We have the exclusive right to contest or bring action against any third party regarding the third party's use of any of the Marks or Copyrighted Materials and will exercise such right as we deem appropriate. We will control all actions but not be obligated to take any action. In any defense or prosecution of any litigation relating to the Marks, Copyrighted Materials or components of the System undertaken by us, you must cooperate with us, sign any and all documents, and take all actions as may be desirable or necessary in the opinion of our counsel, to carry out such defense or prosecution. At our option, you will join in any action, in which case we will bear all your out-of-pocket costs for such participation.

You must modify or discontinue the use of a Mark if we modify or discontinue the Mark. If this happens, it is your responsibility to pay for converting your operations to reflect any new or modified logo as we may designate (for example, changing signs, letterhead, and business cards). You must not directly or indirectly contest our right to our trademarks, trade secrets or business techniques that are part of our business.

We do not know of any infringing uses that could materially affect your use of our Marks. You should understand that there could be other businesses using trademarks, trade names, or other commercial symbols similar to our Marks with superior rights to our rights. Before opening your All American Pet Resorts[®] Business, you should research this possibility, using telephone directories, trade directories, Internet directories, or otherwise in order to avoid the possibility of having to change the name of your All American Pet Resorts[®] Business.

ITEM 14 PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

The information contained in the Operations Manual is proprietary and is protected by copyright and other laws. The designs contained in the Marks, the layout of our advertising materials, the content and format of our Products, as well as any other writings, recordings in print or electronic form are also protected by copyright and other laws. Although we have not filed an application for copyright registration for the Operations Manual, the Marks, the advertising materials, the content and format of our Products, or any other writings and recordings, we claim common law and federal copyrights in these items. We grant you the right to use this proprietary and copyrighted information ("Copyrighted Works") in connection with your operation of your All American Pet Resorts® Business, but such copyrights remain our sole property.

There are currently no effective determinations of the United States Copyright Office or any court regarding any Copyrighted Works of ours, nor are any proceedings pending, nor are there any currently effective agreements between us and third parties pertaining to the Copyrighted Works that will or may significantly limit your use of our Copyrighted Works.

Our Operations Manual, electronic information and communications, sales and promotional materials, the development and use of our System, standards, specifications, policies, procedures, information, concepts and systems on, knowledge of and experience in the development, operation and franchising of All American Pet Resorts® Businesses, formulations for and packaging of Products, and training techniques used to provide Services sold at All American Pet Resorts® Businesses, information concerning Product and Service sales, operating results, financial performance and other financial data of All American Pet Resorts® Businesses and other related materials are proprietary and confidential ("Confidential Information") and are considered to be our property to be used by you only as described in the Franchise Agreement or the Operations Manual. Where appropriate, certain information has also been identified as trade secrets ("Trade Secrets"). You must maintain the confidentiality of our Confidential Information and Trade Secrets and adopt reasonable procedures to prevent unauthorized disclosure of our Trade Secrets and Confidential Information.

We will disclose parts of the Confidential Information and Trade Secrets to you as we deem necessary or advisable for the development of your All American Pet Resorts[®] Business during training and in guidance and assistance furnished to you under the Franchise Agreement, and you may learn or obtain from us additional Confidential Information and Trade Secrets during the term of the Franchise Agreement. The Confidential Information and Trade Secrets are valuable assets of ours and are disclosed to you on the condition that you, and your owners, if you are a business entity, and employees agree to maintain the information in confidence by entering into a confidentiality agreement that we can enforce. Nothing contained in the Franchise Agreement will be construed to prohibit you from using the Confidential Information or Trade Secrets in the operation of other All American Pet Resorts[®] Business during the term of the Franchise Agreement.

No patents are material to us at this time. We currently do not have any pending patent applications.

You must notify us within 3 days after receiving notice of any claim, demand or cause of action based upon or arising from any attempt by any other person, firm or corporation to use the Copyrighted Works. Upon receipt of timely notice of an action, claim or demand against you relating to the Copyrighted Works, we have the sole right, but not the duty, to defend any such action. We have the exclusive right to contest or bring action against any third party regarding the third party's use of any of the Copyrighted Works and will exercise such right as we deem appropriate. We will control all actions but not be obligated to take any action. In any defense or prosecution of any litigation relating to the Copyrighted Works undertaken by us, you must cooperate with us, sign any and all documents, and take all actions as may be desirable or necessary in the opinion of our counsel, to carry out such defense or prosecution. At our option, you will join in any action, in which case we will bear all your out-of-pocket costs for such participation. You must modify or discontinue the use of the Copyrighted Works if we modify or discontinue use of the Copyrighted Works. If this happens, we will reimburse you for your tangible out of pocket cost of compliance.

ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

If you are an individual, you may directly supervise the All American Pet Resorts® Business at your Pet Resort or you may hire a Designated Business Manager to perform the supervision. A Designated Business Manager is an individual who will have day-to-day management of the Business, exercise on-site supervision, and personally participate in the direct operation of the Business. If you are a business entity, your Designated Business Manager must perform the direct, on-site supervision.

If we believe you lack sufficient business experience, you must designate a Designated Business Manager to act as the operating manager for your All American Pet Resorts[®] Business. The Designated Business Manager must attend and successfully complete the initial training program and must abide by the obligations in the Franchise Agreement and the Operations Manual.

If you are a legal or business entity, each individual who owns, directly or indirectly, any interest in you (and, if you are an individual, your immediate family defined as your spouse and any adult children involved in any way with the All American Pet Resorts[®] Business) must sign the Guaranty and Assumption of Franchisee's Obligations assuming and agreeing to discharge all of your obligations and comply with all restrictions under the Franchise Agreement (Attachment D to the Franchise Agreement) and our Non-Disclosure and Non-Competition Agreement attached to this Disclosure Document as Exhibit G. If you have an Affiliated Lessor as defined in the Franchise Agreement, they must also sign the Guaranty and Assumption of Franchisee's Obligations assuming and agreeing to discharge all of your obligations and comply with all restrictions under the Franchise Agreement (Attachment D-1 to the Franchise Agreement) and our Non-Disclosure and Non-Competition Agreement attached to this Disclosure Document as Exhibit G.

All individuals who own an equity interest in an entity operating an All American Pet Resorts® Business must sign the Franchise Agreement.

If you or your Designated Business Manager are unable to operate the All American Pet Resorts[®] Business because of your or your Designated Business Manager's death or disability or under certain circumstances where you or your Designated Business Manager are unable to devote your full time and attention to the All American Pet Resorts[®] Business, you must notify us immediately. We can provide management services to assist you to operate your All American Pet Resorts[®] Business.

ITEM 16 RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

You must refrain from using or permitting the use of your All American Pet Resorts[®] Business for any other purpose or activity at any time without first obtaining our written consent.

You must sell or offer for sale only those Services and Products which are authorized by us and which meet our standards and specifications. You must follow our policies, procedures, methods, and techniques. You must sell or offer for sale all types of services and products specified by us. We may change or add to our required Services and Products as we deem appropriate with prior notice to you. You must discontinue selling and offering for sale any Services or Products, which we may disapprove in writing at any time. We reserve the right to establish maximum resale prices for use with multi-area marketing programs and special price promotions. You may not sell products or services to customers located outside your Territory (including through the Internet) without our prior written approval.

ITEM 17 RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this disclosure document.

		Section in franchise or	
	Provision	other agreement	Summary
a.	Length of the franchise term	Section 4	The initial term will commence on the date you
			sign the Franchise Agreement and continue
			through the Operating Term of 10 years.
b.	Renewal or extension of the		If you are in good standing, you can add unlimited
	term		additional terms of 10 years each.

		Section in	
	Provision	franchise or	Summow.
		other agreement	Summary
c.	Requirements for franchisee	Section 4	Sign Successor Franchise Agreement, which may
	to renew or extend		have materially different terms than the Franchise
			Agreement, be current in payments, sign release.
			Pay renewal fee.
			If you seek to renew your franchise at the
			expiration of the initial term or any renewal term,
			you may be asked to sign a new franchise
			agreement that contains terms and conditions
			materially different from those in your previous
			franchise agreement, such as different fee
			requirements and territorial rights.
d.	Termination by franchisee	Section 18	Termination by you is a termination without cause
	·		and a breach of the Franchise Agreement.
	Termination by franchisor without cause	Not Applicable	Not applicable.
f.	Termination by franchisor	Section 18	Can terminate upon certain violations of the
	with cause		Franchise Agreement by you.
g.	"Cause" defined – curable	Section 18	You have 30 days to cure the default listed in
	defaults		Section 18.2.
h.	"Cause" defined – non-	Section 18	Non-curable defaults: the defaults listed in Section
	curable defaults		18.1
i.	Franchisee's obligations on	Sections 11, 13, 15	_
	termination/non-renewal	& 18	payment of amounts due, compliance with post-
			term non-compete obligations and return of
			Operations Manual, all Confidential Information,
_			Trade Secrets and records.
j.	Assignment of contract by franchisor	Section 16	No restriction on our right to assign
k.	"Transfer" by franchisee –	Section 16	Includes transfer of contract, equity or assets of the
	defined		All American Pet Resorts® Business or ownership
-	F1:	G4: 16	change
1.	Franchisor approval of	Section 16	We have the right to approve in writing all
	transfer by franchisee		transfers, including but not limited to any and all
			transfers of shares of interest in your All American
	Conditions for franchisor's	Section 16	Pet Resorts [®] . New franchisee qualifies, Transfer Fee paid by
1111.	approval of transfer	Section 10	you, purchase agreement approved, training
	approvar of transfer		arranged, release signed by you and current
			Franchise Agreement signed by new franchisee,
			commercial building inspection.
n	Franchisor's right of first	Section 17	We can match any offer for your business.
11.	refusal to acquire	Section 17	The can materially offer for your outliness.
	franchisee's business		
<u> </u>	Transmisse a dualities		

	Provision	Section in franchise or other agreement	Summary
о.	Franchisor's option to purchase franchisee's business	Section 17	We may, but are not required to, purchase your inventory and equipment at fair market value if your Franchise is terminated for any reason.
p.	Death or disability of franchisee	Section 16	Your estate or legal representative must apply to us for the right to transfer to the next of kin within 120 days.
q.	Non-competition covenants during the term of the franchise	Section 15	No involvement in competing business anywhere in US
r.	Non-competition covenants after the franchise is terminated or expires	Sections 15	No competing business for 3 years (i) in the Territory or any other Franchisee's Territory; (ii) 100 miles of the Territory or any other Franchisee's Territory or (iii) 100 miles of any of our Affiliate owned All American Pet Resorts® Business
s.	Modification of agreement	Sections 4 & 21	No modifications of Franchise Agreement during term generally, but Operations Manual subject to change. Modifications permitted upon the extension of your rights to separate All American Pet Resorts® Business.
t.	Integration/merger clause	Section 21	Only the terms of the Franchise Agreement are binding (subject to state law). Any representations or promises outside of the disclosure document or franchise agreement may not be enforceable.
u.	Dispute resolution by arbitration or mediation	Section 20	Except for certain claims, all disputes must be arbitrated in Michigan, under the CPR Rules for Non-Administered Arbitration (subject to state law).
v.	Choice of forum	Section 21	Arbitration or litigation must be in Michigan (subject to state law).
W.	Choice of law	Section 21	Michigan law applies, except as provided in a State Specific Addendum.

ITEM 18 PUBLIC FIGURES

We do not currently use any public figure to promote our franchise. We do use local personalities to promote All American Pet Resorts® but not to promote the purchase of a franchise.

ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only

if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

Section I of this Item 19 reflects the average revenue generated by the 9 All American Pet Resorts® businesses (each, a "Business" and collectively, the "Businesses"), that were open and in operation for the full year of each year for the time period beginning January 1, 2019 through December 31, 2023 (the "Measurement Period") (with the exception of temporary closures in 2020 caused by the COVID-19 pandemic). One of the nine Businesses is operated by our affiliate but it is not materially different from franchised businesses. The information included in Section I was gathered from information provided to us by each franchisee and affiliate from their respective annual financial reports, which were provided to us by the franchisees and affiliate or their designated accounting representative.

Section II of this Item 19 reflects the resort revenue history for 2019 through 2023 by 9 Businesses that were open and in operation for the full year of each year for the Measurement Period (with the exception of temporary closures in 2020 caused by the COVID-19 pandemic). The information included in Section II below was gathered from information provided to us by each franchisee and affiliate from their respective financial records including monthly revenue reports and annual financial reports, which were provided to us by the franchisees and affiliate or their designated accounting representative.

Section III of this Item 19 reflects the distribution of the different types of services rendered in 2019 through 2020 at 5 of the Businesses, in 2021 at 8 of the Businesses, in 2022 and 2023 for the 9 Businesses that were open and in operation for the full year of each year in the Measurement Period (with the exception of temporary closures in 2020 caused by the COVID-19 pandemic) and use operations software that allows tracking of separate service types. The information included in Section III below was gathered from information provided to us by the franchisees from their respective annual financial reports, which were provided to us by the franchisees and affiliate or their designated accounting representative.

Section IV of this Item 19 reflects an estimated earnings history before interest, taxes, depreciation, and amortization ("EBITDA") for 2019, 2020, 2021, 2022, and 2023 for each of the 9 Businesses that were open and in operation for at least one full year during the Measurement Period, adjusted to reflect our current standard Royalty Fee and Brand Fund Fee rates for new franchisees joining the All American Pet Resorts[®] System after the date of this FDD. The information included in Section IV below was gathered from information provided to us by each franchisee and affiliate from their respective annual financial reports, which were provided to us by the franchisees and affiliate or their designated accounting representative.

Section V of this Item 19 reflects an estimated earnings before interest, taxes, depreciation, and amortization ("EBITDA") for 2023 for each of the 9 Businesses that were open and in operation for one full year during the Measurement Period, adjusted to reflect our current standard Royalty Fee and Brand Fund Fee rates for new franchisees joining the All American Pet Resorts® System after the date of this FDD. The information included in Section V below was gathered from information provided to us by each franchisee and affiliate from their respective annual

financial reports, which were provided to us by the franchisees and affiliate or their designated accounting representative.

Section VI of this Item 19 reflects average earnings before interest, taxes, depreciation, and amortization ("EBITDA") for 2019 to 2023 for each of the 9 Businesses that were open and in operation for at least one full year during the Measurement Period, adjusted to reflect our current standard Royalty Fee and Brand Fund Fee rates for new franchisees joining the All American Pet Resorts[®] System after the date of this FDD. The information included in Section VI below was gathered from information provided to us by each franchisee and affiliate from their respective annual financial reports, which were provided to us by the franchisees and affiliate or their designated accounting representative.

Section VII of this Item 19 reflects average earnings before interest, taxes, depreciation, and amortization ("EBITDA") for 2019 to 2023 by percent of sales for each of the 9 Businesses that were open and in operation for at least one full year during the Measurement Period, adjusted to reflect our current standard Royalty Fee and Brand Fund Fee rates for new franchisees joining the All American Pet Resorts[®] System after the date of this FDD. The information included in Section VII below was gathered from information provided to us by each franchisee and affiliate from their respective annual financial reports, which were provided to us by the franchisees and affiliate or their designated accounting representative.

Sections VIII, IX, X, and XI of this Item 19 reflect actual earnings before interest, taxes, depreciation, and amortization ("EBITDA") for 2022, 2021, 2020, and 2019 respectively for each of the 9 Businesses that were open and in operation for at least one full year during the Measurement Period, adjusted to reflect our current standard Royalty Fee and Brand Fund Fee rates for new franchisees joining the All American Pet Resorts® System after the date of this FDD. The information included in Sections VIII, IX, X, and XI below was gathered from information provided to us by each franchisee and affiliate from their respective annual financial reports, which were provided to us by the franchisees and affiliate or their designated accounting representative.

Written substantiation for the financial performance representation will be made available to the prospective franchisee upon reasonable request. The characteristics of the Franchised Businesses and affiliate included in this Item 19 may differ materially from the outlet that will be offered to you.

I. Average Resort Revenue per Year

	2019	2020	2021	2022	2023
Average*	\$1,319,967	\$818,676	\$1,371,375	\$1,585,976	\$1,730,288
Median	\$1,089,278	\$595,538	\$1,016,063	\$1,332,509	\$1,449,613
Highest	\$2,347,467	\$1,389,209	\$2,297,367	\$2,917,251	\$2,854,183
Lowest	\$644,933	\$438,169	\$816,059	\$912,276	\$1,010,394

Notes:

1. *One of the outlets included in the calculations is a non-franchised, affiliate operated location.

II. Resort Revenue History

Outlet	Date Opened	No. of Suites ¹	2019	2020	2021	2022	2023
1**	Feb. 2005	162	\$2,143,953	\$1,389,209	\$2,297,367	\$2,480,854	\$2,784,996
2	June 2008	293	\$2,347,467	\$1,343,659	\$2,203,672	\$2,917,251	\$2,854,183
3	Sept. 2008	67	\$862,692	\$533,151	\$816,059	\$912,276	\$1,010,394
4	Jan. 2010	132	\$644,933	\$438,169	\$845,875	\$1,006,542	\$1,135,228
5	Nov. 2012	80	\$947,464	\$589,560	\$970,872	\$1,050,164	\$1,219,304
6	Jan. 2013	77	\$953,643	\$445,710	\$955,819	\$1,332,509	\$1,353,871
7	Mar. 2013	113	\$1,751,194	\$1,194,790	\$1,751,036	\$1,762,300	\$2,102,554
8	Dec. 2013	100	\$1,089,278	\$595,538	\$1,016,063	\$1,242,351	\$1,449,613
9	Oct. 2017	112	\$1,139,083	\$838,299	\$1,485,613	\$1,569,541	\$1,662,446

Notes:

- 1. Sales reported by Franchised Business owners and affiliate.
- 2. **Outlet No. 1 is a non-franchised, affiliate operated location.

III. Service Revenue Distribution

	2019	2020	2021	2022	2023
Boarding	71.1%	68.1%	68.2%	73.7%	72.9%
Daycare	14.8%	18.2%	17.4%	15.7%	13.5%
Other	14.1%	13.7%	14.4%	10.6%	13.6%

- 1. This distribution percentage is based on 5 locations for 2019-2020, 8 locations for 2021, and 9 locations for 2022 2023 that utilize a specific operations software. Not all units are included.
- 2. Daycare revenue includes revenue associated with daycare packages.
- 3. The average boarding per stay is approximately 4 nights.

IV. Resort EBITDA History

Outlet	2019 EBITDA	2020 EBITDA	2021 EBITDA	2022 EBITDA	2023 EBITDA
1	\$396,170	\$52,560	\$561,092	\$574,825	\$759,987
2	\$475,167	(\$152,269)	\$453,018	\$850,842	\$465,147
3	\$153,967	\$30,503	\$105,131	\$185,469	\$85,933
4	(\$4,911)	(\$90,304)	\$(91,973)	\$(17,712)	\$118,329
5	\$216,113	\$85,771	\$188,056	\$194,222	\$212,909
6	\$300,360	(\$24,757)	\$264,078	\$415,365	\$288,372
7	\$525,671	\$22,525	\$332,837	\$548,776	\$524,848
8	(\$46,462)	(\$118,127)	\$89,203	\$114,219	\$138,133
9	\$137,784	(\$76,183)	\$(98,589)	\$243,109	\$161,484

- 1. Total Gross Sales represent revenue that royalties are paid on and as defined in Item 6 "Other Fees" Note (1).
- 2. Cost of Goods Sold includes grooming supplies, kennel supplies & expenses, general supplies, and price adjustments.
- 3. Payroll Expenses include direct/indirect wages, employee benefits, payroll taxes, payroll expenses. This does not include any expenses related to owner compensation/distribution.
- 4. Rent expense includes the cost of the rent payment only.
- 5. Fixed Expenses include expenses associated with advertising, office expenses, professional fees & uniforms.
- 6. Property Expenses include security, utilities, building maintenance, building repairs, telephone, internet, water & sewer expenses related to the property.
- 7. Variable Expenses include expenses associated with bank & credit card fees, insurance, personal property taxes, real estate taxes, licenses, permits, penalties, fines & workers compensation.
- 8. Royalty Fees collected from these locations are not uniform. The Royalty Fee figures reflected in the chart above have been adjusted to reflect a standard 7% royalty rate.
- 9. Brand Fund Fees collected from these locations are not uniform. The Brand Fund Fee figures reflected in the chart above have been adjusted to reflect a standard 2% brand fund rate.

V. 2023 Resort EBITDA

Outlet	1	2	3	4	5	6	7	8	9
Total Gross Sales1	\$2,784,996	\$2,854,183	\$1,010,394	\$1,135,228	\$1,219,304	\$1,353,871	\$2,102,554	\$1,449,613	\$1,662,446
Cost of Goods Sold2	\$73,358	\$203,316	\$27,228	\$52,279	\$40,358	\$60,745	\$47,531	\$123,223	\$94,850
Payroll Expenses3	\$1,010,692	\$1,051,478	\$578,744	\$568,171	\$455,644	\$478,366	\$683,763	\$685,089	\$834,022
Rent Expense4	\$300,000	\$0	\$61,632	\$64,200	\$166,500	\$102,086	\$330,000	\$185,000	\$72,000
Fixed Expenses5	\$159,214	\$291,585	\$36,225	\$96,199	\$86,590	\$140,898	\$65,161	\$58,324	\$115,808
Property Expense6	\$98,297	\$122,389	\$72,504	\$71,956	\$57,246	\$76,477	\$111,422	\$60,112	\$116,663
Variable Expenses7	\$132,798	\$463,392	\$57,192	\$61,923	\$90,320	\$85,079	\$150,599	\$69,267	\$117,999
Royalty Fees8	\$194,950	\$199,793	\$70,728	\$79,466	\$85,351	\$94,771	\$147,179	\$101,473	\$116,371
Brand Fund Fees9	\$55,700	\$57,084	\$20,208	\$22,705	\$24,386	\$27,077	\$42,051	\$28,992	\$33,249
Total Expenses	\$2,025,009	\$2,389,036	\$924,460	\$1,016,899	\$1,006,395	\$1,065,499	\$1,577,706	\$1,311,480	\$1,500,962
Total % of Expenses	72.7%	83.7%	91.5%	89.6%	82.5%	78.7%	75.0%	90.5%	90.3%
EBITDA (\$)	\$759,987	\$465,147	\$85,933	\$118,329	\$212,909	\$288,372	\$524,848	\$138,133	\$161,484
EBITDA (%)	27.3%	16.3%	8.5%	10.4%	17.5%	21.3%	25.0%	9.5%	9.7%

- 1. Total Gross Sales represent revenue that royalties are paid on and as defined in Item 6 "Other Fees" Note (1).
- 2. Cost of Goods Sold includes grooming supplies, kennel supplies & expenses, general supplies, and price adjustments.
- 3. Payroll Expenses include direct/indirect wages, employee benefits, payroll taxes, payroll expenses. This does not include any expenses related to owner compensation/distribution.
- 4. Rent expense includes the cost of the rent payment only.
- 5. Fixed Expenses include expenses associated with advertising, office expenses, professional fees & uniforms.
- 6. Property Expenses include security, utilities, building maintenance, building repairs, telephone, internet, water & sewer expenses related to the property.
- 7. Variable Expenses include expenses associated with bank & credit card fees, insurance, personal property taxes, real estate taxes, licenses, permits, penalties, fines & workers compensation.
- 8. Royalty Fees collected from these locations are not uniform. The Royalty Fee figures reflected in the chart above have been adjusted to reflect a standard 7% royalty rate.

9. Brand Fund Fees collected from these locations are not uniform. The Brand Fund Fee figures reflected in the chart above have been adjusted to reflect a standard 2% brand fund rate.

VI. Average EBITDA by Resort for the Period 2019 through 2023

Outlet	1	2	3	4	5	6	7	8	9
Total Gross Sales ¹	\$2,219,276	\$2,333,246	\$826,914	\$814,149	\$955,473	\$1,008,310	\$1,712,375	\$1,078,569	\$1,338,996
Cost of Goods Sold ²	\$61,494	\$172,245	\$25,119	\$48,552	\$34,876	\$41,997	\$43,492	\$106,710	\$77,236
Payroll Expenses ³	\$895,702	\$810,860	\$441,197	\$459,545	\$344,944	\$308,264	\$579,994	\$456,004	\$696,144
Rent Expense ⁴	\$290,000	\$35,508	\$61,118	\$63,285	\$131,700	\$96,492	\$297,600	\$192,400	\$78,400
Fixed Expenses ⁵	\$85,642	\$214,562	\$27,709	\$84,684	\$59,198	\$96,645	\$71,116	\$71,546	\$105,431
Property Expense ⁶	\$87,610	\$130,896	\$43,182	\$64,377	\$54,080	\$61,175	\$77,230	\$58,727	\$95,284
Variable Expenses ⁷	\$130,167	\$340,802	\$41,967	\$37,749	\$65,268	\$64,305	\$97,898	\$60,717	\$92,470
Royalty Fee ⁸	\$155,349	\$163,327	\$57,884	\$56,990	\$66,883	\$70,582	\$119,866	\$75,500	\$93,730
Brand Fund Fee ⁹	\$44,386	\$46,665	\$16,538	\$16,283	\$19,109	\$20,166	\$34,247	\$21,571	\$26,780
Total Expenses	\$1,750,349	\$1,914,866	\$714,714	\$831,464	\$776,059	\$759,627	\$1,321,444	\$1,043,176	\$1,265,475
Total Expenses	78.9%	82.1%	86.4%	102.1%	81.2%	75.3%	77.2%	96.7%	94.5%
EBITDA (\$)	\$468,927	\$418,381	\$112,201	\$(17,314)	\$179,414	\$248,683	\$390,931	\$35,393	\$73,521
EBITDA (%)	21.1%	17.9%	13.6%	-2.1%	18.8%	24.7%	22.8%	3.3%	5.5%

- 1. Total Gross Sales represent revenue that royalties are paid on and as defined in Item 6 "Other Fees" Note (1).
- 2. Cost of Goods Sold includes grooming supplies, kennel supplies & expenses, general supplies, and price adjustments.
- 3. Payroll Expenses include direct/indirect wages, employee benefits, payroll taxes, payroll expenses. This does not include any expenses related to owner compensation/distribution.
- 4. Rent expense includes the cost of the rent payment only.
- 5. Fixed Expenses include expenses associated with advertising, office expenses, professional fees & uniforms.
- 6. Property Expenses include security, utilities, building maintenance, building repairs, telephone, internet, water & sewer expenses related to the property.

- 7. Variable Expenses include expenses associated with bank & credit card fees, insurance, personal property taxes, real estate taxes, licenses, permits, penalties, fines & workers compensation.
- 8. Royalty Fees collected from these locations are not uniform. The Royalty Fee figures reflected in the chart above have been adjusted to reflect a standard 7% royalty rate.
- 9. Brand Fund Fees collected from these locations are not uniform. The Brand Fund Fee figures reflected in the chart above have been adjusted to reflect a standard 2% brand fund rate.

VII. Average EBITDA by Resort for the Period 2019 through 2023 as a Percent of Sales

Outlet	1	2	3	4	5	6	7	8	9
Total Gross Sales ¹	\$2,219,276	\$2,333,246	\$826,914	\$814,149	\$955,473	\$1,008,310	\$1,712,375	\$1,078,569	\$1,338,996
Cost of Goods Sold ²	2.8%	7.4%	3.0%	6.0%	3.7%	4.2%	2.5%	9.9%	5.8%
Payroll Expenses ³	40.4%	34.8%	53.4%	56.4%	36.1%	30.6%	33.9%	42.3%	52.0%
Rent Expense ⁴	13.1%	1.5%	7.4%	7.8%	13.8%	9.6%	17.4%	17.8%	5.9%
Fixed Expenses ⁵	3.9%	9.2%	3.4%	10.4%	6.2%	9.6%	4.2%	6.6%	7.9%
Property Expense ⁶	3.9%	5.6%	5.2%	7.9%	5.7%	6.1%	4.5%	5.4%	7.1%
Variable Expenses ⁷	5.9%	14.6%	5.1%	4.6%	6.8%	6.4%	5.7%	5.6%	6.9%
Royalty Fee ⁸	7.0%	7.0%	7.0%	7.0%	7.0%	7.0%	7.0%	7.0%	7.0%
Brand Fund Fee ⁹	2.0%	2.0%	2.0%	2.0%	2.0%	2.0%	2.0%	2.0%	2.0%
Total Expenses	78.9%	82.1%	86.4%	102.1%	81.2%	75.3%	77.2%	96.7%	94.5%
EBITDA (%)	21.1%	17.9%	13.6%	-2.1%	18.8%	24.7%	22.8%	3.3%	5.5%

- 1. Total Gross Sales represent revenue that royalties are paid on and as defined in Item 6 "Other Fees" Note (1).
- 2. Cost of Goods Sold includes grooming supplies, kennel supplies & expenses, general supplies, and price adjustments.
- 3. Payroll Expenses include direct/indirect wages, employee benefits, payroll taxes, payroll expenses. This does not include any expenses related to owner compensation/distribution.
- 4. Rent expense includes the cost of the rent payment only.
- 5. Fixed Expenses include expenses associated with advertising, office expenses, professional fees & uniforms.
- 6. Property Expenses include security, utilities, building maintenance, building repairs, telephone, internet, water & sewer expenses related to the property.

- 7. Variable Expenses include expenses associated with bank & credit card fees, insurance, personal property taxes, real estate taxes, licenses, permits, penalties, fines & workers compensation.
- 8. Royalty Fees collected from these locations are not uniform. The Royalty Fee figures reflected in the chart above have been adjusted to reflect a standard 7% royalty rate.
- 9. Brand Fund Fees collected from these locations are not uniform. The Brand Fund Fee figures reflected in the chart above have been adjusted to reflect a standard 2% brand fund rate

VIII. 2022 Actual EBITDA by Resort

Outlet	1	2	3	4	5	6	7	8	9
Total Gross Sales ¹	\$2,480,854	\$2,917,251	\$912,276	\$1,006,542	\$1,050,164	\$1,332,509	\$1,762,300	\$1,242,351	\$1,569,541
Cost of Goods Sold ²	\$74,210	\$210,875	\$23,870	\$47,696	\$34,386	\$59,756	\$43,506	\$121,084	\$97,706
Payroll Expenses	\$994,097	\$815,068	\$435,005	\$617,352	\$382,616	\$396,279	\$580,317	\$551,819	\$691,356
Rent Expense ⁴	\$300,000	\$0	\$61,632	\$68,400	\$144,000	\$100,027	\$168,000	\$156,000	\$72,000
Fixed Expenses	\$71,899	\$193,176	\$27,702	\$100,628	\$60,774	\$95,532	\$102,031	\$54,556	\$110,566
Property Expense ⁶	\$95,351	\$150,723	\$47,684	\$61,432	\$63,751	\$75,967	\$79,676	\$68,781	\$92,807
Variable Expenses	\$147,195	\$434,014	\$48,809	\$38,157	\$75,901	\$69,657	\$81,387	\$64,080	\$120,738
Royalty Fee ⁸	\$173,660	\$204,208	\$63,859	\$70,458	\$73,511	\$93,276	\$123,361	\$86,965	\$109,868
Brand Fund Fee ⁹	\$49,617	\$58,345	\$18,246	\$20,131	\$21,003	\$26,650	\$35,246	\$24,847	\$31,391
Total Expenses	\$1,906,029	\$2,066,409	\$726,807	\$1,024,254	\$855,943	\$917,144	\$1,213,524	\$1,128,132	\$1,326,432
Total Expenses (%)	76.8%	70.8%	79.7%	101.8%	81.5%	68.8%	68.9%	90.8%	84.5%
EBITDA (\$)	\$574,825	\$850,842	\$185,469	\$(17,711)	\$194,221	\$415,365	\$548,776	\$114,219	\$243,109
EBITDA (%)	23.2%	29.2%	20.3%	-1.8%	18.5%	31.2%	31.1%	9.2%	15.5%

Notes:

1. Total Gross Sales represent revenue that royalties are paid on and as defined in Item 6 "Other Fees" Note (1).

- 2. Cost of Goods Sold includes grooming supplies, kennel supplies & expenses, general supplies, and price adjustments.
- 3. Payroll Expenses include direct/indirect wages, employee benefits, payroll taxes, payroll expenses. This does not include any expenses related to owner compensation/distribution.
- 4. Rent expense includes the cost of the rent payment only.
- 5. Fixed Expenses include expenses associated with advertising, office expenses, professional fees & uniforms.
- 6. Property Expenses include security, utilities, building maintenance, building repairs, telephone, internet, water & sewer expenses related to the property.
- 7. Variable Expenses include expenses associated with bank & credit card fees, insurance, personal property taxes, real estate taxes, licenses, permits, penalties, fines & workers compensation.
- 8. Royalty Fees collected from these locations are not uniform. The Royalty Fee figures reflected in the chart above have been adjusted to reflect a standard 7% royalty rate.
- 9. Brand Fund Fees collected from these locations are not uniform. The Brand Fund Fee figures reflected in the chart above have been adjusted to reflect a standard 2% brand fund rate.

IX. 2021 Actual EBITDA by Resort

Outlet	1	2	3	4	5	6	7	8	9
Total Gross Sales ¹	\$2,297,367	\$2,203,672	\$816,059	\$845,875	\$970,872	\$955,819	\$1,751,036	\$1,016,063	\$1,485,613
Cost of Goods Sold ²	\$57,474	\$136,594	\$38,116	\$54,426	\$35,360	\$31,578	\$47,581	\$100,753	\$81,560
Payroll Expenses ³	\$899,850	\$749,300	\$444,182	\$508,718	\$341,107	\$263,530	\$581,947	\$378,990	\$860,688
Rent Expense ⁴	\$300,000	\$0	\$59,064	\$68,400	\$144,000	\$95,146	\$408,000	\$156,000	\$191,000
Fixed Expenses ⁵	\$69,521	\$198,716	\$21,830	\$98,131	\$63,865	\$90,459	\$76,263	\$88,732	\$102,885
Property Expense ⁶	\$75,999	\$154,858	\$42,978	\$96,293	\$52,199	\$63,720	\$52,467	\$49,487	\$113,596
Variable Expenses ⁷	\$126,668	\$312,856	\$31,313	\$35,751	\$58,907	\$61,284	\$94,348	\$61,452	\$100,768
Royalty Fee ⁸	160,816	\$154,257	\$57,124	\$59,211	\$67,961	\$66,907	\$122,573	\$71,124	\$103,993
Brand Fund Fee ⁹	\$45,947	\$44,073	\$16,321	\$16,918	\$19,417	\$19,116	\$35,021	\$20,321	\$29,712
Total Expenses	\$1,736,275	\$1,750,654	\$710,928	\$937,848	\$782,816	\$691,741	\$1,418,199	\$926,860	\$1,584,202
Total Expenses	\$75.6%	79.4%	87.1%	110.9%	80.6%	72.4%	81.0%	91.2%	106.6%
EBITDA (\$)	\$561,092	\$453,018	\$105,131	\$(91,973)	\$188,056	\$264,078	\$332,837	\$89,203	\$(98,589)
EBITDA (%)	24.4%	20.6%	12.9%	-10.9%	19.4%	27.6%	19.0%	8.8%	-6.6%

- 1. Total Gross Sales represent revenue that royalties are paid on and as defined in Item 6 "Other Fees" Note (1).
- 2. Cost of Goods Sold includes grooming supplies, kennel supplies & expenses, general supplies, and price adjustments.
- 3. Payroll Expenses include direct/indirect wages, employee benefits, payroll taxes, payroll expenses. This does not include any expenses related to owner compensation/distribution.
- 4. Rent expense includes the cost of the rent payment only.
- 5. Fixed Expenses include expenses associated with advertising, office expenses, professional fees & uniforms.
- 6. Property Expenses include security, utilities, building maintenance, building repairs, telephone, internet, water & sewer expenses related to the property.

- 7. Variable Expenses include expenses associated with bank & credit card fees, insurance, personal property taxes, real estate taxes, licenses, permits, penalties, fines & workers compensation.
- 8. Royalty Fees collected from these locations are not uniform. The Royalty Fee figures reflected in the chart above have been adjusted to reflect a standard 7% royalty rate.
- 9. Brand Fund Fees collected from these locations are not uniform. The Brand Fund Fee figures reflected in the chart above have been adjusted to reflect a standard 2% brand fund rate.

X. 2020 Actual EBITDA by Resort

Outlet	1	2	3	4	5	6	7	8	9
Total Gross Sales ¹	\$1,389,209	\$1,343,65 9	\$533,151	\$438,169	\$589,560	\$445,710	\$1,194,790	\$595,538	\$838,299
Cost of Goods Sold ²	\$43,492	\$224,245	\$11,669	\$30,125	\$34,848	\$21,486	\$28,039	\$64,802	\$48,606
Payroll Expenses ³	\$681,526	\$682,744	\$305,754	\$264,408	\$227,070	\$154,278	\$459,443	\$246,332	\$526,156
Rent Expense ⁴	\$250,000	\$11,105	\$61,632	\$51,300	\$60,000	\$93,816	\$354,000	\$153,000	\$36,000
Fixed Expenses ⁵	\$62,628	\$113,034	\$24,849	\$83,207	\$39,545	\$75,995	\$59,323	\$86,569	\$105,553
Property Expense ⁶	\$73,130	\$53,039	\$23,791	\$38,611	\$41,609	\$40,028	\$79,817	\$57,332	\$64,557
Variable Expenses ⁷	\$100,844	\$290,832	\$26,969	\$21,387	\$47,657	\$44,750	\$84,112	\$52,032	\$58,163
Royalty Fee ⁸	\$97,245	\$94,056	\$37,321	\$30,672	\$41,269	\$31,200	\$83,635	\$41,688	\$58,681
Brand Fund Fee ⁹	\$27,784	\$26,873	\$10,663	\$8,763	\$11,791	\$8,914	\$23,896	\$11,911	\$16,766
Total Expenses	\$1,336,649	\$1,495,92 8	\$502,648	\$528,473	\$503,789	\$470,467	\$1,172,265	\$713,665	\$914,482
Total Expenses	96.2%	111.3%	94.3%	120.6%	85.5%	105.6%	98.1%	119.8%	109.1%
EBITDA (\$)	\$52,560	\$(152,269)	\$30,503	\$(90,304)	\$85,771	\$(24,757)	\$22,525	\$(118,127)	\$(76,183)
EBITDA (%)	3.8%	-11.3%	5.7%	-20.6%	14.5%	-5.6%	1.9%	-19.8%	-9.1%

- 1. Total Gross Sales represent revenue that royalties are paid on and as defined in Item 6 "Other Fees" Note (1).
- 2. Cost of Goods Sold includes grooming supplies, kennel supplies & expenses, general supplies, and price adjustments.

- 3. Payroll Expenses include direct/indirect wages, employee benefits, payroll taxes, payroll expenses. This does not include any expenses related to owner compensation/distribution.
- 4. Rent expense includes the cost of the rent payment only.
- 5. Fixed Expenses include expenses associated with advertising, office expenses, professional fees & uniforms.
- 6. Property Expenses include security, utilities, building maintenance, building repairs, telephone, internet, water & sewer expenses related to the property.
- 7. Variable Expenses include expenses associated with bank & credit card fees, insurance, personal property taxes, real estate taxes, licenses, permits, penalties, fines & workers compensation.
- 8. Royalty Fees collected from these locations are not uniform. The Royalty Fee figures reflected in the chart above have been adjusted to reflect a standard 7% royalty rate.
- 9. Brand Fund Fees collected from these locations are not uniform. The Brand Fund Fee figures reflected in the chart above have been adjusted to reflect a standard 2% brand fund rate.

XI. 2019 Actual EBITDA by Resort

Outlet	1	2	3	4	5	6	7	8	9
Total Gross Sales ¹	\$2,143,953	\$2,347,467	\$862,692	\$644,933	\$947,464	\$953,643	\$1,751,194	\$1,089,278	\$1,139,083
Cost of Goods Sold ²	\$58,935	\$86,194	\$24,712	\$58,232	\$29,430	\$36,421	\$50,801	\$123,690	\$63,459
Payroll Expenses ³	\$892,343	\$755,711	\$442,299	\$339,074	\$318,281	\$248,869	\$594,498	\$417,790	\$568,500
Rent Expense ⁴	\$300,000	\$166,435	\$61,632	\$64,125	\$144,000	\$91,383	\$228,000	\$312,000	\$21,000
Fixed Expenses ⁵	\$64,948	\$276,301	\$27,939	\$45,254	\$45,217	\$80,343	\$52,802	\$69,547	\$92,343
Property Expense ⁶	\$95,271	\$173,472	\$28,951	\$53,591	\$55,596	\$49,683	\$62,770	\$57,923	\$88,796
Variable Expenses ⁷	\$143,329	\$202,915	\$45,550	\$31,525	\$53,555	\$60,757	\$79,045	\$56,756	\$64,684
Royalty Fee ⁸	\$150,077	\$164,323	\$60,388	\$45,145	\$66,322	\$66,755	\$122,584	\$76,249	\$79,736
Brand Fund Fee ⁹	\$42,879	\$46,949	\$17,254	\$12,899	\$18,949	\$19,073	\$35,024	\$21,786	\$22,782
Total Expenses (\$)	\$1,747,782	\$1,872,300	\$708,725	\$649,845	\$731,351	\$653,284	\$1,225,523	\$1,135,741	\$1,001,299
Total Expenses (%)	81.5%	79.8%	82.2%	100.8%	77.2%	68.5%	70.0%	104.3%	87.9%

EBITDA (\$)	\$396,171	\$475,167	\$153,967	\$(4,912)	\$216,113	\$300,359	\$525,671	\$(46,463)	\$137,784
EBITDA (%)	18.5%	20.2%	17.8%	-0.8%	22.8%	31.5%	30.0%	-4.3%	12.1%

Notes:

- 1. Total Gross Sales represent revenue that royalties are paid on and as defined in Item 6 "Other Fees" Note (1).
- 2. Cost of Goods Sold includes grooming supplies, kennel supplies & expenses, general supplies, and price adjustments.
- 3. Payroll Expenses include direct/indirect wages, employee benefits, payroll taxes, payroll expenses. This does not include any expenses related to owner compensation/distribution.
- 4. Rent expense includes the cost of the rent payment only.
- 5. Fixed Expenses include expenses associated with advertising, office expenses, professional fees & uniforms.
- 6. Property Expenses include security, utilities, building maintenance, building repairs, telephone, internet, water & sewer expenses related to the property.
- 7. Variable Expenses include expenses associated with bank & credit card fees, insurance, personal property taxes, real estate taxes, licenses, permits, penalties, fines & workers compensation.
- 8. Royalty Fees collected from these locations are not uniform. The Royalty Fee figures reflected in the chart above have been adjusted to reflect a standard 7% royalty rate.
- 9. Brand Fund Fees collected from these locations are not uniform. The Brand Fund Fee figures reflected in the chart above have been adjusted to reflect a standard 2% brand fund rate.

Additional Notes:

- A. Some outlets have earned this amount. Your individual results may differ. There is no assurance you'll earn as much.
- B. You are responsible for developing your own business plan for your business, including capital budgets, financial statements, projections and other elements appropriate to your particular circumstances. As part of your planning, you need to take into account the expenses you will incur including labor and other operations expenses. Additional expenses that you may incur include, but are not limited to, royalty and brand fund fees detailed in Item 6 of this disclosure document, interest on debt service, insurance, legal and accounting charges, and depreciation/ amortization. Franchisees listed in this disclosure document, may be one source of information.
- C. In developing the business plan for your All American Pet Resorts® Business, you should make a necessary allowance for changes in financial results to income, expenses, or both, that may result from operation of your business during periods of, or in geographic areas

- suffering from, economic downturns, inflation, unemployment, or other negative economic influences.
- D. Although this financial performance representation includes some expense information, these numbers are calculated by Franchisor's accountants based on financial information provided by Franchisees. This financial performance representation does not include all costs of sales, operating expenses, or other costs or expenses that must be deducted from gross revenue or gross sales figures to obtain your net income or profit. You should conduct an independent investigation of the costs and expenses you will incur in operating your All American Pet Resorts[®] Business. Franchisees listed in this disclosure document, may be one source of information.
- E. Other than the preceding financial performance representation, we do not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Stephan Dimitroff, All American Pet Resorts, LLC Our principal business address is 41850 West Eleven Mile Road, Suite 202, Novi, MI, 48375, the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20 OUTLETS AND FRANCHISEE INFORMATION

TABLE NO. 1 SYSTEM-WIDE OUTLET SUMMARY FOR FISCAL YEARS 2021 TO 2023

Outlet Type	Year	Outlets at the Start of the Year	Outlets at the End of the Year	Net Change
	2021	9	9	0
Franchised	2022	9	9	0
	2023	9	10	+1
	2021	0	0	0
Company Owned	2022	0	0	0
	2023	0	0	0
	2021	9	9	0
Total	2022	9	9	0
	2023	9	10	+1

All American Pet Resorts[®] Royal Oak has a license to use the "All American Pet Resorts[®]" name. We included the All American Pet Resorts[®] Royal Oak in our franchised outlets count.

TABLE NO. 2 TRANSFER OF OUTLETS FROM FRANCHISEES TO NEW OWNERS (OTHER THAN THE FRANCHISOR) FOR FISCAL YEARS 2021 TO 2023

State	Year	Number of Transfers
	2021	0
All States	2022	0
	2023	0
	2021	0
Total	2022	0
	2023	0

TABLE NO. 3 STATUS OF FRANCHISED OUTLETS FOR FISCAL YEARS 2021 TO 2023

		Outlets at				Reacquired	Ceased Operations-	Outlets at
		Start of	Outlets		Non-	by	Other	End of the
State	Year	the Year	Opened	Terminations	Renewals	Franchisor	Reasons	Year
	2021	2	0	0	0	0	0	2
Florida	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	5	0	0	0	0	0	5
Michigan	2022	5	0	0	0	0	0	5
	2023	5	0	0	0	0	0	5
New	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
Jersey	2023	1	0	0	0	0	0	1
North	2021	0	0	0	0	0	0	0
Carolina	2022	0	0	0	0	0	0	0
Caronna	2023	0	1	0	0	0	0	1
	2021	1	0	0	0	0	0	1
Texas	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	9	0	0	0	0	0	9
Totals	2022	9	0	0	0	0	0	9
	2023	9	1	0	0	0	0	10

TABLE NO. 4 STATUS OF COMPANY OWNED OUTLETS FOR FISCAL YEARS 2021 TO 2023

State	Year	Outlets at Start of Year	Outlets Opened	Outlets Reacquired From Franchisee	Outlets Closed	Outlets Sold to Franchisee	Outlets at End of Year
	2021	0	0	0	0	0	0
All States	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
Totals	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0

TABLE NO. 5 PROJECTED OPENINGS AS OF DECEMBER 31, 2023

State	Franchise Agreements Signed but Outlet Not Opened	Projected New Franchise Outlets in the Next Fiscal Year	Projected New Company Owned Outlets in the Next Fiscal Year
Florida	0	1	0
Michigan	0	1	0
New Jersey	0	0	0
North Carolina	2	1	0
Texas	1	3	0
Virginia	0	1	0
Totals	3	7	0

The names, addresses and telephone numbers of all current franchisees, including those who have not opened locations yet, and Affiliate-owned outlets are listed in Exhibit C. Exhibit C also includes a list of the franchised outlets that were previously owned by a franchisee, reacquired by us or our Affiliate, and then resold to a new franchisee. **Exhibit C also includes a list of those franchisees who had an outlet terminated, canceled, transferred, not renewed or otherwise voluntarily or involuntarily ceased to do business under the franchise agreement during the period ended December 31, 2023**. No franchisees have failed to communicate with us within 10 weeks of the date of this Disclosure Document. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

During our last three fiscal years, franchisees have signed confidentiality clauses. In some instances, current and former franchisees sign provisions restricting their ability to speak openly about their experience with the All American Pet Resorts[®] System. You may wish to speak with current and former franchisees but be aware that not all such franchisees will be able to communicate with you.

A list of any name, address, telephone number, email address, and web address (to the extent known) of each trademark-specific franchisee organization associated with the All American Pet Resorts[®] System is listed on Exhibit D to this Disclosure Document.

ITEM 21 FINANCIAL STATEMENTS

Attached to the Disclosure Document as Exhibit A are our audited financial statements as of our fiscal years ending December 31, 2021, December 31, 2022, and December 31, 2023 with the independent auditor's report attached.

ITEM 22 CONTRACTS

Attached are the following agreements proposed for use in connection with our offering of franchises:

Exhibit:

- B. Franchise Agreement
- F. State-Specific Addendum to Disclosure Document and Franchise Agreement
- G. Non-Disclosure and Non-Competition Agreement
- H. Statement of Franchisee
- J. Application and Confidentiality Agreement
- K. Form General Release of Claims

ITEM 23 RECEIPT

The last two pages of the disclosure document (following the exhibits and attachments) are receipt pages acknowledging your receipt of the disclosure document. One copy is for your records, and one copy must be signed and dated by you and returned to us.



EXHIBIT A ALL AMERICAN PET RESORTS, LLC FINANCIAL STATEMENTS

All American Pet Resorts, LLC (a Michigan Limited Liability Company) Financial Statements December 31, 2023, 2022, and 2021

All American Pet Resorts, LLC **Table of Contents**

December 31, 2023, 2022, and 2021

Financial Statements

Independent Auditor's Report	1-2
Balance Sheets.	
Statements of Income	4
Statements of Changes in Members' Equity	5
Statements of Cash Flows.	6
Notes to the Financial Statements.	7-11



INDEPENDENT AUDITOR'S REPORT

To the Members of All American Pet Resorts, LLC

Opinion

We have audited the accompanying financial statements of All American Pet Resorts, LLC (a Michigan limited liability company), which comprise the balance sheets as of December 31, 2023, 2022, and 2021, and the related statements of income, changes in members' equity, and cash flows for the years then ended, and the related notes to the financial statements.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of All American Pet Resorts, LLC as of December 31, 2023, 2022, and 2021, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of All American Pet Resorts, LLC and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about All American Pet Resorts, LLC's ability to continue as a going concern within one year after the date that the financial statements are available to be issued.

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with generally accepted auditing standards will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with generally accepted auditing standards, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to fraud or
 error, and design and perform audit procedures responsive to those risks. Such procedures include
 examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are
 appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of All
 American Pet Resorts, LLC's internal control. Accordingly, no such opinion is expressed.

33762 Schoolcraft Road, Livonia, Michigan 48150-1506
734-427-2030 Fax 734-427-3004 CND@CNDCPA.com www.CNDCPA.com

- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting
 estimates made by management, as well as evaluate the overall presentation of the financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about All American Pet Resorts, LLC's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control related matters that we identified during the audit.

Other Information Included in the Annual Franchise Disclosure Document

Management is responsible for the other information included in the annual franchise disclosure document. The other information comprises the general information about the franchising operations but does not include the financial statements and our auditor's report thereon. Our opinion on the financial statements does not cover the other information, and we do not express an opinion or any form of assurance thereon. In connection with our audit of the financial statements, our responsibility is to read the other information and consider whether a material inconsistency exists between the other information and the financial statements, or the other information otherwise appears to be materially misstated. If, based on the work performed, we conclude that an uncorrected material misstatement of the other information exists, we are required to describe it in our report.

Cole, Newton & Duran
Cole, Newton & Duran

Livonia, Michigan March 28, 2024



All American Pet Resorts, LLC Balance Sheets December 31, 2023, 2022, and 2021

ASS	ETS					
		2023	2022		2021	
Current Assets						
Cash and cash equivalents	\$	260,337	\$	393,499	\$	308,653
Accounts receivable		67,483		71,127		58,773
Total Current Assets		327,820		464,626		367,426
Office Equipment		6,174		6,174		6,174
Less accumulated depreciation		(6,174)		(6,033)		(5,752)
Net Office Equipment		-		141		422
Other Assets						
Operating lease right of use asset		8,869		40,504		-
Total Other Assets		8,869		40,504		-
Total Assets	\$	336,689	\$	505,271	\$	367,848
LIABILITIES & ME	MBER	S' EQUITY				
Current Liabilities						
Accounts payable	\$	24,441	\$	28,097	\$	12,514
Deferred franchise fee revenue		149,750		46,667		46,667
Other deferred revenue		4,500		7,000		-
Accrued payroll		1,240		689		-
Current operating lease liability		8,869		34,105		-

188,800

188,800

147,889

336,689

\$

116,558

6,399

122,957

382,314

505,271

\$

59,181

59,181

308,667

367,848

See Notes to the Financial Statements

Total Current Liabilities

Total Liabilities and Members' Equity

Operating lease liability, long term

Total Liabilities

Other Liabilities

Members' Equity

All American Pet Resorts, LLC Statements of Income For the Years Ended December 31, 2023, 2022, and 2021

Revenue	2023	2022	2021	
Royalty revenue	\$ 810,173	\$ 742,623	\$ 645,639	
Brand development revenue	182,178	165,093	120,866	
Franchise fee revenue	51,917		4,333	
Total Revenue	1,044,268	907,716	770,838	
Operating Expenses				
Accounting fees	25,420	22,922	20,500	
Advertising	13,986	24,917	17,500	
Amortization	-	-	2,063	
Brand development	229,885	121,497	74,553	
Communication expense	1,596	1,556	-	
Consulting	23,550	29,909	127,747	
Contributions	-	2,312	450	
Depreciation	141	281	4,781	
Dues and subscriptions	16,809	11,409	12,925	
Employee health benefits	19,850	18,900	18,625	
Insurance	3,551	3,219	-	
Licensing fees	-	420	(425)	
Meetings and facilities	8,560	4,518	9,408	
Office expenses	23,332	21,501	15,377	
Payroll taxes	30,104	16,745	13,691	
Postage and shipping	332	362	315	
Professional fees	128,234	275,516	376,270	
Quality control	-	-	641	
Rent	34,085	40,650	19,100	
Retirement expense	783	-	-	
Repairs and maintenance	-	362	623	
Salaries and wages	582,655	213,564	171,167	
Software and licensing	1,544	833	379	
Supplies	-	-	2,575	
Training	3,776	4,000	-	
Travel	13,000	12,676	11,287	
Total Operating Expenses	1,161,193	828,069	899,552	
Net Operating Income (Loss)	(116,925)	79,647	(128,714)	
Other Income				
Other income	20,500		20,833	
Total Other Income	20,500		20,833	
Net Income (Loss)	\$ (96,425)	\$ 79,647	\$ (107,881)	

See Notes to the Financial Statements

All American Pet Resorts, LLC Statements of Changes in Members' Equity For the Years Ended December 31, 2023, 2022, and 2021

	2023		2022		2021	
Balance at Beginning of Year	\$	382,314	\$	308,667	\$	421,048
Member distributions		(138,000)		(6,000)		(4,500)
Net Income (loss)		(96,425)		79,647		(107,881)
Balance at End of Year	\$	147,889	\$	382,314	\$	308,667

See Notes to the Financial Statements

All American Pet Resorts, LLC Statements of Cash Flows For the Years Ended December 31, 2023, 2022, and 2021

	2023		2022		2021	
Cash Flows from Operating Activities						
Net Income (Loss)	\$	(96,425)	\$	79,647	\$	(107,881)
Adjustments to reconcile net income to						
net cash provided by (used in) operating activities						
Amortization		-		-		2,063
Depreciation		141		281		4,781
PPP forgiveness		-		-		(20,833)
(Increase) decrease in assets:						
Accounts receivable		3,644		(12,354)		(18,842)
Prepaid expenses and deposits		-		-		3,843
Increase (decrease) in liabilities						
Accounts payable		(3,656)		15,583		(17,394)
Accrued payroll		551		689		46,667
Deferred revenue and other liabilities		100,583		7,000		-
Net cash provided by (used in) operating activities		4,838		90,846		(107,596)
Cash Flows from Investing Activities						
Purchase of fixed assets		-		-		(4,500)
Net cash provided by (used in) investing activities	_	-	_	-	_	(4,500)
Cash Flows from Financing Activities						
Member distributions		(138,000)		(6,000)		(4,500)
Net cash provided by (used in) financing activities		(138,000)		(6,000)		(4,500)
Net Increase (Decrease) in Cash and Cash Equivalents		(133,162)		84,846		(116,596)
Cash and Cash Equivalents at Beginning of Year		393,499		308,653		425,249
Cash and Cash Equivalents at End of Year	\$	260,337	\$	393,499	\$	308,653
SUPPLEMENTAL DISCLOSURES						
Cash paid for interest	\$		\$	-	\$	-
Cash paid for income taxes	\$	-	\$	-	\$	-

See Notes to the Financial Statements

All American Pet Resorts, LLC Notes to the Financial Statements December 31, 2023, 2022, and 2021

NOTE 1 - Nature of Activities

All American Pet Resorts, LLC (the Company) is a pet care franchisor that offers a business model for providing pet boarding, daycare, and grooming services in pet resorts operating in the continental United States of America. Ten franchises are operating as of December 31, 2023. Revenue is derived primarily from royalties and brand development fees received from franchisees.

NOTE 2 - Summary of Significant Accounting Policies

Basis of accounting

The financial statements have been prepared on the accrual basis of accounting in accordance with accounting principles generally accepted in the United States of America and, accordingly, reflect all significant receivables, payables, and other liabilities.

Use of estimates

The preparation of financial statements in conformity with generally accepted accounting principles requires the Company to make estimates and assumptions that affect certain reported amounts and disclosures. Accordingly, actual results could differ from those estimates.

Fair value of financial instruments

Financial Accounting Standards Board (FASB) guidance on fair value measurements defines fair value, establishes a framework for measuring fair value, and expands disclosure of fair value measurements. The guidance applies to all assets and liabilities that are measured and reported on a fair value basis. The carrying amounts of financial instruments, including cash, accounts receivable, accounts payable, and accrued liabilities approximate fair value due to the short-term maturity of these instruments.

Cash and cash equivalents

The Company considers all short-term debt securities purchased with an original maturity of three months or less to be cash equivalents.

Accounts receivable

Revenue is provided by franchisee assessments for royalties and brand development. Management has reviewed the accounts as of December 31, 2023, 2022, and 2021, and believes they are all collectible. No allowance for credit losses has been provided. Bad debt is charged against operations in the period in which it is deemed uncollectible.

Property and equipment

Office equipment is stated at cost. Expenditures for major renewals and betterments that extend the useful lives of office equipment are capitalized. Expenditures for maintenance and repairs that do not improve or extend the useful lives of the equipment are charged to expenses as incurred. Depreciation is calculated using the straight-line method of depreciation over the estimated useful life of five years. Depreciation expense for the years ending December 31, 2023, 2022, and 2021 was \$141, \$281, and \$4,781, respectively. When items are disposed of, the costs and accumulated depreciation are eliminated from the accounts and any gain or loss is included in the results of operations.

Startup costs

Startup costs consist of legal costs and other miscellaneous startup costs incurred by the Company when it began operations in 2005. The Company amortized the startup costs using the straight-line method over 15 years. The startup costs were fully amortized in 2021. Amortization expense for 2023, 2022, and 2021, was \$0, \$0, and \$2,063, respectively.

All American Pet Resorts, LLC Notes to the Financial Statements December 31, 2023, 2022, and 2021

NOTE 2 - Summary of Significant Accounting Policies (continued)

Revenue recognition

The Company's revenues consist of fees from locations operated by franchisees. Revenues from franchisees include royalties based on a percentage of sales and initial fees. Initial franchise and transfer fees are recognized as the Company satisfies the performance obligation over the franchise term. Franchise fee revenue for the years ending December 31, 2023, 2022, and 2021, was \$51,917, \$0, and \$3,333, respectively. Deferred franchise fee revenue as of December 31, 2023, 2022, and 2021 was \$149,750, \$46,667, and \$46,667, respectively.

Advertising costs

The Company expenses advertising costs as they are incurred. Advertising expense for 2023, 2022, and 2021, was \$13,986, \$24,917, and \$17,500, respectively.

Income tax status

Effective January 1, 2007, the Company elected under the Internal Revenue Code to be taxed as an S-Corporation. In lieu of corporate income taxes, the members are taxed on their proportionate share of the Company's taxable income. Therefore, no provision or liability for federal income taxes has been included in the financial statements. The Company is no longer subject to Internal Revenue Tax examinations for years prior to 2020. The Company has evaluated FASB ASC 740, Income Taxes, and has concluded it has no uncertain positions.

Leases

The Company leases a building. The determination of whether an arrangement is a lease is made at the lease's inception. Under ASC 842, a contract is (or contains) a lease if it conveys the right to control the use of an identified asset for a period of time in exchange for consideration. Control is defined under the standard as having both the right to obtain substantially all of the economic benefits from use of the asset and the right to direct the use of the asset. Management only reassesses its determination if the terms and conditions of the contract are changed. Leases with an initial term of 12 months or less are not recorded on the statement of financial position. Lease expense is recognized for these leases on a straight-line basis over the lease term.

Operating leases are included in operating lease right-of-use ("ROU") assets, current operating lease liability, and operating lease liability long term on the balance sheet.

ROU assets represent the right to use an underlying asset for the lease term, and lease liabilities represent our obligation to make lease payments. Operating lease ROU assets and liabilities are recognized at the lease commencement date based on the present value of lease payments over the lease term. The Company has elected to use the practical expedient related to the discount rate and uses the risk-free interest rate based on the three-month U.S. Treasury bill at lease commencement. Operating lease ROU assets also includes any lease payments made and excludes any lease incentives. Lease expense for lease payments is recognized on a straight-line basis over the lease term. The Company's lease terms may include options to extend or terminate the lease when it is reasonably certain that we will exercise the option.

Subsequent events

Management has evaluated events and transactions for potential recognition or disclosure through the date of the auditor's report, which is the date the financial statements were available to be issued.

All American Pet Resorts, LLC Notes to the Financial Statements December 31, 2023, 2022, and 2021

NOTE 3 - Concentrations and Credit Risk

Cash

The Company maintains its cash balances at a national financial institution where the balances are insured by the Federal Deposit Insurance Corporation (FDIC) up to \$250,000. Amounts on deposit may at times exceed the insured amount. Uninsured balances as of December 31, 2023, were \$14,387. The Company has experienced no losses with respect to uninsured amounts on deposit.

NOTE 4 - Related Party Transactions

Member franchise ownership

Three members own a combined 100% of one franchise.

Consulting fees

A member of the Company was paid for consulting services performed for the Company. Fees paid to this member for the years ended December 31, 2023, 2022, and 2021, were \$0, \$190,000, and \$285,000, respectively. The fee paid in 2021 was for multiple years of service. There were no payables for these services for the years ending December 31, 2023, 2022, and 2021.

NOTE 5 - Operating Leases

On June 25, 2021, the Company leased new office facilities, starting October 1, 2021, for a term of 30 months. The lease calls for annual rent increases.

The following summarizes the line items in the balance sheets which include amounts for operating and finance leases as of December 31:

	2023		2022		2021	
Operating lease right of use asset	\$	8,869	\$	40,504	\$	-
Other current liabilities	\$	8,869	\$	34,105	\$	-
Operating lease liabilities		-		6,399		-
Total operating lease liabilities	\$	8,869	\$	40,504	\$	-

The following summarizes the weighted average remaining lease term and discount rate as of December 31:

	2023	2022	2021	_
Weighted Average Remaining Lease Term	0.25 years	1.25 years	N/A	
Weighted Average Discount Rate	4.764%	4.764%	N/A	

Future maturities of lease liabilities as of December 31, 2023 were as follows:

Present value of lease liabilities	\$ 8,869
Less: Interest	 (401)
Total lease liabilities payments	9,270
Year ending December 31, 2024	\$ 9,270

All American Pet Resorts, LLC Notes to the Financial Statements December 31, 2023, 2022, and 2021

NOTE 5 - Operating Leases (continued)

The following summarizes the components of operating lease expense for the year ended December 31:

	 2023	2022	2021
Operating lease cost	\$ 34,085	\$ 40,650	\$ 19,100

The following summarizes cash flow information related to leases for the year ended December 31:

	 2023	 2022	 2021
Operating cash flows from operating leases	\$ 34,085	\$ 40,650	\$ 19,100

Subsequent to December 31, 2023, the Company entered into an additional operating lease that has not yet commenced of approximately \$76,000. This operating lease will commence in March 2024 with a lease term of 2 years.

NOTE 6 - PPP Loan

On May 23, 2020, the Company was granted a loan (the Loan) from Wells Fargo Bank in the aggregate amount of \$20,833 pursuant to the Paycheck Protection Program (the PPP) under Division A, Title I of the CARES Act, which was enacted March 27, 2020.

The Loan, which was in the form of a Note dated May 23, 2020, was set to mature on May 23, 2022, and bore interest at a rate of 1% per annum. Funds from the Loan could only be used for payroll costs, costs used to continue group health care benefits, rent, utilities, and interest on debt obligations incurred before February 15, 2020. The Company used the entire Loan amount for qualifying expenses. Under the terms of the PPP, certain amounts of the Loan may be forgiven if they are used for qualifying expenses as described in the CARES Act. The unforgiven portion of the PPP loan is payable over two years at an interest rate of 1%, with payments deferred for the first ten months. The note was forgiven on April 13, 2021 and recorded as other income for the year ended December 31, 2021.

NOTE 7 - Revenue Recognition

The Company sells individual franchises. The franchise agreements typically require the franchisee to pay an initial, non-refundable fee prior to opening the respective location(s), and continuing royalty and brand development fees on a monthly basis based upon a percentage of franchisee sales. The initial term of a franchise agreement is typically 10 years. Subject to the Company's approval, a franchisee may generally renew the franchise agreement upon its expiration. If a contract is terminated prior to its term, it is a breach of contract, and a penalty is assessed based on a formula reviewed and approved by management. Revenue generated from a contract breach is termed settlement income by the Company and included in licensing fees and other income.

Under the terms of their franchise agreements, the Company typically promises to provide franchise rights, pre-opening services such as blueprints, operational materials, planning and functional training courses, and ongoing services, such as management of the marketing fund. The Company has determined that certain pre-opening activities, and the franchise rights and related ongoing services, represented two separate performance obligations. The franchise fee revenue has been allocated to the two separate performance obligations using a residual approach. The Company has estimated the value of performance obligations related to certain pre-opening activities deemed to be distinct based on cost plus an applicable margin and assigned the remaining amount of the initial franchise fee to the franchise rights and ongoing services.

All American Pet Resorts, LLC Notes to the Financial Statements December 31, 2023, 2022, and 2021

NOTE 7 - Revenue Recognition (continued)

The Company has elected to use the practical expedient regarding the pre-opening activities, which allows franchisors that are not public businesses to account for pre-opening services as a single performance obligation and is earned at a point in time. Revenue allocated to franchise rights and ongoing services is deferred until the store opens and recognized on a straight-line basis over the duration of the agreement, as this ensures that revenue recognition aligns with the customer's access to the franchise right. This revenue is earned over time.

Royalty revenue and brand development revenue is recognized during the respective franchise agreement based on the revenue earned each period as the underlying franchise store sales occur. Payment typically occurs within 30 days.

There are two items involving revenue recognition of contracts that require the Company to make subjective judgments: the determination of which performance obligations are distinct within the context of the overall contract and the estimated standalone selling price of each obligation. In instances where the contract includes significant customization or modification services, the customization and modification services are generally combined and recorded as one distinct performance obligation.

The following table disaggregates the Company's revenue based on the timing of satisfaction of performance obligations for the year ended December 31:

	 2023	 2022	 2021
Performance obligations satisfied at a point in time Performance obligations satisfied over time	\$ 41,667 1,002,601	\$ 907,716	\$ 3,333 767,505
Total revenue from contracts with customers	\$ 1,044,268	\$ 907,716	\$ 770,838

The following summarizes contract assets and contract liabilities as of December 31:

 2023		2022		2021
\$ 67,483	\$	71,127	\$	58,773
\$ 149,750	\$	46,667	\$	46,667
\$	\$ 67,483	\$ 67,483 \$	\$ 67,483 \$ 71,127	\$ 67,483 \$ 71,127 \$

There were no changes in judgments related to revenue recognition.

The Company uses the practical expedient to record revenue as if there is no significant financing component when the receivable is due within one year.

NOTE 8 - Retirement Plan

The Company has a defined contribution plan covering substantially all employees. In December of 2023, the plan was amended to include an employer matching contribution. Prior to December of 2023, employer contributions were discretionary. Employer contributions for the years ended December 31, 2023, 2022, and 2021 was \$783, \$0, and \$0, respectively.

NOTE 9 - Members' Equity

The Company is organized as a Michigan domestic limited liability company. The members' liability is limited to the financial contributions made by the members.



EXHIBIT B ALL AMERICAN PET RESORTS, LLC FRANCHISE AGREEMENT



ALL AMERICAN PET RESORTS, LLC EXHIBIT B

FRANCHISE AGREEMENT

Franchise #:		
Franchisee: _		
Date:		
Territory:		

TABLE OF CONTENTS

Section	<u>on</u>	<u>Page</u>
1.	DEFINITIONS	2
2.	COVENANTS, REPRESENTATIONS, AND WARRANTIES OF FRANCHISEE.	5
3.	GRANT OF FRANCHISE.	7
4.	TERM OF THE AGREEMENT AND LICENSE.	7
5.	TERRITORY.	9
6.	FEES.	11
7.	ACCOUNTING, RECORDS, AUDITS AND LATE PAYMENT CHARGES	12
8.	SERVICES AND ASSISTANCE	14
9.	FRANCHISEE'S DUTIES, OBLIGATIONS AND OPERATING STANDARDS	20
10.	PURCHASE OF EQUIPMENT, INVENTORY AND SUPPLIES.	26
11.	MARKS, COPYRIGHTED WORKS AND OWNERSHIP OF IMPROVEMENTS	27
12.	ADVERTISING AND PROMOTION	31
13.	INSURANCE AND INDEMNITY.	34
14.	RELATIONSHIP	36
15.	RESTRICTIVE COVENANTS	37
16.	ASSIGNMENT.	40
17.	OPTION TO PURCHASE — RIGHT OF FIRST REFUSAL.	45
18.	DEFAULT AND TERMINATION.	47
19.	NOTICES	53
20.	DISPUTE RESOLUTION.	54
21.	MISCELLANEOUS.	55
22.	ACKNOWLEDGEMENT	58

ATTACHMENTS:

- A. Initial Franchise Fee
- B. Territory
- C. Guaranty and Assumption of Franchisee's Obligations
- D. Statement of Ownership
- D-1 Affiliated Lessor Statement of Ownership
- E. EFT Authorization
- F. Collateral Assignment of Telephone Numbers, Telephone Listings and Internet Addresses
- G. Lease Addendum
- G-1 Collateral Assignment of Lease
- H. SBA Addendum

FRANCHISE AGREEMENT

THIS FRANCHISE AGREEMENT	("Agreement") is made this day of
	Date"), by and between ALL AMERICAN PET
RESORTS, LLC, a Michigan limited liability c	company, located at 41850 West Eleven Mile Road,
Suite 202, Novi, MI 48375 ("Franchisor") and	, located
at	("Franchisee").

RECITALS

WHEREAS, Franchisor has developed a comprehensive system for the operation of a business offering services for boarding, daycare and grooming pets, and the sale of related Products ("All American Pet Resorts[®] Business").

WHEREAS, the All American Pet Resorts[®] Businesses are operated under a business format using a unique system with high standards of service, including valuable know-how, information, Trade Secrets, Confidential Information, training methods, standards, designs, methods of trademark usage, copyrights, sources and specifications, confidential electronic and other communications, methods of Internet usage, and research and development ("System").

WHEREAS, the distinguishing characteristics of the System include the trademark "All American Pet Resorts®" and other trademarks and trade names, confidential operating procedures, confidential Operations Manual, standards and specifications for equipment, services and products, methods of service, management and marketing programs and sales techniques and strategies. All of these distinguishing characteristics may be changed, improved, and further developed by Franchisor from time to time. They are Franchisor's Confidential Information and Trade Secrets and are designated by and identified with the Marks described in this Agreement.

WHEREAS, Franchisor continues to use, develop and control the use of the Marks in order to identify for the public the source of services and products marketed under the System, and which represent the System's high standards of quality, service and customer satisfaction.

WHEREAS, Franchisee acknowledges the benefits to be derived from being identified with the System, and also recognizes the value of the Marks and the continued uniformity of image to Franchisee, Franchisor, and other franchisees of Franchisor.

WHEREAS, Franchisee acknowledges the importance to the System of Franchisor's high and uniform standards of quality, service and customer satisfaction, and further recognizes the necessity of opening and operating an All American Pet Resorts[®] Business in conformity with the System.

WHEREAS, Franchisee recognizes that in order to enhance the value of the System and goodwill associated with it, this Agreement places detailed obligations on Franchisee, including strict adherence to Franchisor's reasonable present and future requirements regarding the types of products sold, services offered, advertising used, operational techniques, marketing and sales strategies and related matters.

WHEREAS, Franchisee is aware of the foregoing and is desirous of obtaining the right to use the System and in association therewith, the right to use the Marks, and wishes to be assisted, trained, and franchised to operate an All American Pet Resorts[®] Business pursuant to the provisions and within the Territory specified in this Agreement, subject to the terms and conditions contained in this Agreement.

WHEREAS, Franchisee entered into an Application and Confidentiality Agreement (the "Application Agreement") (attached as Exhibit J to the Franchise Disclosure Document) indicating Franchisee's bona fide intent to enter into this Agreement.

The parties therefore agree as follows:

1. **DEFINITIONS.**

For the purposes of this Agreement, the following are hereby defined:

- 1.1 "Affiliate" means any person or entity that controls, is controlled by, or is in common control with, Franchisor.
- 1.2 "Affiliated Lessor" means any entity that controls, is controlled by, or is under common control with, Franchisee or any Franchisee's Owner(s) or in which Franchisee Owner(s) have an equity interest that leases premises to Franchisee or any person or entity for the operation of the All American Pet Resorts® Business.
- 1.3 "Agreement" means this agreement, attachments, exhibits, and all instruments in amendment hereof.
- 1.4 "All American Pet Resorts® Business" means the business operations conducted or to be conducted by Franchisee consisting of a business offering services for boarding, daycare and grooming pets, and the sale of related Products using Franchisor's System and in association with the Marks.
 - 1.5 "Application Agreement" has the meaning in the Recitals.
- 1.6 "Brand Development Fee" and "Brand Development Fund" shall be defined as set forth in Section 12.4.
- 1.7 "Operations Manual" means, but is not limited to, collectively, all directives, books, pamphlets, bulletins, brochures, memoranda, order forms, packing slips, invoices, letters, e-mail, Internet or intranet data, or other publications, documents, software programs, video tapes, DVDs, transmittances or communications, in whatever form (including electronic form) prepared by or on behalf of Franchisor for use by Franchisees generally or for Franchisee in particular, setting forth information, advice and standards, requirements, marketing information and procedures, operating procedures, instructions or policies relating to the operation of the All American Pet Resorts[®] Business or the operation of franchises, as same may be added to, deleted or otherwise amended by Franchisor from time to time. The form and content of the Operations

Manual maintained by Franchisor shall prevail in the event of any dispute regarding the form of or content of the Operations Manual between Franchisor and Franchisee.

- 1.8 "Business Interruption Insurance" means insurance that covers business losses related to a covered event which result in business closure, including lost revenue, mortgage, rent and lease payments, loan payments, taxes, payroll, relocation costs and training, as well as Royalty Fee, Brand Development Fee and any other related payments which must be paid to Franchisor.
- 1.9 "Cash Flow" means all cash funds derived from operations of the Franchised Business, without reduction for any non-cash charges, but less cash funds used to pay current operating expenses of the Franchised Business.
- 1.10 "Confidential Information" means all knowledge, know-how, standards, methods and procedures related to the establishment and operation of the System and includes all records pertaining to customers, suppliers, and other service providers of, and/or related in any way to, Franchisee's All American Pet Resorts® Business including, without limitation, all databases (whether in print, electronic or other form), all names, addresses, phone numbers, e-mail addresses, customer purchase records, manuals, promotional and marketing materials, marketing strategies and any other data which Franchisor designates as confidential.
- 1.11 "Designated Business Manager" means an individual who will have day-to-day management of the Business after successful completion of our initial training, exercise on-site supervision, personally participate in the direct operation of the Business, attend required conferences and meetings and has authority to make business decisions in your absence, unavailability, incapacity or death.
- 1.12 "Franchisor's System" or "System" means the standards, systems, concepts, identifications, collateral, methods, and procedures developed or used by Franchisor, or which may hereafter be developed or used by Franchisor, for the sales and marketing of Franchisor's Services and Products.
- 1.13 "Franchise" shall mean the business operations conducted or to be conducted using Franchisor's System and in association therewith the Marks.
- 1.14 "Franchisee Owner" or "Owner" means any person or entity with an equity interest, or any other type of ownership interest in Franchisee.
- 1.15 "Gross Revenues" means the total of all receipts derived from all sales of products and services at Franchisee's All American Pet Resorts® Business, labor, insurance claims for lost profits to the extent a claim is paid by the insurer, and all other products and services sold or performed by Franchisee or Franchisee's All American Pet Resorts® Business or by means of the business conducted under this Agreement, whether the receipts are evidenced by cash, credit, checks, gift certificates, scrip, coupons, services, property or other means of exchange. Gross Revenues do not include:
 - (a) the amount of any tax imposed by any federal, state, municipal or other governmental authority directly on sales and collected from customers, provided that the

amount of any such tax is shown separately and in fact paid by Franchisee to the appropriate governmental authority; and

(b) all customer refunds, valid discounts and coupons, and credits made by the All American Pet Resorts[®] Business (exclusions will not include any reductions for credit card user fees, returned checks or reserves for bad credit or doubtful accounts).

Gross Revenues shall be deemed received by Franchisee at the time the Services or Products from which they were derived are delivered or rendered or at the time the relevant sale takes place, whichever occurs first, regardless of whether final payment (e.g., collection on a customer's personal check) actually has been received by Franchisee. Gross Revenues consisting of property or Services shall be valued at the retail prices applicable and in effect at the time that they are received.

- 1.16 "Individual Advertising Expense" shall be defined as set forth in Section 12.1.
- 1.17 "Interim Period" shall be defined as set forth in Section 4.5.
- 1.18 "Lease" means any agreement (whether oral or written) under which the right to occupy a Pet Resort has been obtained, and any amendment made thereto from time to time, including without limitation, any offer to lease, license or lease agreement. Franchisee acknowledges and agrees that before any Lease will be accepted by Franchisor, the Lease must incorporate the terms of the Lease Addendum attached to this Agreement as Attachment F.
- 1.19 "Marks" shall mean the trademark "ALL AMERICAN PET RESORTS®" and "All American Pet Resorts...Because You Care®" to the extent of Franchisor's rights to same, together with such other trade names, trademarks, symbols, logos, distinctive names, service marks, certification marks, logo designs, insignia or otherwise which may be designated by Franchisor from time to time as part of the System for use by Franchisees, and not thereafter withdrawn.
- 1.20 "Operating Term" means the time period commencing on the date Franchisee signs this Franchise Agreement and continuing for a period of 10 years.
- 1.21 "Patent(s)" means any patents issued by the United States Patent and Trademark Office to Franchisor or its Affiliates.
- 1.22 "Pet Resort" or "Approved Location" means the approved location from which Franchisee provides Services and sells Products in connection with the All American Pet Resorts® Business.
- 1.23 "Products" means all supplies, material and equipment sold, prepared or otherwise dealt with in connection with the All American Pet Resorts® Business and associated with the Marks.
 - 1.24 "Purchase Offer" shall be defined as set forth in Section 16.6.
 - 1.25 "Royalty Fee" shall be defined as set forth in Section 6.3.

- 1.26 "Services" means the sale and provision of services for boarding, daycare and/or grooming for pets conducted or otherwise dealt with in connection with the All American Pet Resorts® Business and associated with the Marks.
- 1.27 "Territory" shall mean the geographical area conveyed to franchisee as further described in Attachment B and Section 3.1 of this Agreement.
- 1.28 "Trade Secret(s)" shall mean information, including a formula, pattern, compilation, program, device, method, training technique or process related to the System that both derives independent economic value, actual or potential, from not being generally known to, and not being readily ascertainable by proper means by other persons who can obtain economic value from its disclosure or use and is the subject of efforts that are reasonable under the circumstances to maintain its secrecy.

2. COVENANTS, REPRESENTATIONS, AND WARRANTIES OF FRANCHISEE.

Franchisee covenants, represents and warrants as follows and acknowledges that Franchisor is relying upon such covenants, representations and warranties in making its decision to enter into this Agreement.

- 2.1 Franchisee acknowledges that it has received, has had ample time to read, and has read this Agreement, and all related agreements with Franchisor. Franchisee acknowledges that Franchisor has advised it to obtain independent legal and accounting advice with respect to this Agreement and the transactions arising out of this Agreement. Franchisee further acknowledges that it has had an adequate opportunity to be advised by legal, accounting and other professional advisors of its own choosing regarding all pertinent aspects of the All American Pet Resorts[®] Business, Franchisor and this Agreement.
- 2.2 Franchisee has, or has made firm arrangements to acquire, funds to commence, open and operate the All American Pet Resorts[®] Business, and it is financially and otherwise able to accept the risks attendant upon entering into this Agreement.
- 2.3 All statements made by Franchisee in writing in connection with its application for this franchise were, to the best of its knowledge, true when made and continue to be true as of the date of this Agreement.
- 2.4 There are no material financial obligations of Franchisee, whether actual or contingent, which are outstanding as of the date of this Agreement other than those disclosed to Franchisor by Franchisee in writing.
- 2.5 Franchisee is not a party to or subject to any court or administrative order or action of any governmental authority which would limit or interfere in any way with the performance by Franchisee of its obligation hereunder.
- 2.6 Franchisee is not a party to any litigation or legal proceedings other than those which have been disclosed to Franchisor by Franchisee in writing.

- 2.7 Franchisee represents that it is not a party to or subject to agreements that might conflict with the terms of this Agreement and agrees not to enter into any conflicting agreements during the Operating Term or any Interim Period.
- 2.8 Franchisee agrees and acknowledges that it has not been induced to enter into this Agreement in reliance upon, nor as a result of, any statements, representations, warranties, conditions, covenants, promises or inducements, whatsoever, whether oral or written, and whether directly related to the contents hereof or collateral thereto, made by Franchisor, its officers, directors, agents, employees or contractors except as provided herein and in the Franchise Disclosure Document supplied to Franchisee. Franchisee acknowledges that the Franchise has been granted in reliance upon the information supplied to Franchisor in Franchisee's application for a Franchise.
- 2.9 Franchisee and its owners agree to comply with and/or to assist Franchisor to the fullest extent possible in Franchisor's efforts to comply with Anti-Terrorism Laws (as defined below). In connection with such compliance, Franchisee and its owners certify, represent, and warrant that none of their property or interests is subject to being "blocked" under any of the Anti-Terrorism Laws and that Franchisee and its owners are not otherwise in violation of any of the Anti-Terrorism Laws.
 - (a) Franchisee and its owners certify that they, their employees, and everyone associated with Franchisee are not listed in the Annex to Executive Order 13224 (which can be accessed at http://www.treasury.gov/offices/enforcement/ofac/sanctions/terrorism.html). Franchisee agrees not to hire (or, if already employed, retain the employment of) any individual who is listed in the Annex.
 - (b) Franchisee certifies that it has no knowledge or information that, if generally known, would result in Franchisee, its owners, their employees, or anyone associated with Franchisee to be listed in the Annex to Executive Order 13224.
 - (c) Franchisee is solely responsible for ascertaining what actions it must take to comply with the Anti-Terrorism Laws, and Franchisee specifically acknowledges and agrees that its indemnification responsibilities set forth in this Agreement pertain to its obligations under this Section 2.9.
 - (d) Any misrepresentation under this Section or any violation of the Anti-Terrorism Laws by Franchisee, its owners, agents, its employees shall constitute grounds for immediate termination of this Agreement and any other agreement Franchisee has entered with Franchisor or any of Franchisor's affiliates.
 - (e) "Anti-Terrorism Laws" means Executive Order 13224 issued by the President of the United States, the Terrorism Sanctions Regulations (Title 31, Part 595 of the U.S. Code of Federal Regulations), the Foreign Terrorist Organizations Sanctions Regulations (Title 31, Part 597 of the U.S. Code of Federal Regulations), the Cuban Assets Control Regulations (Title 31, Part 515 of the U.S. Code of Federal Regulations), the USA PATRIOT Act, and all other present and future federal, state and local laws, ordinances, regulations, policies, lists and any other requirements of any governmental authority

(including, without limitation, the United States Department of Treasury Office of Foreign Assets Control and any government agency outside the U.S.) addressing or in any way relating to terrorist acts and/or acts of war.

3. **GRANT OF FRANCHISE.**

- 3.1 Subject to all the terms and conditions of this Agreement, Franchisor hereby grants to Franchisee, and Franchisee accepts, for the Operating Term of this Agreement and any Interim Period the right and license ("License") to:
 - (a) Operate an All American Pet Resorts[®] Business upon the terms and conditions of this Agreement in one (1) territory described in Attachment B ("Territory");
 - (b) Use the Marks and the System;
 - (c) Use any Patents; and
 - (d) Offer and market ONLY Franchisor's approved Services and Products, unless Franchisor approves in writing Franchisee's request to offer and market complementary and non-competing services or products.
- 3.2 The License does not include the right to sell Products to any vendor who would in turn sell to consumers.
- 3.3 Therefore, Franchisor expressly reserves the right to add to, subtract from, revise, modify or change from time to time the System or any part thereof, and Franchisee agrees to, within ninety (90) days or such earlier period required by Franchisor in the event of a safety issue or issue pertaining to Franchisor's Trademarks, Confidential Information, or other intellectual property rights, to accept and comply with any such reasonable addition, subtraction, revision, modification or change and to make such reasonable expenditures as may be necessary to comply pursuant to Section 9.
- 3.4 Franchisee recognizes that the rights that are granted to Franchisee are for the specific Territory, defined in Section 5.1 below and no other, and cannot be transferred to an alternate Territory, without the prior written approval of Franchisor, which approval may be granted or withheld as Franchisor deems appropriate.

4. TERM OF THE AGREEMENT AND LICENSE.

- 4.1 Unless terminated in accordance with the provisions of this Agreement, this Agreement and the License granted hereunder will be effective as of the Effective Date noted in the opening paragraph and shall continue until the expiration of the Operating Term.
- 4.2 Upon the expiration of the Operating Term and any Interim Period, Franchisor will extend Franchisee's rights to operate the All American Pet Resorts[®] Business for unlimited additional terms ("Successor Term") of ten (10) years, if Franchisee:

- (a) Sends written notice to Franchisor, as provided in Section 19, not less than one year prior to the expiration of the Operating Term stating Franchisee's desire to extend its rights to operate the All American Pet Resorts® Business;
- (b) Executes a new agreement ("Successor Franchise Agreement") and all other agreements in the form then being used by Franchisor to grant new franchises, the terms of which may be materially different from this Agreement;
- (c) Executes a general release of all claims Franchisee may have against Franchisor, its officers, directors, members, shareholders, agents, Affiliates, and employees, whether in their corporate and/or individual capacities. This release shall include all claims arising under any federal, state, or local law, rule, or ordinance arising out of or concerning this Agreement (to the fullest extent permitted by law) and shall be in a form satisfactory to Franchisor;
- (d) Upgrades the Pet Resort and computer system used in operations of the All American Pet Resorts® Business to Franchisor's current standards;
- (e) Complies with all other provisions contained in the Operations Manual, as modified periodically by Franchisor;
 - (f) Provides proof of current licenses, insurance and permits; and
 - (g) Pays the \$5,000.00 renewal fee.
- 4.3 Despite Franchisee's compliance with the conditions of Section 4.2, Franchisor may refuse to extend Franchisee's rights to operate the All American Pet Resorts[®] Business if Franchisee:
 - (a) Has failed to remedy any breach of this Agreement specified by Franchisor in a written notice to Franchisee; or
 - (b) Has committed and received notice of 2 or more breaches of this Agreement in the 24 months prior to the end of the Operating Term, even if such breaches were timely remedied; or
 - (c) Has not given Franchisor a written notice of intent to extend its rights to operate the All American Pet Resorts[®] Business no less than 1 year prior to expiration of the Operating Term;
 - (d) Is not current in payment obligations to Franchisor or to Franchisee's, landlord, suppliers, or trade creditors; or
 - (e) Does not satisfy all then-current requirements for qualification of new franchisees.
- 4.4 Franchisor reserves the right to change any term(s) of the Franchise Agreement form to be signed by Franchisee upon the extension of Franchisee's rights to operate the All

American Pet Resorts[®] Business (except as specified below). There shall not, however, be another Initial Franchise Fee charged in connection with the extension of Franchisee's rights to operate the All American Pet Resorts[®] Business of the Franchise Agreement. IN FRANCHISOR'S SOLE DETERMINATION, FRANCHISEE MAY BE DEEMED TO HAVE IRREVOCABLY DECLINED TO EXTEND FRANCHISEE'S RIGHTS TO OPERATE THE FRANCHISE (AND ITS OPTION SHALL THEREUPON TERMINATE) IF IT FAILS TO EXECUTE AND RETURN TO FRANCHISOR THE SUCCESSOR FRANCHISE AGREEMENT AND OTHER DOCUMENTS REQUIRED BY FRANCHISOR WITHIN THIRTY DAYS AFTER THEIR DELIVERY TO FRANCHISEE, OR FAILS TO COMPLY IN ANY OTHER WAY WITH THE PROVISIONS OF THIS SECTION 4.

4.5 If Franchisee does not sign a Successor Franchise Agreement prior to the expiration of this Agreement and continues to accept the benefits of this Agreement after the expiration of this Agreement, then at the option of Franchisor, this Agreement may be treated either as (i) expired as of the date of expiration with Franchisee then operating without a franchise to do so and in violation of Franchisor's rights; or (ii) continued on a month-to-month basis ("Interim Period") until one party provides the other with written notice of such party's intent to terminate the Interim Period, in which case the Interim Period will terminate thirty (30) days after receipt of the notice to terminate the Interim Period. In the latter case, all obligations of Franchisee shall remain in full force and effect during the Interim Period as if this Agreement had not expired, and all obligations and restrictions imposed on Franchisee upon expiration of this Agreement shall be deemed to take effect upon termination of the Interim Period.

5. **TERRITORY.**

- 5.1 During the Operating Term and any Interim Period, so long as Franchisee is in compliance with all of its obligations hereunder (except as otherwise provided in this Agreement), and subject to Franchisor's reservation of rights set forth in Section 5.2 neither Franchisor nor any Affiliate will establish or license another person or entity to establish an All American Pet Resorts® Business using the Marks licensed to Franchisee within the Territory encompassed by the boundaries set forth in Attachment B, which will be provided by Franchisor and will be attached to this Agreement within thirty (30) days following Franchisor's approval of the Pet Resort and will be incorporated herein by this reference. Except as otherwise specifically provided in this Agreement, this Agreement does not restrict Franchisor or its Affiliates and does not grant rights to Franchisee to pursue any of Franchisor's or its Affiliates other business concepts other than the All American Pet Resorts® Business.
- 5.2 Franchisee acknowledges that the Franchise granted hereunder is non-exclusive and that Franchisor and its Affiliates retain the exclusive right, among others:
 - (a) to use, and to license others to use, the Marks and System for the operation of All American Pet Resorts[®] Businesses at any location other than in the Territory, regardless of proximity to the Territory;
 - (b) to use, license and franchise the use of trademarks or service marks other than the Marks, whether in alternative channels of distribution or at any location including

the Territory, in association with operations that are the same as, similar to or different than the All American Pet Resorts[®] Business;

- (c) to use the Marks and the System in connection with the provision of other services and products or in alternative channels of distribution such as those described in Section 5.2(d), at any location including the Territory;
- (d) to offer the Services or Products, or grant others the right to offer the Services or Products, whether using the Marks or other trademarks or service marks, through alternative channels of distribution, including without limitation, wholesalers, retail outlets or other distribution outlets (other than All American Pet Resorts® Businesses), or by Internet commerce (e-commerce), mail order or otherwise, whether inside or outside the Territory;
- (e) to use and own any websites utilizing a domain name incorporating one or more of the words "American", "Pet" and/or "Resorts" or similar derivatives thereof. Franchisor retains the sole right to market on the Internet and to use the Marks on the Internet, including all use of websites, social media platforms, domain names, URL's, directory addresses, metatags, linking, advertising, and co-branding and other arrangements. Franchisee may not independently market on the Internet, or use any domain name, address, locator, link, metatag, or search technique, with words or symbols similar to the Marks or otherwise establish any presence on the Internet without Franchisor's prior written approval. Franchisor intends that any Franchisee website be accessed only through Franchisor's home page. Franchisee will provide Franchisor with content for Franchisor's Internet marketing and will sign Internet and intranet usage agreements, if any. Franchisor retains the right to approve any linking or other use of its website;
- (f) to acquire businesses that are the same as or similar to the All American Pet Resorts[®] Business and operate such businesses regardless of where such businesses are located; provided, however, that if any such business is located inside the Territory, it may not be operated using the "All American Pet Resorts[®]" trademarks or trade names without Franchisee's approval, which will not be unreasonably withheld;
- (g) to be acquired by any third party which operates businesses that are the same as or similar to the All American Pet Resorts[®] Business regardless of where such businesses are located, including inside the Territory. In the event that Franchisor is acquired by a third party in accordance with this Section 5.2(g), such third party will assume the obligations of Franchisor under this Agreement;
- (h) to implement multi-area marketing programs which may allow Franchisor or others to solicit or sell to customers anywhere. Franchisor also reserve the right to issue mandatory policies to coordinate such multi-area marketing programs
- 5.3 Franchisor will establish the boundaries of the Territory using zip codes, radial distance or geographical elements. Franchisee acknowledges and agrees that once the Territory

has been established, it will not be changed regardless of any increase or decrease of the population in the Territory.

6. FEES.

- 6.1 Upon the execution of this Agreement, Franchisee shall pay to Franchisor, as a non-recurring initial franchise fee ("Initial Franchise Fee"), the sum set forth on Attachment A, plus, if due and payable, all applicable federal, state or municipal taxes. The Initial Franchise Fee shall be due upon the signing of this Franchise Agreement and shall be paid by means of cashier's check, money order or wire transfer. The Initial Franchise Fee shall be deemed to have been fully earned by Franchisor when paid and is non-refundable once paid.
- 6.2 Within fifteen (15) days of execution of this Agreement, Franchisee shall pay Franchisor a Real Estate & Resort Coordination Fee of \$20,000. If Franchisee is approved to open subsequent locations, this fee shall be due per location and must be paid to Franchisor at the same time Franchisee provides written notice to Franchisor of its intent to open additional location(s).
- 6.3 Beginning on the first day of the second month of the Operating Term (payment due for the first month of the Operating Term on the second month of the Operating Term) and continuing thereafter for each month or partial month of the Operating Term of this Agreement and any Interim Period, Franchisee shall pay to Franchisor a royalty fee equal to 7% of all Gross Revenues for the previous month ("Royalty Fee").
- 6.4 The Royalty Fee shall be payable to Franchisor on or before the 10th day of each month for the preceding calendar month and shall be payable through the entire Operating Term of this Agreement and any Interim Period. Franchisee shall pay the Royalty Fee monthly via electronic funds transfer. Franchisee shall not subordinate to any other obligation its obligation to pay the Royalty Fee or any other fee or charge hereunder. Each Royalty Fee payment will be based on a report as set forth in Section 6.4(a).
 - (a) Each Royalty Fee payment shall be, without exception, based on a statement of the previous month's Gross Revenues as generated by the Operations Software. Gross revenue reports are due to Franchisor on the first day of every month via fax or electronic email. Each failure to include a fully completed gross revenue report of the previous month's Gross Revenues with the Royalty Fees payable to Franchisor when due shall constitute a material breach of this Agreement.
 - (b) Franchisor requires Franchisee to remit fees via Franchisor initiated electronic funds transfer ("EFT") or other similar means utilizing a Franchisor approved computer system or otherwise. The EFT Authorization is attached to the Franchise Agreement as Attachment E. Under this procedure, Franchisee shall authorize Franchisor to initiate debit entries and/or credit correction entries to a designated checking or savings account for payments of fees and other amounts payable to Franchisor and any interest charged due thereon. Franchisee shall make funds available to Franchisor for withdrawal by electronic transfer no later than the due date for payment therefore. If Franchisee has not timely reported the Gross Revenues to Franchisor for any reporting period, then Franchisor shall be authorized, at Franchisor's option, to debit Franchisee's account in an

amount equal to (a) the fees transferred from Franchisee's account for the last reporting period for which a report of the Gross Revenues was provided to Franchisor as required hereunder; (b) the Royalty Fee and Brand Development Fee and any penalty fees; or (c) the amount due based on information retrieved by Franchisor.

6.5 Non-Compliance Fee

If Franchisee violates or breaches any term of this Agreement, including its failure to pay or to have adequate amounts available for electronic transfer of amounts Franchisee or its affiliates owe us or our affiliates, Franchisee's failure to timely provide required reports and financial information, and/or Franchisee's failure to attend in its entirety all mandatory meetings such as Franchisor's franchise convention, Franchisor may require Franchisee to pay \$250 for the first offense; \$500 for second offense and \$500 plus a daily fee of \$100 until default cured. The daily charge will continue until Franchisor determines that Franchisor has cured all deficiencies and is compliant with all terms of the Franchise Agreement. This fee will be due upon receipt of invoice. These non-compliance charges are intended to compensate Franchisor for additional expenses and certain losses it will incur as a result of Franchisee's non-compliance and are not a penalty or an expression of the total amount of such damages. Franchisor may periodically change or eliminate this charge. Imposition of this fee does not waive any rights Franchisor otherwise has under the Franchise Agreement.

7. ACCOUNTING, RECORDS, AUDITS AND LATE PAYMENT CHARGES.

- 7.1 Franchisee shall keep such complete records of its All American Pet Resorts® Business as a prudent and careful businessperson would normally keep. Franchisee must use the accounting system and the pre-formatted template required by Franchisor, if any. Franchisee shall keep its financial books and records as Franchisor may from time to time direct in the Operations Manual or otherwise, including retention of all invoices, order forms, payroll records, cash register tapes, check records, bank deposit receipts, sales tax records, refunds, cash disbursements, journals and general ledgers. Franchisee shall advise Franchisor of the location of all original documents and shall not destroy any records without the written consent of Franchisor.
- 7.2 Franchisee shall prepare on a current basis, complete and accurate records concerning all financial, marketing and other operating aspects of the All American Pet Resorts® Business conducted under this Agreement. Franchisee shall maintain an accounting system which accurately reflects all operational aspects of the All American Pet Resorts® Business including uniform reports as may be required by Franchisor. Franchisee's records shall include tax returns, daily reports, statements of Gross Revenues (to be prepared each month for the preceding month), profit and loss statements (to be prepared semi-annually by an independent Certified Public Accountant), and balance sheets (to be prepared at least annually by an independent Certified Public Accountant).
- 7.3 Franchisee shall also submit to Franchisor current financial statements and other reports as Franchisor may reasonably request to evaluate or compile research and performance data on any operational aspect of the All American Pet Resorts® Business. Franchisee shall provide Franchisor with its unaudited financial documents on or before February 15 of each year. Franchisee shall submit Minimum Individual Marketing Expense statements to Franchisor once

each quarter, beginning April 1 of each year. On or before April 15 of each year, Franchisee shall provide Franchisor with a copy of its federal tax return for the previous tax year.

- 7.4 The records required under this Section 7 pertain only to Franchisee's operation of the All American Pet Resorts[®] Business. Franchisor has no right to inspect, audit or copy the records of any unrelated business activity Franchisee may have. Franchisee shall keep the books and records of the All American Pet Resorts[®] Business separate from the records of any unrelated business activity or personal activity. Franchisor is entitled to independently or remotely access your All American Pet Resorts[®] information and data electronically through our proprietary data management and internet system, pursuant to Item 11.
- From the date Franchisee and Franchisor sign this Agreement until three (3) years 7.5 after the end of the Operating Term of this Agreement, including any Interim Period, Franchisor or Franchisor's authorized agent shall have the right to request, receive, inspect and audit any of the records referred to above wherever they may be located, including without limitation, any and all financial records. Franchisor agrees to do inspections and audits at reasonable times. Franchisee agrees to keep all records and reports for six (6) years from the date such records are created. Should any inspection or audit disclose a deficiency in the payment of any Royalty Fee, Brand Development Fee or other amounts required to be paid under this Agreement, Franchisee shall immediately pay the deficiency to Franchisor, without prejudice to any other remedy of Franchisor under this Agreement. In addition, if the deficiency for any audit period discloses a deficiency in the amount of any Royalty Fee, Brand Development Fee or other amounts due by 2% or more, Franchisee will also immediately pay to Franchisor the entire cost of the inspection or audit including travel, lodging, meals, salaries and other expenses of the inspecting or auditing personnel. For the purposes of this Section 7.5, an audit period will be each fiscal year. Should the audit disclose an overpayment of any Royalty Fees, Brand Development Fees, or other amounts due, Franchisor shall credit the amount of the overpayment to Franchisee's payments of Royalty Fees, and Brand Development Fees next falling due.
- 7.6 If Franchisee's records and procedures are insufficient to permit a proper determination of Gross Revenues, Franchisor shall have the right to deliver to Franchisee an estimate, made by Franchisor, of Gross Revenues for the period under consideration and Franchisee shall immediately pay to Franchisor any amount shown thereby to be owing on account of the Royalty Fee, Brand Development Fee and other sums due on account of any understatement. Any such estimate shall be final and binding upon Franchisee.
- 7.7 To encourage prompt payment and to cover the costs and expenses involved in handling and processing late payments, Franchisee shall also pay, within five (5) days of the date of Franchisor's notice of late payment to Franchisee, a late interest charge equal to the lesser of (i) 1.5% per month; or (ii) the highest legal rate permitted by applicable law, whichever is lower, on all payments due to Franchisor during the period of time said payments are due and unpaid. Each failure to pay Royalty Fees, Brand Development Fees, and other amounts payable to Franchisor when due shall constitute a material breach of this Agreement. Franchisee acknowledges that this Section 7.7 shall not constitute Franchisor's agreement to accept such payments after same are due or a commitment by Franchisor to extend credit to, or otherwise finance Franchisee's operation of the All American Pet Resorts® Business. Further, Franchisee acknowledges that failure to pay all

such amounts when due shall, notwithstanding the provisions of this Section 7.7, constitute grounds for termination of this Agreement, as provided in this Agreement.

- 7.8 Any report of Franchisor's auditor rendered from time to time pursuant to this Section 7, shall be final and binding upon all of the parties hereto.
- 7.9 Franchisee hereby authorizes Franchisor to make reasonable inquiries of Franchisee's bank, suppliers, accountants and trade creditors concerning the All American Pet Resorts[®] Business and hereby directs such persons and companies to provide to Franchisor such information and copies of documents pertaining to the All American Pet Resorts[®] Business as Franchisor may request.
- 7.10 Franchisee acknowledges and agrees that Franchisor owns all business records ("Business Records") with respect to customers and other service professionals of, and/or related to, the All American Pet Resorts® Business including, without limitation, all databases (whether in print, electronic or other form), including all names, addresses, telephone numbers, e-mail addresses, customer purchase records, and all other records contained in the database, and all other Business Records created and maintained by Franchisee. Franchisee further acknowledges and agrees that, at all times during operations under this Franchise Agreement, and after the termination, expiration or cancellation of this Agreement, Franchisor may access such Business Records, and may utilize, transfer, or analyze such Business Records as Franchisor determines to be in the best interest of the System.
- 7.11 To encourage prompt delivery of all Business Records, Certificates of Insurance, Gross Revenue statements and any other documentation or record that may be requested by Franchisor under this Agreement, Franchisee shall pay, upon demand, a late report fee in the amount of \$100 per record or document requested if Franchisee fails to deliver such record or document when due after third occurrence.
- 7.12 If Franchisee fails to pay the Royalty Fee or any other sums due to Franchisor under this Agreement by the due date 2 times during the Operating Term or any Interim Period, in addition to all other remedies which may be available, Franchisor reserves the right to require Franchisee to pay the Royalty Fee or any other sums due to Franchisor under this Agreement via weekly debits.
- 7.13 Franchisee agrees that, during the Operating Term and for 3 years after the expiration and termination of this Agreement, Franchisee shall supply to Franchiser Franchisee's All American Pet Resorts® Business location address and telephone number.

8. SERVICES AND ASSISTANCE.

- 8.1 The Initial Franchise Fee and Royalty Fee are paid for the License, which includes the use of the Marks, the System and the use of Franchisor's Trade Secrets and Confidential Information provided pursuant to this Agreement and for certain services rendered by Franchisor.
- 8.2 Franchisor shall offer Franchisee initial and continuing services, as Franchisor deems necessary or advisable in furthering Franchisee's All American Pet Resorts® Business and

the business of the System as a whole and in connection with protecting the Marks and goodwill of Franchisor. Failure by Franchisor to provide any particular service, either initial or continuing, shall not excuse Franchisee from any of its obligations under this Agreement.

- 8.3 Currently, initial services provided by Franchisor prior to Franchisee opening the All American Pet Resorts® Business shall include:
 - (a) Designating Franchisee's Territory in Attachment B.
 - (b) Furnishing Franchisee with prototype plans and specifications franchisor may have in its possession for All American Pet Resorts, as more fully provided in Section 10 (Franchisor is not obligated to create customized plans for Franchisee).
 - (c) Approving in writing Franchisee's proposed Pet Resort in accordance with Section 8.4.
 - (d) After Franchisee's receipt of all required licenses and permits, providing Franchisee, or a person designated to manage the All American Pet Resorts[®] Business ("Designated Business Manager"), as applicable, and one (1) additional person without extra charge with an initial training program. The initial training program shall be twenty (20) days or less, based on Franchisee's level of experience in the industry and as determined by Franchisor in its sole discretion, and shall be held in part at a facility designated by Franchisor and Franchisee's Business. Training will include a discussion of the System, techniques, procedures, methods of training and operation, advertising, sales techniques, promotional ideas, marketing plans, customer relations, instructions on quality standards and practical experience in the operation of an All American Pet Resorts.[®] Initial training might entail working at another franchisee's location.
 - Loaning Franchisee during the Operating Term (including any Interim Period) 1 copy of Franchisor's confidential Operations Manual containing mandatory and suggested specifications, standards, operating procedures and rules prescribed from time to time by Franchisor as further stipulated in this Section 8, and containing information relative to other obligations of Franchisee hereunder. Specifications, standards and operating procedures prescribed from time to time by Franchisor in the Operations Manual or otherwise communicated to Franchisee in writing shall constitute provisions of this Agreement as if fully set forth herein. Franchisee shall operate the All American Pet Resorts[®] Business strictly in accordance with the Operations Manual. Failure to comply with the standards set forth in the Operations Manual shall constitute a material breach of this Agreement. Franchisor reserves the right to provide the Operations Manual and updates to the Operations Manual in electronic form or other form determined by Franchisor. Franchisor shall have the right to add to, and otherwise modify, the Operations Manual from time to time to reflect changes in authorized Products and Services, business image or the operation of the All American Pet Resorts[®] Business; provided, however, no such addition or modification shall alter Franchisee's fundamental status and rights under this Agreement. Some of the revisions to the Operations Manual may include changes with respect to: (i) sales and marketing strategies; (ii) equipment and supplies; (iii) accounting and reporting systems and forms; (iv) insurance requirements; (v) operating procedures;

- (vi) Services; and (vii) Products. In connection with the foregoing, Franchisee acknowledges and agrees:
 - (i) to accept, implement and adopt any such modifications at its own cost, except as provided in Section 9.5 of this Agreement; provided, however, that Franchisee will not be required to expend in any calendar year an amount that, when added to other amounts Franchisor is required Franchisee to spend hereunder during the current and two prior calendar years, exceeds 1% of Franchisee's Gross Revenues for the three calendar years preceding the required expenditure
 - (ii) to keep its Operations Manual with replacement pages and insertions as instructed by Franchisor; and
 - (iii) that the Operations Manual is loaned to Franchisee and shall at all times remain the sole and exclusive property of Franchisor, and upon termination of this Agreement for any reason whatsoever, Franchisee shall forthwith return the Operations Manual together with all copies of any portion of the Operations Manual which Franchisee may have made, to Franchisor.
- (f) Providing Franchisee with a set of letterhead, business cards, brochures and other start-up materials determined by Franchisor each at no charge. Additional copies of letterhead, business cards and other start-up materials after the initial inventory may be provided at Franchisee's expense.
- (g) Franchisor may provide other pre-opening and grand opening assistance during the first 90 days of operations of Franchisee's All American Pet Resorts[®] Business. Franchisor may charge Franchisee a reasonable fee for such requested services.
- Franchisee is solely responsible for locating a site from which to operate the Pet Resort and negotiating a Lease for the property. Upon request, and at Franchisor's sole discretion and Franchisee's cost, Franchisor may provide assistance to Franchisee in analyzing a location and in negotiating a Lease. Specifically, Franchisor will send a representative into Franchisee's Territory to assist Franchisee with site selection for the Pet Resort or to obtain the necessary zoning or building permit approval on no more than 2 separate occasions without charging Franchisee a fee. Franchisor will pay all costs and expenses associated with sending this representative into Franchisee's Territory on these first 2 occasions. If Franchisee requests that Franchisor send a representative into Franchisee's Territory on more than 2 occasions to assist Franchisee in the site selection process, Franchisor reserves the right to charge a nonrefundable site selection assistance fee ("Site Selection Assistance Fee") in accordance with the fee schedule set forth in the Operations Manual. In addition to the Site Selection Assistance Fee, Franchisee will also be responsible for paying all living expenses and travel expenses for Franchisor's representative to travel to Franchisee's Territory to provide Franchisee with this additional site selection assistance. Franchisor will analyze a location by examining population density, census data, and proximity of the proposed location to other All American Pet Resorts[®] Businesses, or any other criteria as set forth in Section 9.2(b). The Pet Resort site is subject to Franchisor's written approval, which approval Franchisor may grant or deny as it deems appropriate, in its reasonable business judgment. Franchisee agrees that the location of the Pet Resort is a factor in the potential success

of the All American Pet Resorts[®] Business and Franchisor may reject any location as it deems appropriate. However, Franchisee agrees that Franchisor's assistance in no way constitutes a representation or warranty with respect to the success or viability of the property or the Lease. Franchisee acknowledges that Franchisor's approval of the Pet Resort site indicates only that Franchisor believes that the site falls within acceptable criteria established by Franchisor as of the approval date. Franchisee shall not change the location of the Pet Resort without the prior written consent of Franchisor.

- 8.5 Currently, the services provided by Franchisor to Franchisee after Franchisee opens the All American Pet Resorts[®] Business shall include:
 - (a) Making a representative reasonably available to speak with Franchisee on the telephone during normal business hours, to discuss Franchisee's operational issues and support needs in Franchisor's discretion and as Franchisor deems appropriate.
 - (b) Holding periodic conferences to discuss sales techniques, new Product developments, bookkeeping, training, accounting, inventory control, performance standards, advertising programs, merchandising procedures and other topics in Franchisor's discretion and as Franchisor deems appropriate. Franchisee must pay all its travel and living expenses to attend.
 - (c) Franchisor may also hold mandatory biennial or annual conferences and conventions to discuss sales techniques, new Service and Product developments, training, bookkeeping, accounting, performance standards, advertising programs, merchandising procedures and other topics in Franchisor's discretion and as Franchisor deems appropriate. Unless Franchisee received express written consent from the Franchisor not to attend the convention, Franchisee must attend any such franchise conference and convention. Franchisee must pay the conference fee, if any, and all personal travel and living expenses. Informing Franchisee of mandatory specifications, standards and procedures for the operations of the All American Pet Resorts Business in Franchisor's discretion and as Franchisor deems appropriate.
 - (d) Researching new Products, Services and training methods, from time to time, and providing Franchisee with information concerning developments of this research in Franchisor's discretion and as Franchisor deems appropriate.
 - (e) Maintaining the Brand Development Fund and using these funds to develop promotional and advertising programs and public relations coverage for All American Pet Resorts[®] Businesses in Franchisor's discretion and as Franchisor deems appropriate.
 - (f) Providing advertising materials to Franchisee in the form of a graphic art package included in the Operations Manual and as further provided in Section 12.
 - (g) A representative of Franchisor may provide additional assistance in Franchisor's discretion and as Franchisor deems appropriate. There may be additional charges for this additional assistance. If Franchisor provides additional assistance,

Franchisor and Franchisee must agree in advance on the charges for the visit and the length of the visit.

- (h) Providing Franchisee with a quarterly newsletter, in Franchisor's discretion and as Franchisor deems appropriate.
- 8.6 If Franchisee believes Franchisor has failed to adequately provide pre-opening services to Franchisee as provided in this Agreement in Sections 8.3 and 8.4, Franchisee shall notify Franchisor in writing within 30 days following the opening of the All American Pet Resorts® Business. Absent the timely provision of such notice to Franchisor, Franchisee shall be deemed to conclusively acknowledge that all pre-opening and opening services required to be provided by Franchisor were sufficient and satisfactory in Franchisee's judgment.
- To assure protection of Franchisee's customers, the Marks and System for the benefit of Franchisee and all other All American Pet Resorts® Businesses, Franchisee hereby makes, constitutes and appoints Franchisor, or Franchisor's designee, as Franchisee's true and lawful attorney-in-fact, with the powers set forth below, which may, in Franchisor's determination, be exercised if any of the following events occur: (i) Franchisee is convicted of a criminal act which carries with it imprisonment for more than thirty (30) days or imprisoned for any reason more than thirty (30) days or which involves moral turpitude or is reasonably likely, in Franchisor's business judgment, to materially and unfavorably affect the System, Marks, goodwill, or reputation thereof; (ii) Franchisee is convicted by any federal, state, or local authority of an offense which voids or nullifies any licenses to operate an All American Pet Resorts® Business; (iii) Franchisee or the Designated Business Manager become mentally or physically incapacitated as determined by a court of competent jurisdiction; (iv) Franchisee is charged or convicted of, or acknowledges or admits to, a crime of moral turpitude, or any crime that is reasonably likely, in the Franchisor's business judgment, to materially and unfavorably affect the System, Marks, goodwill or reputation thereof; or (v) Franchisee commits any of the acts set forth in Sections 18.1(b), 18.1(c), 18.1(d), 18.1(e), 18.1(g), 18.1(j), 18.1(o), 18.1(p) or 18.1(q) of this Agreement. Upon the occurrence of any of the foregoing events, Franchisor may, but is not obligated to, take possession and control of and operate the All American Pet Resorts[®] Business for Franchisee's benefit. Such right shall continue only for such period of time as the reason or reasons for Franchisor's taking possession and control of the All American Pet Resorts[®] Business, as set forth above, continue; provided, however, that Franchisor reserves the right to terminate its management of the All American Pet Resorts[®] Business at any time. Franchisor shall act diligently, in good faith and honestly in Franchisor's actions hereunder. In furtherance of the foregoing, Franchisor may:
 - (a) collect any and all revenues due and payable to the All American Pet Resorts[®] Business and endorse Franchisee's name on checks received;
 - (b) pay any and all expenses incurred to operate the All American Pet Resorts[®] Business including, but not limited to, wages, salaries and other compensation to Franchisee's employees, to Franchisor and persons Franchisor employs on Franchisee's behalf to manage the All American Pet Resorts[®] Business and to others for professional services;

- (c) pay any amounts due to Franchisor or Franchisor's affiliates, including the continuing Royalty Fees, Brand Development fees, amounts due for purchases of product and supplies and amounts due under any financing agreements;
- (d) incur debts in the ordinary course of business for inventory, materials, supplies and other items needed for the operation of the All American Pet Resorts[®] Business:
- (e) execute documents or instruments on Franchisee's behalf as such documents or instruments relate to the All American Pet Resorts[®] Business or Franchisor's System, Trademarks, Copyrighted Material or other proprietary rights;
- (f) receive a reasonable fee for Franchisor's services hereunder (as more fully set forth below);
- (g) institute legal or administrative proceedings on behalf of and defend actions brought against the All American Pet Resorts® Business; and
- (h) take any other action Franchisor deems necessary or appropriate in furtherance of this provision.

Franchisor shall maintain separate books and records of Franchisor's actions hereunder in accordance with the format required by the System. The net proceeds, if any, from Franchisor's operation of the All American Pet Resorts® Business shall be deposited into a separate bank account or accounts under Franchisor's direction and control as trustee for Franchisee. Upon Franchisor's termination of the rights granted hereunder, such net proceeds, if any, shall be distributed to Franchisee or as Franchisee directs. Franchisor shall not be liable to Franchisee except for willful misconduct or gross negligence. Franchisee shall indemnify and hold Franchisor harmless from and against any loss, claim, expense, damage, liability or other obligation of any nature, including legal fees and expenses arising from or in any manner connected with Franchisor's actions hereunder, excepting only those arising from or connected with Franchisor's willful misconduct or gross negligence. Franchisee's appointment of Franchisor as attorney-in-fact is irrevocable and is coupled with an interest.

In addition to any other fees due under this Agreement and reimbursement of Franchisor's out-of-pocket expenses to provide management services, Franchisor is entitled to receive a fee for the management services provided by Franchisor equal to: (i) a management fee of 7% of the Gross Revenues of the All American Pet Resorts® Business ("Management Fee"); and, (ii) an incentive fee of 10% of the Cash Flow after payment of all expenses of operation of the All American Pet Resorts® Business, including Franchisor's out-of-pocket expenses and Management Fee ("Incentive Fee"). The Management Fee shall be paid within ten (10) days after the end of each month, or part thereof, and the Incentive Fee shall be paid within ten (10) days after the end of each calendar quarter, or part thereof, that management services are provided by Franchisor. Any portion of the Management Fee or the Incentive Fee which is not paid when due shall bear interest at the rate set forth in Section 7.7 until such amount is paid in full.

In addition to the foregoing, Franchisee acknowledges that if Franchisor becomes aware, by any means, of any situation where any license required for the All American Pet Resorts[®] Business is or may be suspended, terminated or affected in any manner that could result in the closing of the All American Pet Resorts[®] Business for any period time, Franchisor may, but is not obligated to, notify any federal, state or local authority. If Franchisor takes such action, Franchisee agrees that Franchisor shall not be liable to Franchisee for any costs, expenses or damages resulting therefrom.

8.8 Franchisor is not obligated to perform services set forth in this Agreement to Franchisee's particular level of satisfaction, but as a function of Franchisor's experience, knowledge and judgment. Franchisor does not represent or warrant that any other services will be provided to Franchisee, other than as set forth in this Agreement. To the extent any other services, or any specific level or quality of service is expected, Franchisee must obtain a commitment to provide such service or level of service in writing signed by an authorized officer of Franchisor, otherwise Franchisor shall not be obligated to provide any other services or specific level or quality of services.

9. FRANCHISEE'S DUTIES, OBLIGATIONS AND OPERATING STANDARDS.

- 9.1 Franchisee shall, consistent with the terms of this Agreement, diligently develop the All American Pet Resorts[®] Business and use its best efforts to market and promote the required Services and Products.
- 9.2 Franchisee shall complete the construction of Franchisee's Pet Resort, and shall maintain the Pet Resort, in accordance with the following requirements:
 - (a) Franchisee shall, at Franchisee's sole cost and expense, complete the interior build-out and install all furniture, fixtures and equipment as specified by Franchisor in the Operations Manual, and required by this Agreement.
 - Franchisee may purchase or lease the required real property and improvements from any source upon terms approved by Franchisor in writing, such approval not to be unreasonably withheld in Franchisor's business judgment. Proposals of location of the Pet Resort must be submitted to Franchisor within 60 days of the execution of this Agreement or Franchisor may elect to terminate this Agreement. Franchisee must deliver to Franchisor any traffic, competition and demographic or similar location information relating to any proposed site that Franchisor reasonably requests for review at least 20 days before any proposed Lease signing date. Franchisee must deliver to Franchisor a copy of the proposed Lease, in a form acceptable to Franchisor, and such Lease must incorporate the terms of the Lease Addendum attached to this Agreement as Attachment G. If an Affiliated Lessor exists, Franchisee must notify Franchisor of such Affiliated Lessor before submitting location and lease terms to Franchisor for approval. If Franchisor assists Franchisee in negotiating the purchase or lease of the required real property and improvements or negotiating Franchisor's required Lease Addendum terms into the Lease, or Franchisor requires Franchisee to use an independent third-party or attorney to assist in Franchisee's Lease negotiation, Franchisor may charge Franchisee a lease negotiation fee ("Lease Negotiation Fee") as set forth in the Operations Manual which

may be modified by Franchisor periodically. The Lease Negotiation Fee will be due prior to the negotiation of the Lease, and is non-refundable once paid, even if the Lease negotiation is not completed. In the event Franchisee or a Franchisee Owner maintains or obtains an equity interest in an Affiliated Lessor or any entity that owns the real property on which the All American Pet Resorts[®] Business is located, Franchisee is required to ensure that the Affiliated Lessor or real property owner executes a Non-Disclosure and Non-Competition Agreement (Exhibit G to the Franchise Disclosure Document), guaranty (Attachment C to this Agreement), and statement of ownership (Attachment D-1 to this Agreement) at that time. Notwithstanding anything herein to the contrary, Franchisor may extend the time periods set forth in this Section 9.2(b).

- (c) Subject to the terms of this Section, Franchisee must at all times comply with Franchisor's interior and exterior brand requirements, standards, specifications, processes, procedures, requirements and instructions regarding the Pet Resort's physical facilities, including the layout of furnishings and fixtures. During the design phase of Franchisee's All American Pet Resorts® Business Franchisee or its designated representative must spend at least one working day at another All American Pet Resort® (at Franchisee's expense) in order to understand the architectural specifications and the implications of those specifications in the development of Franchisee's All American Pet Resorts® Business. Franchisee must maintain the Pet Resort and any parking areas in good and safe condition, as specified in the Operations Manual. Franchisee must remodel or upgrade the Pet Resort at its sole cost and expense in accordance with Franchisor's standards as set forth in the Operations Manual, which may be modified by Franchisor at any time.
- 9.3 Subject to the terms of this Agreement, including Sections 8.3(e)(i) and 8.3(e)(iii), during the Operating Term and any Interim Period, Franchisee shall strictly comply with all present and future standards, specifications, processes, procedures, requirements, and instructions of Franchisor regarding the operation of the All American Pet Resorts[®] Business and must comply with the following requirements:
 - (a) Prior to opening the All American Pet Resorts[®] Business, Franchisee and Franchisee's Designated Business Manager must attend and successfully complete all initial training programs, a portion of which might entail working at an existing All American Pet Resorts.[®] Franchisee shall be responsible for travel, meals, personal expenses and living expenses incurred by itself, the Designated Business Manager, and additional persons that participate in the initial training program.
 - (b) Unless Franchisee receives express written consent from the Franchisor not to attend the convention, Franchisee or its Designated Business Manager must attend Franchisor's biennial franchise convention and any other mandatory annual conferences at such locations as Franchisor may reasonably designate, and Franchisee will pay all salary and other expenses of persons attending, including any conference fees, travel expenses, meals, living expenses and personal expenses. Failure of Franchisee and/or Franchisee's Designated Business Manager to attend any required meeting in its entirety shall be deemed a breach of this Agreement and subject to the Non-Compliance Fee set forth herein in Section 6.5.

- (c) Subject to Section 9.6, any additional required Service or Product introduced into the System by Franchisor must be offered for sale on a continuing basis at the All American Pet Resorts[®] Business at the time and in the manner required by Franchisor. Franchisor will provide at least 30 days prior written notice of any new required Service or Product introduced into the System. All equipment, products, supplies, tools and other items necessary to add the newly required Services or Products must be acquired, installed, and utilized at the time and in the manner required by Franchisor. The marketing of new Services and Products must begin at the All American Pet Resorts[®] Business as reasonably required by Franchisor.
- (d) No service or product, except approved Services or Products, may be offered for sale from the Pet Resort or in the Territory, unless Franchisee receives the prior written consent of Franchisor (which may be granted or denied as Franchisor deems appropriate).
- (e) Only advertising and promotional materials, Services, equipment, tools, inventory, Products, signage, supplies, and uniforms that meet Franchisor's standards and specifications shall be used at the All American Pet Resorts® Business. Advertising and promotional materials, tools, Services, equipment, inventory, Products, signage, supplies and uniforms produced or approved by Franchisor for use by Franchisee may be used only in the manner and during the period specified by Franchisor.
- (f) Equipment, tools, Services, Products, inventory, supplies, signage, uniforms and other items must be added, eliminated, substituted and modified at the All American Pet Resorts[®] Business as soon as possible in accordance with changes in Franchisor's specifications and requirements.
- (g) The All American Pet Resorts[®] Business and everything related to the All American Pet Resorts[®] Business must be maintained in good condition and must be kept clean, neat and sanitary. All maintenance, repairs and replacements reasonably requested by Franchisor or needed in connection with the All American Pet Resorts[®] Business must be promptly made. All employees must be clean and neat in appearance.
- (h) No alterations of the All American Pet Resorts[®] Business materially affecting the image of the All American Pet Resorts[®] Business may be made except at Franchisor's request or approval, and any alterations must strictly conform to specifications and requirements established or approved by Franchisor.
- (i) The All American Pet Resorts[®] Business and the Services provided and Products sold by Franchisee must comply with all applicable federal, state, and local laws, ordinances, rules, regulations and other requirements applicable to animal care, boarding, daycare and grooming. Franchisee must obtain all business licenses and permits required by federal, state and local laws, ordinances, rules and regulations before operating its All American Pet Resorts[®] Business. If Franchisee does not obtain our approval of a location within four (4) months after the mutual execution of the Franchise Agreement or fails to open and operate its All American Pet Resorts[®] Business at the Approved Location pursuant to our specifications within eighteen (18) months after the mutual execution of

the Franchise Agreement, Franchisor may terminate this Franchise Agreement. In addition to laws and regulations that apply to businesses generally, there are consumer protection laws that exist in several states. Certain requirements, including compliance with federal and/or state solicitation telemarketing (for example, the "do not call" registry), email solicitation, privacy and consumer credit and collection laws are generally applicable to all business that sell directly to the end-user. Given that you will be required to accept consumer payment by credit card, you must ensure that you comply with all laws applicable to processing credit card payments, including compliance with the Payment Card Industry Data Security Standard. Additionally, you should be aware that some states have autorenewal statutes which govern memberships which automatically renew and require, for example, that consumers be provided conspicuous instructions on how to cancel such memberships. You must comply with any privacy and data security laws applicable to any data you collect and/or maintain from customers. You must comply with all local, state and federal laws that apply to your Business and to the public. Those laws include Equal Employment Opportunity Commission ("EEOC"), Occupational Safety and Health Administration ("OSHA"), Federal Trade Commission ("FTC"), pricing laws and employment laws. Such employment laws include regulations concerning wage rates, minimum wage and overtime obligations, mandated employee benefits, employment taxes, worker safety, unemployment compensation, workers' compensation, child labor practices, disabled employees and discrimination in employment practices. You will also be subject to the Americans with Disabilities Act which prohibits practices that discriminate against physically and mentally challenged individuals regarding access to public accommodations and employment opportunities. There may be other laws and codes applicable to your business, and we urge you to make further inquiries about those laws and codes.

- (j) The employees, equipment, tools, supplies, inventory, Products, and other items on hand at the All American Pet Resorts[®] Business, must be at all times sufficient to efficiently meet the anticipated volume of business.
- (k) All debts and taxes arising in connection with the All American Pet Resorts[®] Business, except those duly contested in a bona fide dispute, must be paid when due.
- (l) Franchisee will use its best efforts to ensure customer satisfaction; use good faith in all dealings with customers, potential customers, referral sources, suppliers and creditors; respond to customer complaints in a courteous, prompt and professional manner; use its best efforts to promptly and fairly resolve customer disputes in a mutually agreeable manner; and take such actions as Franchisor deems necessary or appropriate to resolve customer disputes and represent the All American Pet Resorts® Brand. Franchisee agrees to participate in periodic focus groups as directed by Franchisor including but not limited to participation in reasonable expenses related to the service.
- (m) Franchisee shall accept all major debit and credit cards and other the forms of payment specified by Franchisor in the Operations Manual as payment.
- (n) Franchisee shall comply with all terms and pay all fees that may be due under a software license agreement for any software Franchisee is required to use in the operation of its All American Pet Resorts[®] Business as prescribed by Franchisor.

- (o) Franchisee shall comply with the advertising requirements set out in Section 12.
 - (p) Franchisee will not use any materials that are false or misleading.
- (q) Franchisee will ensure that all advertising, labeling, packaging and other materials associated with the Services and Products fully conform to all applicable laws and regulations.
- (r) Franchisee will conduct its business operations in accordance with all applicable laws and regulations, including but not limited to, consumer protection laws and regulations. Franchisee will control the quality of the Services and Products to avoid quality problems or product liability claims that could reflect adversely on Franchisee or Franchiser in the minds of consumers
- 9.4 In prescribing standards, specifications, processes, procedures, requirements or instructions under Section 9.2 or any other provision of this Agreement, Franchisor will provide guidance to Franchisee, as required in Franchisor's sole determination, in determining the prices to be charged by Franchisee for Services or Products. Franchisor shall not have control over the day-to-day managerial operations of the All American Pet Resorts[®] Business, and Franchisee shall be free to establish its own prices.
- 9.5 Franchisor and Franchisor's representatives will have the right during business hours to inspect the All American Pet Resorts® Business and all other facilities used for providing Services and selling approved Products. Franchisor and Franchisor's representatives will have the right to observe the manner in which Franchisee is rendering its Services and conducting its operations of the All American Pet Resorts® Business. Franchisor and Franchisor's representatives will have the right to discuss with Franchisee, or other personnel Franchisee may designate, all matters that may pertain to compliance with this Agreement and with Franchisor's standards, specifications, requirements, instructions and procedures and Franchisor may video tape or take photographs of Franchisee's training styles and techniques as it relates to the All American Pet Resorts® Business without Franchisee's approval. Franchisor and Franchisor's representatives will have the right to have any of Franchisor's required Services rendered by any employee at the All American Pet Resorts® Business. Franchisee shall in all respects cooperate with Franchisor's rights under this Section 9.5; provided that Franchisor's exercise of these rights shall not unreasonably interfere with Franchisee's conduct of the All American Pet Resorts® Business.
- 9.6 Franchisee will not be required to offer or sell new Services or Products as set out in Section 9.3(c) if Franchisee demonstrates to Franchisor's reasonable satisfaction that:
 - (a) A substantial capital improvement not contemplated by this Agreement or in the Operations Manual is required, thereby resulting in a material hardship to Franchisee; or
 - (b) A material reduction in Gross Revenues or profitability would result therefrom. For the purposes of this Section 9.6(b), a 40% decrease in Gross Revenues from the average Gross Revenues for the prior 12 months would be considered a material

reduction in Gross Revenues (subject to seasonal factors that may be applicable to the Territory), and a 30% reduction in profitability from the average profitability during the previous 12 months (subject to seasonal factors that may be applicable to the Territory) would be considered a material reduction in profitability.

- 9.7 Franchisor may require Franchisee's compliance with the provisions of this Section 9 even if it does not require such compliance by all franchisees.
- 9.8 If Franchisee is an individual, Franchisee must have day-to-day management of the Business, direct on-site supervision and personally participate in the direct operation of the Business and attend required conferences and meetings. If Franchisee is a corporation or other business entity, if Franchisee has, in Franchisor's sole judgment, insufficient experience in a business similar to the franchise or experience in business management in general, or if Franchisee so desires, then Franchisee shall nominate a Designated Business Manager having the required experience who shall have direct responsibility for all operations of the All American Pet Resorts® Business. Franchisee is required to immediately notify Franchisor in writing of any change in your Designated Business Manager. Any new Designated Business Manager must complete Franchisor's initial training prior to commencing his or her role as Designated Business Manager.
- 9.9 Franchisee shall at all times have sufficient computer skills to operate Franchisee's computer, understand how to utilize any software Franchisor requires to be used in the All American Pet Resorts® Business, and to access email and the Internet.
- 9.10 Franchisee acknowledges and understands that computer systems are vulnerable to computer viruses, bugs, power disruptions, communication line disruptions, Internet access failures, Internet content failures, date-related problems, and attacks by hackers and other unauthorized intruders. Franchisor does not guarantee that information or communication systems supplied by Franchisor or its suppliers will not be vulnerable to these problems. Franchisee acknowledges and agrees that Franchisee is solely responsible for protecting itself from these problems. Franchisee must also take reasonable steps to verify that Franchisee's suppliers, lenders, landlords, customers, and governmental agencies on which Franchisee relies, are reasonably protected. This may include taking reasonable steps to secure Franchisee's systems, including, but not limited to, firewalls, access code protection, anti-virus systems, and use of backup systems.
- 9.11 Franchisee shall acquire, maintain, and upgrade hardware, software, information processing and communication systems, and Internet and other network access providers, as prescribed in the Operations Manual and as modified periodically by Franchisor. Franchisee shall comply with any separate software or other license agreements that Franchisor or its designee use in connection with the System. Franchisee shall utilize Franchisor's required software, proprietary database management and intranet system, when available, as the exclusive means for tracking and maintaining customer, vendor, and lead information, and for such other uses as prescribed by Franchisor periodically in the Operations Manual. Monthly sales and royalty reporting may occur through mandatory software including the automatic draft via electronic transfer of Royalty Fees and Brand Development Fees.
- 9.12 Franchisee shall at all times maintain an active email account and Franchisee, a Designated Business Manager or employee shall check the account at least once each day. If

available, Franchisee shall maintain an email account on Franchisor's proprietary database management and intranet system or as otherwise specified by Franchisor.

- 9.13 Within four (4) months after the date of this Agreement, Franchisee must have obtained an Approved Location. Within eighteen (18) months after the date of this Agreement, Franchisee must be open and operating its All American Pet Resorts® Business pursuant to our specifications. Franchisee and its Designated Business Manager must have completed the training program to the satisfaction of Franchisor prior to commencing operation of its All American Pet Resorts® Business.
- 9.14 Franchisee may not open its All American Pet Resorts® Business until: (1) Franchisor notifies Franchisee in writing that all of Franchisee's obligations have been fulfilled; (2) the initial training program has been completed to Franchisor's satisfaction; (3) all amounts due to Franchisor have been paid; (4) Franchisor has been furnished with copies of all insurance policies and certificates required by Section 13, or other documentation of insurance coverage and payment of premiums that Franchisor may request; (5) Franchisee notifies Franchisor that all approvals and conditions set forth in this Agreement have been met; (6) Franchisee has obtained all necessary permits and licenses; (7) Franchisee has provided Franchisor with a fully executed copy of the Lease for Franchisee's Pet Resort negotiated in accordance with the terms of Section 9.2(b); and, (8) Franchisee has ordered, received and installed all equipment, supplies, inventory, tools, products, uniforms and computer hardware and software required by Franchisor. Franchisee will begin operating the All American Pet Resorts® Business immediately after Franchisor determines that the All American Pet Resorts® Business is ready for opening.

10. PURCHASE OF EQUIPMENT, INVENTORY AND SUPPLIES.

- 10.1 Franchisee must purchase all Products, Services, equipment, tools, inventory, supplies and hardware from Franchisor's designated or approved suppliers, manufacturers and distributors. The standards and specifications for equipment, computer hardware and software, camera equipment, inventory, tools, signage, supplies, Training Facilities, Services and Products required by Franchisor shall be maintained in the Operations Manual or otherwise provided to Franchisee. Franchisor has the right to require Franchisee to discontinue purchasing any Products, Services, equipment, tools, inventory, supplies, hardware or software from a designated or approved supplier, manufacturer or distributor and may designate and approve new suppliers, manufacturers or distributors at any time.
- 10.2 Franchisee acknowledges and agrees that Franchisor may receive from designated and approved suppliers of Franchisee's Products, Services, equipment, tools, inventory, supplies and hardware and software, periodic volume rebates or other revenue as a result of Franchisee's purchases. Franchisee further acknowledges and agrees that Franchisor shall be entitled to keep for its own use and account such rebates and revenue so long as such rebates and revenue do not cause Franchisee to pay a higher price for the same Products, Services, equipment, tools, inventory, supplies and hardware and software, than the price charged by similarly situated suppliers.
- 10.3 The names of Franchisor's designed and approved suppliers, manufacturers and distributors shall be maintained in the Operations Manual. Franchisor reserves the right to approve

all of the Products, supplies, Services, equipment, tools, inventory, hardware and software used in connection with Franchisee's All American Pet Resorts[®] Business.

10.4 If Franchisee desires to utilize any Services or Products that Franchisor has not approved (for services and products that require supplier approval), Franchisee shall first send Franchisor sufficient information, specifications and/or samples for Franchisor to determine whether the service or product complies with its standards and specifications, or whether the supplier meets its criteria. Franchisee shall bear all reasonable expenses incurred by Franchisor in connection with determining whether it shall approve a service, item or supplier. Franchisor will decide within a reasonable time (usually 30 days but in any event within 60 days) after receiving the required information whether Franchisee may purchase or lease such services or items or from such supplier. Approval of a supplier may be conditioned on the supplier's ability to provide sufficient quantity of product; quality of services or products at competitive prices; production and delivery capability; and dependability and general reputation. Nothing in this Section shall be construed to require Franchisor to approve any particular supplier, or to require Franchisor to make available to prospective suppliers, standards and specifications that Franchisor deems confidential.

11. MARKS, COPYRIGHTED WORKS AND OWNERSHIP OF IMPROVEMENTS.

11.1 Franchisee acknowledges and agrees that:

- (a) Franchisor is the owner of all right, title and interest, together with all the goodwill of the Marks. Franchisee further acknowledges that the Marks designate the origin or sponsorship of the System, the All American Pet Resorts[®] Business, and the Products and Services, and that Franchisor desires to protect the goodwill of the Marks and to preserve and enhance the value of the Marks. In the event that Franchisee acquires any rights, title or interest in the Marks, Franchisee agrees to assign and hereby assigns all such rights, title or interest to Franchisor.
- All right, title and interest in and to all materials, including but not limited (b) to, all artwork and designs, created by Franchisor, and used with the Marks or in association with the All American Pet Resorts[®] Business ("Copyrighted Materials") are the property of Franchisor. Additionally, all Copyrighted Materials created by Franchisee or any other person or entity retained or employed by Franchisee are works made for hire within the meaning of the United States Copyright Act and are the property of Franchisor, who shall be entitled to use and license others to use the Copyrighted Materials unencumbered by moral rights. To the extent the Copyrighted Materials are not works made for hire or rights in the Copyrighted Materials do not automatically accrue to Franchisor, Franchisee irrevocably assigns and agrees to assign to Franchisor, its successors and assigns, the entire right, title, and interest in perpetuity throughout the world in and to any and all rights, including all copyrights and related rights, in such Copyrighted Materials, which Franchisee and the author of such Copyrighted Materials warrant and represent as being created by and wholly original with the author. Where applicable, Franchisee agrees to obtain any other assignments of rights in the Copyrighted Materials from another person or entity necessary to ensure Franchisor's right in the Copyrighted Materials as required in this Section 11.1(b).

- (c) Franchisee will never dispute, contest, or challenge, directly or indirectly, the validity or enforceability of the Marks or Copyrighted Materials or Franchisor's ownership of the Marks or Copyrighted Materials, nor counsel, procure, or assist anyone else to do the same, nor will it take any action that is inconsistent with Franchisor's ownership of the Marks or Copyrighted Materials, nor will it represent that it has any right, title, or interest in the Marks or Copyrighted Materials other than those expressly granted by this Agreement.
- (d) Franchisor may decide to apply to register or to register any trademarks or copyrights with respect to the Services, Products and any other products and services and the Copyrighted Materials. Failure of Franchisor to obtain or maintain in effect any such application or registration is not a breach of this Agreement. Franchisee will not, before or after termination or expiration of the Agreement, register or apply to register any of the Marks or any trademark, service mark or logo confusingly similar thereto or any Copyrighted Materials, anywhere in the world.
- (e) Upon Franchisor's request, Franchisee will cooperate fully, both before and after termination or expiration of this Agreement and at Franchisor's expense, in confirming, perfecting, preserving, and enforcing Franchisor's rights in the Marks and Copyrighted Materials, including but not limited to, executing and delivering to Franchisor such documents as Franchisor reasonably requests for any such purpose, including but not limited to, assignments, powers of attorney, and copies of commercial documents showing sale and advertising of the Services and Products and other products and services. Franchisee hereby irrevocably appoints Franchisor as its attorney-in-fact for the purpose of executing such documents.
- (f) All usage of the Marks by Franchisee and any goodwill established by Franchisee's use of the Marks shall inure to the exclusive benefit of Franchisor. This Agreement does not confer any goodwill or other interests in the Marks to Franchisee upon expiration or termination of the Agreement.
- (g) FRANCHISOR MAKES NO REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AS TO THE USE, EXCLUSIVE OWNERSHIP, VALIDITY OR ENFORCEABILITY OF THE MARKS OR COPYRIGHTED MATERIALS. Franchisor is the owner of certain trademarks registered with the United States Patent and Trademark Office. Notwithstanding the foregoing, if Franchisee is required by Franchisor to use one or more substitute trademarks or copyrighted materials, then Franchisor will not be obligated to reimburse Franchisee for the costs of such compliance per Section 11.3(a).

11.2 Franchisee acknowledges and agrees that:

(a) Franchisee's right to use the Marks and Copyrighted Materials are derived solely from this Agreement. Franchisee may use the Marks and Copyrighted Materials only in its operation of the All American Pet Resorts® Business and only in compliance with this Agreement and all applicable standards, specifications, and operating procedures prescribed by Franchisor in the Operations Manual and elsewhere from time to time during

the Operating Term and any Interim Period. Franchisee will make every effort consistent to protect, maintain, and promote the Marks as identifying the System and only the System.

- (b) Any unauthorized use of the Marks or Copyrighted Materials by Franchisee constitutes a breach of this Agreement and an infringement of the rights of Franchisor and in and to the Marks and Copyrighted Materials.
- (c) Franchisee will not use any Marks or portion of any Marks as part of a corporate or trade name, or with any prefix, suffix or other modifying words, terms, designs or symbols, or in any modified form. Franchisee shall obtain such fictitious or assumed name registrations as may be required by Franchisor or under applicable law.
- (d) In order to preserve the validity and integrity of the Marks and Copyrighted Materials licensed herein and to assure that Franchisee is properly employing the same in the operation of its All American Pet Resorts[®] Business, Franchisor or its agents shall have the right of entry and inspection of Franchisee's All American Pet Resorts[®] Business and operating procedures pursuant to Section 9.5.
- (e) Franchisee will safeguard and maintain the reputation and prestige of the Marks and Copyrighted Materials and will not do anything that would tarnish the image of or adversely affect the value, reputation or goodwill associated with the Marks. Franchisee will not do anything that would dilute, directly or indirectly, the value of the goodwill attached to the Marks, nor counsel, procure, or assist anyone else to do the same.
- (f) Franchisee will use the Marks and Copyrighted Materials only in lettering, logos, print styles, forms, and formats, including but not limited to, advertising and promotional materials, invoices, signage, business checks, business cards, invoices, stationery, and promotional items such as clothing, pens, mugs, etc., which have been approved by Franchisor in accordance with this Agreement, and promptly follow instructions regarding the Marks and Copyrighted Materials as provided in the Operations Manual and otherwise given by Franchisor from time to time.
- (g) Franchisee will use the following copyright notice at least once on each piece of advertising, promotional, or other material used in connection with the Products and Services:
 - © (year of first publication). All American Pet Resorts, LLC, All Rights Reserved.
- (h) Franchisee will use the Marks with a superscript "®" or "TM", as specified by Franchisor, unless and until advised by Franchisor to use a different notice.
- 11.3 Franchisee acknowledges and agrees that:
- (a) If, in Franchisor's reasonable determination, the use of Marks or Copyrighted Materials in connection with the Services, Products, other products and services or the All American Pet Resorts[®] Business will infringe or potentially infringe

upon the rights of any third party, weakens or impairs Franchisor's rights in the Marks or Copyrighted Materials, or it otherwise becomes advisable at any time for Franchisor to modify or discontinue of the Marks or Copyrighted Materials then upon notice from Franchisor, Franchisee will immediately terminate or modify such use in the manner prescribed by Franchisor. Franchisor may require Franchisee to use 1 or more additional or substitute trade names, trademarks, service marks or other commercial symbols or copyrighted materials. Franchisee shall be responsible for the costs of compliance with this requirement (including but not limited to the cost of printing new letterhead and business cards). Franchisee will have no rights of damages, offset, or right to terminate this Agreement as a result thereof and Franchisor shall have no liability or obligation whatsoever with respect to Franchisee's modification or discontinuance of any Marks or Copyrighted Materials.

- Franchisee shall notify Franchisor within three (3) days after receiving (b) notice of any claim, demand or cause of action based upon or arising from any attempt by any other person, firm or corporation to use the Marks or any colorable imitation thereof or the Copyrighted Materials. Upon receipt of timely notice of an action, claim or demand against Franchisee relating to the Marks or Copyrighted Materials, Franchisor shall have the sole right, but not the duty, to defend any such action. Franchisor shall have the exclusive right to contest or bring action against any third party regarding the third party's use of any of the Marks or Copyrighted Materials and shall exercise such right as Franchisor deems appropriate. Franchisor shall control all actions but not be obligated to take any action. In any defense or prosecution of any litigation relating to the Marks, Copyrighted Materials or components of the System undertaken by Franchisor, Franchisee shall cooperate with Franchisor, execute any and all documents, and take all actions as may be desirable or necessary in the opinion of Franchisor's counsel, to carry out such defense or prosecution. At Franchisor's option, Franchisee will join in any action, in which case Franchisor shall bear all the out-of-pocket costs of Franchisee for such participation.
- 11.4 All provisions of this Agreement applicable to the Marks and Copyrighted Materials apply to any and all additional trademarks, service marks, commercial symbols and copyrighted materials authorized for use by and licensed to Franchisee by Franchisor after the date of this Agreement.
- 11.5 If Franchisee during the Operating Term of the franchise relationship or any Interim Period, conceives or develops any improvements or additions to the System, Copyrighted Materials, website or any other documents or information pertaining to or relating to the System or the All American Pet Resorts[®] Business, or any new trade names, trade and service marks, logos, or commercial symbols related to the All American Pet Resorts[®] Business or any advertising and promotional ideas or inventions related to the All American Pet Resorts[®] Business (collectively, the "Improvements"), Franchisee shall fully disclose the Improvements to Franchisor, without disclosure of the Improvements to others, and shall obtain Franchisor's written approval prior to using such Improvements. Any such Improvement may be used by Franchise shall assign and does hereby assign to Franchisor, all right, title and interest in and to the Improvements, including the right to grant sublicenses to any such Improvement. Franchisor may make

application for and own copyrights, patents, trade names, trademarks and service marks relating to any such Improvement and Franchisee shall cooperate with Franchisor in securing such rights. Franchisor may also consider such Improvements as the property and Trade Secrets of Franchisor. In return, Franchisor shall authorize Franchisee to utilize any Improvement that may be developed by other franchisees and is authorized generally for use by other franchisees.

12. **ADVERTISING AND PROMOTION.**

- 12.1 Franchisee acknowledges that local advertising is required to advise the public of the All American Pet Resorts® Business. Prior to the first day of the fourth month of the Operating Period, Franchisee will spend a total of \$20,000 on promotional advertising within the Territory ("Start-Up Advertising Period"). Upon the expiration of the Start-Up Advertising Period, and during the remaining Operating Term and any Interim Period, Franchisee shall spend a minimum of 1.0% of the Gross Revenues for the preceding month or \$1,000 per month, whichever is greater ("Individual Advertising Expense"), for advertising and promotion within the Territory. Franchisee may not advertise outside its territory without Franchisor's approval.
- 12.2 During the Operating Term and any Interim Period, Franchisee shall furnish Franchisor an accounting of Franchisee's previous month's expenditures for advertising and promotion on a form approved by Franchisor.
- 12.3 Franchisor will make available to Franchisee all advertising and promotion materials for the All American Pet Resorts® Business which are issued by Franchisor, to the extent available. Franchisee may not develop advertising materials for use in the All American Pet Resorts® Business without Franchisor's written approval. If Franchisor approves the advertising materials prepared by Franchisee, Franchisor may make available to other franchisees such advertising and promotion materials. All advertising must include the following statement: "Franchises Available." Franchisee must pay duplication costs of any advertising or promotion material provided by Franchisor.
- 12.4 Franchisor has formed a national marketing fund ("Brand Development Fund"). On or before the 10th day of each month beginning on the first day of the second month of the Operating Term, Franchisee shall remit (via EFT) 2% of the Gross Revenues for the preceding month or portion thereof to Franchisor ("Brand Development Fee"). Franchisor may increase the Brand Development Fee as provided in this Section 12.4(f) and will provide Franchisee with 30 days prior notice before increasing the Brand Development Fee. No action taken by Franchisee shall diminish Franchisee's obligations to pay the Brand Development Fee to the Brand Development Fund. The Brand Development Fee is in addition to Franchisee's obligations in Section 12.1.
 - (a) Advertising materials and services will be provided to Franchisee through the Brand Development Fund. Franchisor may occasionally provide for placement of advertising, development of promotional materials, and undertaking public relations activities on behalf of the entire System, including franchisees, or on behalf of a particular region, that may not include Franchisee, through the Brand Development Fund. Franchisor reserves the right to use the Brand Development Fee from the Brand Development Fund to place advertising in national media or regional media (including broadcast, print or other

media) in the future. Franchisee acknowledges that the Brand Development Fund is intended to maximize the general brand recognition of the System. Franchisor is not obligated to expend Brand Development Funds on Franchisee's behalf or benefit or expend Brand Development Funds equivalent or proportionate to Franchisee's Brand Development Fees on Franchisee's behalf or benefit.

- (b) The Brand Development Fund will be used to promote the System, Services and/or Products sold by Franchisees and will not be used to sell additional franchises, provided all advertisements shall include the phrase "Franchises Available." Franchisor's accounting and marketing personnel or a representative designated by Franchisor will administer the Brand Development Fund. The Brand Development Fund will collect Brand Development Fees from all franchisees. All payments to the Brand Development Fund must be spent on advertising, public relations, market research, promotion, marketing of goods and services provided by Franchisor, outside vendors, including but not limited to marketing agencies, and administration of the Brand Development Fund, including but not limited to, salaries, overhead, administrative, accounting, collection and legal costs and expenses. The Brand Development Funds will be maintained by Franchisor in a separate account. An annual un-audited financial statement of the Brand Development Fund, at the expense of the Brand Development Fund, will be available 120 days after Franchisor's fiscal year end to Franchisee for review once a year upon request.
- (c) The Brand Development Fees collected by the Brand Development Fund are non-refundable. The Brand Development Fund may be terminated at any time by Franchisor. In the event that the Brand Development Fund is terminated, any remaining balance in the Brand Development Fund will be expended as provided for in Section 12.4(b) or returned to Franchisee on a pro-rata basis.
- (d) Franchisee shall fully participate in all such promotional campaigns, prize contests, special offers, and other programs, national, regional, or local in nature (including the introduction of new Services, Products, new franchises or other marketing programs directed or approved by Franchisor), which are prescribed from time to time by Franchisor. Franchisee shall be responsible for the costs of such participation. In addition, Franchisee shall honor any coupons, gift certificates or other authorized promotional offers of Franchisor at Franchisee's sole cost unless otherwise specified in writing by Franchisor. Franchisee will maintain an adequate supply of marketing brochures, pamphlets and promotional materials as may be required by Franchisor from time to time. The cost for such participation will be applied to Franchisee's Individual Advertising Expense obligations set forth in Section 12.1.
- (e) Franchisor (and any designee of Franchisor) will have no direct or indirect liability or obligation to Franchisee or the Brand Development Fund or otherwise with respect to the management, maintenance, direction, administration or otherwise of the Brand Development Fund. Franchisee and Franchisor agree that their rights and obligations with respect to the Brand Development Fund and all related matters are governed solely by this Agreement and neither this Agreement or the Brand Development Fund creates a trust, fiduciary relationship, or similar arrangement.

- (f) When Franchisor establishes a franchisee advisory council (the "FAC"), Franchisor may consult with and seek the advice of the FAC regarding the administration of Brand Development Fund, changes in the Brand Development Fee and the format for marketing and promotional campaigns conducted by the Brand Development Fund or directly by franchisees. Franchisor, however, is not obligated to follow such advice. The FAC will be composed of members of Franchisor's senior management and franchisee representatives. It is Franchisor's intention that franchisee representatives are elected by Franchisees; provided, however, that until there are a sufficient number of franchisees in the System, Franchisor may appoint Franchisee representatives to the FAC. Franchisor must approve the number of meetings to be held by the FAC and the procedures under which the FAC will operate.
- (g) Franchisor may increase the Brand Development Fee to 3% upon thirty (30) days' notice to Franchisee.
- From time to time, Franchisor may designate a local, regional or national advertising coverage area in which Franchisee's All American Pet Resorts® Business and at least one (1) other All American Pet Resorts[®] Business is located for purposes of developing a cooperative local, regional or national advertising or promotional program ("Advertising Coverage Area"). Franchisee agrees to participate in and contribute its share to such cooperative advertising and promotional programs in Franchisee's Advertising Coverage Area in addition to such contributions and expenditures as required pursuant to Section 12.1 herein. The cost of the program shall be allocated among locations in such area and each Franchisee's or Franchisor Affiliate-owned All American Pet Resorts[®] Business's share shall be in proportion to its sales during the preceding twelve (12) month period, or portion of said period. Said contributions to cooperative advertising promotional programs shall be credited toward the local advertising and promotional expenditure required by Section 12.1, up to a maximum of 3% of Franchisee's Gross Sales. Advertising Coverage Area shall be defined as the area covered by the particular advertising medium (television, radio or other medium) as recognized in the advertising industry. Franchisor may establish regional advertising councils ("Advertising Councils") to administer the cooperative advertising program which shall be comprised of Franchisor, franchisees in the Advertising Coverage Area, and Franchisor Affiliate-owned All American Pet Resorts® Businesses. At the time a program is designated, Franchisor shall submit a list to Franchisee of all operating All American Pet Resorts® Franchises and Franchisor Affiliate-owned All American Pet Resorts® Businesses within Franchisee's Advertising Coverage Area.
- 12.6 From time to time, Franchisor may establish regional advertising councils to administer the cooperative advertising program which will be comprised of Franchisor, franchisees in the Advertising Coverage Area, and company-owned or affiliate-owned All American Pet Resorts[®] Businesses.
 - (a) At the time a program is designated, Franchisor will submit a list to Franchisee of all operating All American Pet Resorts[®] Businesses (including companyowned and affiliate-owned outlets) within Franchisee's Advertising Coverage Area. Franchisor will determine in advance how each Council will be organized and governed, who will administer it, and when it must start operation. If a Council has been established for an Advertising Coverage Area where Franchisee's All American Pet Resorts[®] Business

is located when this Agreement is signed, or if any Council is established during the term of this Agreement, Franchisee must become a member of the Council and abide by the rules of the Council. Franchisor reserves the right to form, change, dissolve or merge any Council.

- (b) If Franchisor has established a Council for Franchisee's Advertising Coverage Area, Franchisee must contribute to the Council the amounts required by its written governing documents. All contributions to the Council will be maintained and administered in accordance with the written documents governing the Council. If there are company-owned and affiliate-owned outlets in Franchisee's Advertising Coverage Area offering products and services similar to the All American Pet Resorts® Business, they will make contributions to the Council equal to the contributions required of the All American Pet Resorts® Businesses within that area. The Council will be operated solely as a conduit for collecting and spending cooperative fees for the purposes outlined above. The Council may not use any advertising or promotional plans or materials without Franchisor's prior approval.
- (c) The amount of contribution, and whether other franchisees must contribute the same amount or at the same rate, will be determined by the members of the Council, subject to Franchisor's approval. Franchisor anticipates that each member will have one vote for each All American Pet Resorts® Business operated by the member within the Advertising Coverage Area subject to the Council. Franchisee will be obligated by the Franchise Agreement to pay any increased contributions even if Franchisee votes against the increase. Each Council will have to prepare an annual financial statement reporting its expenditures for the previous year to its members. If a Council is established, Franchisor will provide Franchisee with a copy of the governing documents and its financial statements.

13. INSURANCE AND INDEMNITY.

- 13.1 Franchisee shall, upon commencement of the Operating Term, purchase and at all times maintain in full force and effect:
 - (a) Insurance policies, in such amounts and on such terms, as prescribed by the Operations Manual, issued by an insurance company acceptable to Franchisor at all times during the Operating Term of this Agreement and any Interim Period. Insurance coverage must include, but is not limited to, comprehensive general liability, combined single limit, automobile, bodily injury and all-risk property damage insurance and all other occurrences against claims of any person, employee, customer, agent or otherwise in an amount per occurrence of not less than such amount set forth in the Operations Manual and adjusted by Franchisor periodically, unemployment and workers compensation insurance and any other additional insurance required by the terms of any Lease or lender for the All American Pet Resorts® Business. Franchisee must maintain Business Interruption Insurance covering at least twelve (12) months of operations. Franchisee must maintain Employment Related Practices insurance with minimum limits of \$2,000,000. All required insurance policies must insure Franchisee, Franchisor, Franchisor's Affiliates, and Franchisor's and Franchisor's respective officers, directors, shareholders, members and all other

parties designated by Franchisor, as additional named insureds against any liability which may accrue against them because of the ownership, maintenance or operation by Franchisee of the All American Pet Resorts® Business. Franchisor shall be an additional interest on all required policies. The policies must also stipulate that Franchisor shall receive a 30 day prior written notice of cancellation and must contain endorsements by the insurance companies waiving all rights of subrogation against Franchisor. Original or duplicate copies of all insurance policies, certificates of insurance, or other proof of insurance acceptable to Franchisor, including original endorsements affecting the coverage required by this Section, shall be furnished to Franchisor together with proof of payment within 10 days of issuance thereof. Franchisee shall also furnish Franchisor with certificates and endorsements evidencing such insurance coverage within 10 days after each of the following events: (i) at all policy renewal periods, no less often than annually; and, (ii) at all instances of any change to, addition to, or replacement of any insurance. The certificates and endorsements for each insurance policy are to be signed by a person authorized by that insurer to bind coverage on its behalf. All certificates and endorsements are subject to approval by Franchisor. Franchisor reserves the right to require complete, certified copies of all required insurance policies. In the event Franchisee fails to obtain the required insurance and to keep the same in full force and effect, Franchisor may, but shall not be obligated to, purchase insurance on Franchisee's behalf from an insurance carrier of Franchisor's choice, and Franchisee shall reimburse Franchisor for the full cost of such insurance, along with a reasonable service charge to compensate Franchisor for the time and effort expended to secure such insurance, within 5 days of the date Franchisor delivers an invoice detailing such costs and expenses to Franchisee. Notwithstanding the foregoing, failure of Franchisee to obtain insurance constitutes a material breach of this Agreement entitling Franchisor to terminate this Agreement or exercise any or a combination of the other default remedies set forth in Section 18 of this Agreement. Franchisee shall also procure and pay for all other insurance required by state or federal law. Franchisor reserves the right to modify minimum insurance requirements at any time by updating the Operations Manual.

- (b) All liability insurance policies procured and maintained by Franchisee in connection with the All American Pet Resorts® Business will require the insurance company to provide and pay for attorneys to defend any legal actions, lawsuits or claims brought against Franchisee, Franchisor, Franchisor's Affiliates and their respective officers, directors, agents, employees, and all other entities or individuals designated by Franchisor as additional insureds.
- 13.2 Franchisee shall, during the Operating Term and any Interim Period and after the termination or expiration of the Franchise Agreement, indemnify and defend Franchisor, its Affiliates and their respective officers, directors and employees, and hold them harmless against all claims, demands, losses, damages (including punitive damages), costs, suits, judgments, penalties, expenses (including reasonable attorneys' fees and amounts paid in settlement or compromise) and liabilities of any kind, whether or not ultimately determined to be meritorious (and including damages suffered by Franchisee or any of its property) (collectively, "Damages") for which they are held liable, or which they incur (including travel, investigation and living

expenses of employees and witness fees) in any litigation or proceeding as a result of or arising out of:

- (a) a breach of this Agreement, or any other agreement between the parties, or any breach of a Lease or other instrument by which the right to occupy any Pet Resort or any other premises used by Franchisee to operate the All American Pet Resorts® Business is held, by Franchisee;
- (b) any injury to, or loss of property of, any person in, or on, the Pet Resort or any other premises used by Franchisee to operate the All American Pet Resorts[®] Business;
- (c) Franchisee's taxes, liabilities, costs or expenses of its All American Pet Resorts[®] Business;
- (d) any negligent or willful act or omission of Franchisee, its employees, agents, servants, contractors or others for whom it is, in law, responsible; and
- (e) any advertising or promotional material distributed, broadcasted or in any way disseminated by Franchisee, or on its behalf, unless such material has been produced, or approved in writing, by Franchisor.
- 13.3 Franchisor Indemnification. So long as Franchisee is not subject to an uncured notice of material default of this Franchise Agreement, Franchisor will indemnify, defend and hold Franchisee harmless from all Damages from third party claims or demands for which Franchisee is held liable, by a final unappealable ruling issued by a court or arbitrator with competent jurisdiction, and such court or arbitrator determines that such Damages were in fact incurred by Franchisee directly and solely from Franchisor's gross negligence, intentional misconduct, or willful act or omission when performing Franchisor's support services under this Agreement to Franchisee. Notwithstanding the foregoing, this indemnification will not apply to claims based on Damages that relate to theories of vicarious liability, including agency, apparent agency or joint employment or Franchisor's failure to compel Franchisee to comply with this Agreement. For purpose of this Section 13.3 "Damages" means losses, costs, suits, judgments, penalties, expenses (including reasonable attorneys' fees and amounts paid in settlement or compromise) and liabilities of any kind.

14. **RELATIONSHIP.**

14.1 Franchisee acknowledges that it is an independent contractor and is not an agent, partner, joint venturer or employee of Franchisor and no training or supervision given by, or assistance from, Franchisor shall be deemed to negate such independence. Neither party is liable or responsible for the other's debts or obligations, nor shall either party be obligated for any damages to any person or property directly or indirectly arising out of the operation of the other party's business authorized by or conducted pursuant to this Agreement. Franchisor and Franchisee agree that no partnership, fiduciary relationship, joint venture or employment relationship exists between them. Franchisee shall conspicuously identify itself in all dealings

with the public as a sole operator that is an entity separate from Franchisor and state that Franchisor has no liability for the All American Pet Resorts® Business being conducted from the All American Pet Resorts® Business location. It is expressly agreed that the parties intend by this Agreement to establish between Franchisor and Franchisee the relationship of franchisor and franchisee. It is further agreed that Franchisee has no authority to create or assume in Franchisor's name or on behalf of Franchisor, any obligation, express or implied, or to act or purport to act as agent or representative on behalf of Franchisor for any purpose whatsoever. Franchisee agrees that it will not hold itself out as the agent, employee, partner or co-venturer of Franchisor. All employees hired by or working for Franchisee shall be the employees of Franchisee and shall not, for any purpose, be deemed employees of Franchisor or subject to Franchisor control. Each of the parties agrees to file its own tax, regulatory and payroll reports with respect to its respective employees and operations, saving and indemnifying the other party hereto of and from any liability of any nature whatsoever by virtue thereof.

- 14.2 Neither party hereto shall make any agreements, representations or warranties (except by Franchisor in advertising as provided herein) in the name of, or an behalf of, the other party; neither party hereto shall be obligated by, nor have any liability for, any agreements, representations or warranties made by the other (except by Franchisor in advertising as provided herein) nor shall Franchisor be liable for any damages to any person or property, directly or indirectly, arising out of the operation of Franchisee's All American Pet Resorts® Business, whether caused by Franchisee's negligent or willful action or failure to act.
- 14.3 Franchisor shall have no liability for Franchisee's obligations to pay any third parties, including without limitation, any product vendors, or any value added, sales, use, service, occupation, excise, Gross Revenues, income, property or other tax levied upon Franchisee, Franchisee's property, the All American Pet Resorts[®] Business or upon Franchisor in connection with the sales made or business conducted by Franchisee (except any taxes Franchisor is required by law to collect from Franchisee with respect to purchases from Franchisor).

15. **RESTRICTIVE COVENANTS.**

15.1 Franchisee acknowledges and agrees that:

- (a) Franchisee's entire knowledge of the operation of the All American Pet Resorts[®] Business, the System, and the concepts and methods of promotion franchised hereunder that it has now or obtains in the future is derived from Franchisor's Confidential Information and Trade Secrets. Franchisee further acknowledges and agrees that all of the Confidential Information and Trade Secrets are the sole property of Franchisor, represent valuable assets of Franchisor and that Franchisor has the right to use the Confidential Information and Trade Secrets in any manner it wishes at any time.
- (b) During the Operating Term and any Interim Period, Franchisee, and Franchisees' owners, Designated Business Managers, and employees who have access to the Confidential Information and Trade Secrets agree that they: (1) will not use the Confidential Information or Trade Secrets in any other business or capacity or for their own benefit; (2) will maintain the absolute confidentiality of the Confidential Information and Trade Secrets; (3) will not make unauthorized copies of any portion of the Confidential

Information and Trade Secrets; and (4) will adopt and implement all reasonable procedures Franchisor periodically requires to prevent unauthorized use or disclosure of the Confidential Information and Trade Secrets including requiring employees, Designated Business Managers, training class attendees, and Franchisee owners who have access to the Confidential Information and Trade Secrets to execute such nondisclosure and non-competition agreements as Franchisor may require periodically, and provide Franchisor, at Franchisor's request, with signed copies of each of those agreements. Franchisor will be named as a third party beneficiary on such nondisclosure and non-competition agreements.

- After the Agreement expires or is terminated, Franchisee, and Franchisee's (c) owners, Designated Business Managers and employees who have access to the Confidential Information and Trade Secrets agree that for a period of three (3) years after the termination or expiration of the Agreement (unless such information is a Trade Secret in which case the requirements in this Section 15.1(c) will remain in place for as long as such information constitutes a Trade Secret) they: (1) will not use the Confidential Information or Trade Secrets in any other business or capacity or for their own benefit; (2) will maintain the absolute confidentiality of the Confidential Information and Trade Secrets; (3) will not make unauthorized copies of any portion of the Confidential Information or Trade Secrets; and (4) will adopt and implement all reasonable procedures Franchisor periodically requires to prevent unauthorized use or disclosure of the Confidential Information and Trade Secrets including requiring written non-disclosure and non-competition agreements for those individuals as Franchisor may require and provide Franchisor, at Franchisor's request, with signed copies of each of those agreements. Franchisor will be named as a third party beneficiary on such nondisclosure and noncompetition agreements.
- (d) Notwithstanding the foregoing, the restrictions on the disclosure and use of the Confidential Information will not apply to the following: (a) Confidential Information in the public domain after it was communicated to Franchisee through no fault of Franchisee, its owners, Designated Business Managers or employees; (b) Confidential Information in Franchisee's possession free of any obligation of confidence at the time it was communicated to Franchisee; or (c) the disclosure of the Confidential Information in judicial or administrative proceedings to the extent that Franchisee is legally compelled to disclose the information, if Franchisee has notified Franchisor before disclosure and used Franchisee's best efforts, and afforded Franchisor the opportunity, to obtain an appropriate protective order or other assurance satisfactory to Franchisor of confidential treatment for the information required to be so disclosed.

15.2 Franchisee covenants and agrees that:

(a) During the Operating Term of this Agreement and any Interim Period thereof, Franchisee, its owners and Designated Business Managers shall not, without the prior written consent of Franchisor, either individually or in a partnership, corporation, limited liability company, joint venture or other business entity or jointly or in conjunction with any person, firm, association, syndicate or corporation, as principal, agent, shareholder, member, partner or in any manner whatsoever, carry on or be engaged in or be concerned with or interested in or advise, lend money to, guarantee the debts or

obligations of or permit its name or any part thereof to be used or employed in any business operating in competition with any business offering professional pet boarding, daycare or grooming services or any business similar to the All American Pet Resorts[®] Business ("Competitive Business") as carried on from time to time during the Operating Term of this Agreement, including any Interim Period thereof.

- (b) Upon termination or expiration of the Operating Term or any Interim Period, or the transfer, sale or assignment of this Agreement by Franchisee, neither Franchisee, the Designated Business Manager, Franchisee's owners nor any entity in which Franchisee owner has an ownership interest will have any direct or indirect interest (i.e. through a relative) as a disclosed or beneficial owner, investor, partner, director, officer, employee, consultant, representative or agent, or in any other capacity, for 3 years, in any Competitive Business in: (1) the Territory or any other franchisee's territory; (2) within 100 miles of the Territory or any other franchisee's territory; or (3) within 100 miles of any Franchisor or Affiliate owned All American Pet Resorts® Business.
- 15.3 During the Operating Term (including any Interim Period) of this Agreement and for a period of three (3) years thereafter, Franchisee, Franchisee's owners, and the Designated Business Manager shall not attempt to attain an unfair advantage over other franchisees or Franchisor or any Affiliates thereof by soliciting or accepting applications for employment from any person who is, at the time of such solicitation or application, employed by Franchisor, other franchisees or any Affiliates, nor shall Franchisee directly or indirectly induce or attempt to induce any such person to leave his or her employment as aforesaid.
- 15.4 If any person restricted by this Section 15 refuses to voluntarily comply with the foregoing obligations, the three (3) year period will commence with the entry of any order of a court or arbitrator enforcing this Section 15.
- 15.5 The parties have attempted in Section 15.2 above to limit Franchisee's right to compete only to the extent necessary to protect Franchisor from unfair competition. The parties hereby expressly agree that if the scope of enforceability of the provision of Section 15.2 is disputed at any time by Franchisee, a court or arbitrator, as the case may be, may modify Section 15.2 to the extent that it deems necessary to make such provision enforceable under applicable law. In addition, Franchisor reserves the right to reduce the scope of said provision without Franchisee's consent, at any time or times, effective immediately upon notice to Franchisee. Franchisee expressly acknowledges that it possesses skills and abilities of a general nature and has other opportunities to exploit such skills. Consequently, enforcement of the covenants set forth above will not deprive Franchisee of the ability to earn a living.
- 15.6 Nothing in this Section 15 shall prevent any active officer of Franchisee or member of Franchisee's family either individually or collectively, from owning not more than a total of 5% of the stock of any company which is subject to the reporting requirements of the U.S. Securities and Exchange Act of 1934, provided that Franchisee or any member of Franchisee's family is otherwise not actively involved in the management or operation of that business and does not serve that business in any capacity other than as a shareholder.

- 15.7 Franchisor must be protected against the potential for unfair competition by Franchisee's use of Franchisor's training, assistance, Confidential Information and Trade Secrets in direct competition with Franchisor. Franchisee further acknowledges that Franchisor would not have entered into this Agreement or shared the Confidential Information, Trade Secrets and other information with Franchisee absent Franchisee's agreement to strictly comply with the provisions of this Section 15. Franchisee acknowledges that as a Franchisee of Franchisor, it will have access to Franchisor's Trade Secrets and Confidential Information and therefore be in a unique position to use the special knowledge gained as a franchisee. Franchisee acknowledges that a breach of the covenants contained in this Section 15 will be deemed to threaten immediate and substantial irreparable injury to Franchisor. Accordingly, Franchisee agrees that Franchisor will have the right, without prior notice to Franchisee, to obtain immediate injunctive relief without limiting any other rights or remedies and without posting a bond.
- 15.8 In the event that Franchisee is not an individual, this Section 15 will also apply to the officers, directors, stockholders, partners, members, trustees, beneficiaries and/or principals of Franchisee, any entity in which principals or owners of Franchisee have an interest, whether or not the entity exists on the Effective Date of this Agreement, Franchisee, and any persons controlled by, controlling or under common control with Franchisee.

16. **ASSIGNMENT.**

- 16.1 Franchisee acknowledges that Franchisor's obligations under this Agreement are not personal. Franchisor shall have the absolute right to unconditionally transfer or assign this Agreement or any of its rights or obligation under this Agreement to any person, corporation or other party.
- 16.2 Franchisor reserves the right to assign the franchise System to anyone including the operator of a competing franchise system. Franchisee acknowledges and agrees that Franchisor may sell its assets, the Marks or the System to any third party of Franchisor's choice; may offer its securities privately or publicly; may merge with or acquire other corporations or be acquired by another corporation; may undertake a refinancing, recapitalization, leveraged buyout, or other economic or financial restructuring; or may terminate or cease to exist or dissolve, in any such case without Franchisee's consent and, provided the transferee expressly assumes and undertakes to perform Franchisor's obligations in all material respects, free of any responsibility or liability whatsoever to Franchisee after the transaction occurs.
- 16.3 With regard to any of the above sales, assignment and dispositions, Franchisee expressly and specifically waives any claims, demands, or damages against Franchisor arising from or related to the transfer of the Marks, assets or the System from Franchisor to any other party.
- 16.4 Franchisee understands and acknowledges that the rights and duties set forth in this Agreement are personal to Franchisee. Accordingly, this Agreement, Franchisee's rights and interests hereunder, the property and assets owned and used by Franchisee in connection with the All American Pet Resorts[®] Business and any shares, stock, membership or interest in any corporation, limited liability company, or other entity having an interest in the All American Pet Resorts[®] Business, shall not be voluntarily or involuntarily, directly or indirectly sold, pledged,

assigned, transferred, shared, subdivided, sub-franchised, encumbered or transferred in any way (including, without limitation, in the event of the death of Franchisee if Franchisee is an individual), in part, in any manner whatsoever without the prior written approval of Franchisor (which approval will not be unreasonably withheld or delayed) and in compliance with all terms of this Section 16. In the event of a transfer, the transferee must spend a total of \$15,000, or less as may be approved by Franchisor in its sole discretion, on promotional advertising within the Territory during the first ninety (90) days after the transfer occurs. Any unauthorized sale, assignment, transfer or other conveyance, by operation of law or otherwise, or any attempt to do so, shall be deemed void and grounds for termination of this Agreement by Franchisor. Franchisee shall be responsible for paying all expenses, including without limitation, attorney's fees, incurred by Franchisor in association with any improper transfer by Franchisee.

- 16.5 With and after each valid assignment of this Agreement pursuant to this Section 16, the assignee or assignees of Franchisee shall be deemed to be Franchisee under this Agreement and will be bound by and liable for all of Franchisee's existing and future obligations. No stockholder in any corporation, member in any limited liability company or partner in any partnership which becomes Principals of a Franchisee entity will possess individual rights under this Franchise Agreement. Only the Franchisee entity obtains the rights of a franchisee.
- 16.6 If Franchisee shall at any time determine to sell, in whole, the All American Pet Resorts[®] Business, Franchisee shall obtain a bona fide, executed, written offer for the All American Pet Resorts[®] Business together with all real or personal property, leasehold improvements and other assets used by Franchisee in its All American Pet Resorts[®] Business ("Purchase Offer") from a responsible, arms' length, and fully disclosed purchaser and shall submit an exact copy of such Purchase Offer to Franchisor. Franchisor will have a right of first refusal to purchase the All American Pet Resorts[®] Business as provided in Section 17.
- 16.7 The proposed transferee must follow the same application procedures as a new franchisee and must meet and satisfy the same standards of character, business experience, credit standing, health, financial strength and any other criteria then used by Franchisor for new franchisees. No transfer or assignment of this Agreement will be approved by Franchisor or be effective unless and until all the following conditions are satisfied:
 - (a) Franchisee being then in full compliance herewith and paying to Franchisor all outstanding debts or amounts owing to Franchisor and any Affiliates or suppliers of Franchisor;
 - (b) the transferee executing Franchisor's then-current form franchise agreement (which, as Franchisor may determine, may have terms equal to the remainder of Franchisee's Operating Term, or may include a new full length Operating Term, but which may otherwise contain provisions substantially different from those contained herein, including a higher royalty and greater expenditures for advertising and promotion than are provided hereunder, and such other documents then customarily used by Franchisor to grant franchises), all other documents as may be reasonably requested by Franchisor;

- (c) Franchisee paying Franchisor's then current non-refundable transfer fee ("Transfer Fee") (which is \$10,000 as of the date of this Agreement but may be changed at any time by Franchisor), such Transfer Fee applies regardless of the percentage of ownership transferred and regardless of whether the ownership interest is transferred to a family member of Franchisee's owners;
- (d) The completion of a commercial building inspection at the sole cost of Franchisee and the submission of a report of the same to Franchisor. Franchisor reserves the right to require that any conditions Franchisor deems significant be addressed as a condition of Franchisor's consent to transfer;
- (e) Franchisee's execution of a general release of Franchisor, including its members, officers, directors, agents and employees and Affiliates from such parties' obligations or interests via trust, individual retirement account, or other retirement vehicle under the Agreement;
- (f) the transferee purchasing all of Franchisee's assets used in the All American Pet Resorts[®] Business in accordance with all applicable bulk sales legislation and assuming all of the liabilities of the All American Pet Resorts[®] Business unless such liabilities have been paid prior to the closing of the transaction of purchase and sale or unless the sale is a sale of shares in the capital stock or membership interest of Franchisee;
- (g) the transferee shall be an individual, corporation, limited liability company, partnership or other business entity having adequate financial resources who shall meet all criteria established by Franchisor for franchisees. The transferee shall also complete Franchisor's then-current training program established by Franchisor for franchisees unless: (i) the transferee is a current franchisee in good standing in the System, or (ii) the transferee is or has been a Designated Business Manager for a period of one (1) year or more of an All American Pet Resorts® Business in good standing;
- (h) the parties to the proposed transaction will have entered a binding agreement subject only to the rights of Franchisor set out in Section 17. Franchisor shall be furnished a copy of this binding agreement, and such agreement shall be subject to Franchisor's approval in writing. Franchisee must advise each prospective transferee of this provision and the other terms of this Agreement;
- (i) the proposed transferee or the stockholders, partners, members or owners of a beneficial interest in a proposed corporation, partnership, limited liability company or other entity transferee, providing jointly and severally such personal guarantees as Franchisor may request, guaranteeing the proposed transferee's performance of its obligations under the agreements to be entered into;
- (j) the proposed transferee shall have demonstrated to Franchisor's satisfaction that it, he or she will meet in all respects Franchisor's standards applicable to new franchisees regarding experience, personal and financial reputation and stability, willingness and ability to devote its, his or her full time and best efforts to the operation of the All American Pet Resorts[®] Business, and any other conditions as Franchisor may

reasonably apply in evaluating new franchisees. Franchisor must be provided all information about the proposed transferee as Franchisor may reasonably require. Because of the confidential information and trade secrets available to a franchisee, no assignment to a competitor of Franchisor will be permitted; and

- (k) the Franchisee paying all costs of Franchisor (in addition to the transfer fee and not exceeding an additional \$25,000) with respect to (i) the granting of its approval, as hereinbefore contemplated, including but not limited to all of its legal costs with respect to the preparation and execution of the above noted Franchise Agreement, and all other documents then customarily used by Franchisor to grant franchises as well as review and analysis of any documentation or issues relating to the underlying proposed transaction; and (ii) the transfer, including but not limited to all professional fees (attorney's fees, broker fees, and the like), leasing expenses, document preparation costs and due diligence. A deposit of \$2,500 is due and payable with the transfer request. Franchisor will provide Franchisee with a written invoice for such expenses.
- (l) the proposed transferee agrees to expend at least \$15,000 in grand reopening advertising, or less as may be approved by Franchisor in its sole discretion. Franchisees obligation to pay these costs or expenses are not dependent upon whether the transfer is approved, and Franchisor may require Franchisee to deposit funds in advance of performing services related to transfer. All invoices for costs and expenses incurred must be paid within fifteen (15) days of receipt of invoice from Franchisor.
- 16.8 Notwithstanding anything to the contrary herein contained, Franchisor shall, upon Franchisee's compliance with such requirements as may from time to time be prescribed by Franchisor (including the obtaining of all necessary approvals to the assignment of leases, if any, of the Pet Resort), consent to an assignment or transfer of Franchisee's right, title and interest in and to this Agreement, and the property and assets owned and used by Franchisee in connection therewith and any other agreement then in effect between Franchisee and Franchisor to a corporation, limited liability company or other business entity which is wholly owned and controlled by Franchisee, subject to the following (provided that such assignment shall in no way release Franchisee from any liability under this Agreement):
 - (a) Contemporaneously with such assignment and thereafter upon the appointment or election of any person as director, officer, partner or manager of such corporation, limited liability company or other business entity, such corporation, limited liability company, partnership or other business entity shall cause each shareholder, partner, member, manager, director(s) and officer(s) of the corporation, limited liability company, partnership or other business entity to execute a written agreement with Franchisor under seal, personally guaranteeing full payment and performance of Franchisee's obligations to Franchisor and individually undertaking to be bound, jointly and severally, by all the terms of this Agreement or any new current form of Franchise Agreement and jointly and severally liable;
 - (b) No shares or interest in the capital of such corporation, limited liability company, partnership or other business entity shall be issued nor shall Franchisee directly

or indirectly, voluntarily or involuntarily, by operation of law or otherwise, sell, assign, transfer, convey, donate, pledge, mortgage or otherwise encumber any such shares or interest or offer or attempt to do so or permit the same to be done without Franchisor's prior written consent;

(c) The corporation shall maintain stop transfer instructions against the transfer of shares or membership interests on its records subject to the restrictions of this Section and shall have all outstanding shares or membership interest certificates endorsed with the following legend printed conspicuously upon the face of each share or certificate:

The transfer of this certificate is subject to the terms and conditions of a certain Franchise Agreement with All American Pet Resorts, LLC. Reference is made to said Franchise Agreement and to the restrictive provisions of the articles of this corporation.

- (d) The articles of incorporation, articles of organization, operating agreement, partnership agreement, shareholder agreement, and by-laws of the corporation, limited liability company, partnership or other business entity shall provide that its objectives or business is confined exclusively to the operation of the All American Pet Resorts® Business as provided for in this Agreement, and recite that the issuance and transfer of any shares, membership interest, partnership interest or other interest is restricted by the terms of this Agreement, and copies thereof shall be furnished to Franchisor upon request;
- (e) Franchisor's consent to a transfer of any interest subject to the restrictions of this Section shall not constitute a waiver of any claim it may have against the assignor, nor shall it be deemed a waiver of Franchisor's right to demand exact compliance with any of the terms of this Agreement by the assignee;
- (f) The corporation, partnership, limited liability company or other business entity shall advise Franchisor and keep Franchisor current as to the names and addresses of the directors, officers, members, partners and shareholder of and those persons financially involved in the corporation, partnership, limited liability company or other business entity; and
- (g) Franchisee agrees to devote its full time and best efforts to manage the day-to-day operations of the All American Pet Resorts[®] Business unless it has an operational partner or Designated Business Manager approved by Franchisor.
- 16.9 Upon the death of Franchisee, shareholder, partner, or member, the rights granted by this Agreement may pass to the next of kin or legatees, provided that Franchisee's legal representatives shall within sixty (60) calendar days of Franchisee's death apply in writing to Franchisor for the right to transfer to the next of kin or legatee Franchisee's rights under this Agreement. Franchisor shall not unreasonably withhold its permission so long as the proposed transferees meet each of the requirements set forth in this Section 16 within 30 days of the receipt of a conditional permission for the transfer.

- 16.10 Any attempt by Franchisee to transfer any of its rights or interest under this Agreement or the License, without having received Franchisor's prior written consent, will constitute a material breach of this Agreement. However, if Franchisee dies and its personal representative does not desire to sell the All American Pet Resorts[®] Business, and if under controlling local law Franchisee's interest in the All American Pet Resorts[®] Business, the License and Agreement are distributable to heirs or legatees who are members of his or her immediate family and who otherwise would qualify as assignees, then such attempted assignment by operation of law will not be deemed in violation of this Agreement, provided that such heirs or legatees accept the conditions imposed on otherwise permitted assignees.
 - 16.11 Franchisee shall not have the right to grant a subfranchise.

17. OPTION TO PURCHASE — RIGHT OF FIRST REFUSAL.

- 17.1 Unless otherwise explicitly provided by this Agreement, Franchisor shall be entitled to exercise the rights provided in this Section immediately upon:
 - (a) The expiration without the extension of Franchisee's rights to operate the All American Pet Resorts® Business or the termination for any reason of the License or this Agreement;
 - (b) Any breach, default or other event that gives Franchisor the right to terminate the License or this Agreement, after expiration of any applicable notice and cure period; or
 - (c) The receipt by Franchisor of a copy of a Purchase Offer. This provision does not modify the rights and obligations under Section 16.
- 17.2 Upon any event described in Section 17.1, Franchisor shall have the option to purchase all of Franchisee's rights, title and interest in the All American Pet Resorts® Business, and all its improvements, furniture, fixtures, equipment and Products, and all of Franchisee's accounts, contract rights, customer and vendor lists, work in progress and other business assets. If Franchisee leases premises from an unaffiliated lessor, or if Franchisor chooses not to purchase Franchisee's (or Franchisee's affiliate's) fee simple interest in the premises, Franchisee agrees (as applicable) at Franchisor's election: (a) to assign Franchisee's leasehold interest (including renewal options) whether affiliated or unaffiliated lessor in the premises to Franchisor or its designee; (b) to enter into a sublease with Franchisor or its designee for the remainder of the lease term on the same terms (including renewal options) as Franchisee's lease; or (c) to lease the premises to Franchisor or its designee for an initial ten (10) year term, with two five (5) year renewal terms (at Franchisor's option), on commercially reasonable terms. Nothing in this Section is meant to modify any rights or obligations under Section 16 of this Agreement.
- 17.3 The purchase price for the assets itemized in Section 17.2 will be, subject to Section 17.4: (i) the current fair market value if Section 17.1(a) or 17.1(b) is applicable; or (ii) the price specified in the Purchase Offer received by Franchisee if Section 17.1(c) is applicable. If Franchisee and Franchisor cannot agree on fair market value within thirty (30) days, Franchisor will select an appraiser to determine the fair market value of the assets. If Franchisee rejects the

fair market value determination made by Franchisor's designated appraiser, Franchisee may, within ten (10) days of receiving the appraisal, select its own appraiser to determine the fair market value of the assets. Within ten (10) days of receiving the appraisal from Franchisee's designated appraiser, Franchisor will have the right to rescind its offer to purchase the assets. If the appraisals set by the two appraisers differ by 10% or less, the fair market value of the assets will be the average of the two appraisals. If, however, the appraisals set by the two appraisers differ by more than 10%, Franchisor and Franchisee will have ten (10) days to reach an agreement regarding the fair market value of the assets. If an agreement cannot be reached within the ten (10) day time period, the two appraisers will select a third appraiser to determine the fair market value of the assets. Unless Franchisor rescinds its offer to purchase within ten (10) days of receiving the third appraisal, the amount set by the third appraiser will be conclusive and binding on Franchisor and Franchisee. Franchisor and Franchisee will each be responsible for all costs associated with the appraiser they select. The cost for the third appraiser will be paid by the party whose designated appraiser's appraisal was the furthest away from the third appraiser's appraisal.

- 17.4 If Franchisor elects to exercise any option to purchase provided in this Section 17, Franchisor will have the right to set off all amounts due from Franchisee under the Franchise Agreement or any other agreements between the parties, any commissions or fees payable to any broker, agent or other intermediary and the cost of the appraisal, if any, against any payment. Franchisee shall also have the right to substitute cash for any other form of consideration specified in the Purchase Offer and to pay in full the entire purchase price at the time of closing.
- Franchisor will notify Franchisee of its intention to exercise or not exercise its rights to purchase ("Notice of Intent") within sixty (60) days following an event described in Section 17.1(a) or 17.1(b) or within thirty (30) days following an event described in Section 17.1(c). The Notice of Intent will specify the assets to be purchased, and the current fair market value as determined by Franchisor if Section 17.1(a) or 17.1(b) is applicable. In the event Franchisor is purchasing the assets pursuant to Sections 17.1(a) or 17.1(b), Franchisee will have 14 days following receipt of Franchisor's Notice of Intent to object to any of the prices specified therein, and any disputes over pricing shall be resolved through appraisal as specified by Section 17.3. If Franchisor declines to exercise its rights under this Section within the thirty (30) or sixty (60) day period described above, as applicable, Franchisee may thereafter sell or dispose of the All American Pet Resorts[®] Business to any third party in the event of a sale under Section 17.1(a) or 17.1(b) or to the third party identified in the Purchase Offer in the event of a sale under Section 17.1(c), but not at a lower price nor on more favorable terms than set forth in the Purchase Offer, if any, or the Notice of Intent and subject to the prior written permission of Franchisor and satisfaction of the other conditions to assignment set forth in Section 16. If the sale to such third party purchaser is not completed within 90 days after Franchisor delivers the Notice of Intent to Franchisee, Franchisor shall again have the right of first refusal provided in this Agreement.
- 17.6 If Franchisor provides Franchisee with its Notice of Intent to exercise its rights under this Section 17, the purchase and sale contemplated in this Section shall be consummated as soon as possible. In the event Franchisor is purchasing the assets pursuant to Sections 17.1(a) or 17.1(b), following the delivery of a Notice of Intent as specified in Section 17.5, Franchisor or Franchisor's designee shall have the immediate right to take possession of the All American Pet

Resorts® Business and to carry on and develop the All American Pet Resorts® Business for the exclusive benefit of Franchisor or its designee.

18. **DEFAULT AND TERMINATION.**

- 18.1 Franchisor shall have the right, at its option, to (i) suspend performance of certain or all of its services to Franchisee during the time period Franchisee is in default of this Agreement; or (ii) terminate this Agreement and all rights granted Franchisee hereunder (subject to the provisions of applicable state law governing franchise termination and renewal), effective upon receipt of notice by Franchisee, addressed as provided in Section 19, upon the occurrence of any of the following events:
 - (a) Franchisee, or its owners, members, or guarantors, intentionally or negligently discloses to any unauthorized person the contents of or any part of Franchisor's Operations Manual, Confidential Information or Trade Secrets of Franchisor;
 - (b) Franchisee voluntarily abandons, or otherwise ceases to operate, the All American Pet Resorts[®] Business for a period of 5 consecutive days, or any shorter period that indicates an intent by Franchisee to discontinue operation of the All American Pet Resorts[®] Business, unless such abandonment is due to fire, flood, earthquake or other similar causes beyond Franchisee's control and not related to the availability of funds to Franchisee;
 - (c) Franchisee becomes insolvent or is adjudicated bankrupt; or any action is taken by Franchisee, or by others against Franchisee under any insolvency, bankruptcy or reorganization act, or if Franchisee makes an assignment for the benefit of creditors, or a receiver is appointed for Franchisee; or fails to pay debts as they come due in the general course of business;
 - (d) Any material judgment (or several judgments which in the aggregate are material) is obtained against Franchisee and remains unsatisfied or of record for 30 days or longer (unless a supersedes or other appeal bond has been filed); or if execution is levied against Franchisee's All American Pet Resorts[®] Business or any of the property used in the operation of the All American Pet Resorts[®] Business and is not discharged within 10 days; or if the real or personal property of Franchisee's All American Pet Resorts[®] Business shall be sold after levy thereupon by any sheriff, marshal or constable;
 - (e) Franchisee or any owner of greater than 20% of Franchisee entity or operator is charged or convicted of, or acknowledges and/or admits to, a felony, a crime involving moral turpitude, or any crime that is reasonably likely, in Franchisor's business judgment, to materially and unfavorably affect the System, Marks, goodwill or reputation thereof;
 - (f) Franchisee fails to pay any amounts due Franchisor or Affiliates within 10 days after receiving notice that such fees or amounts are overdue;

- (g) Franchisee fails or refuses to comply with any System Standard relating to the cleanliness or sanitation of the Pet Resort and does not correct such failure or refusal within five (5) days, unless such failure or refusal relates to a violation of any health, safety or sanitation law, ordinance or regulation, in which case such failure or refusal to comply with such System Standard within 72 hours after written notice thereof is delivered to Franchisee;
- (h) Franchisee misuses or fails to follow Franchisor's directions and guidelines concerning use of the Marks and fails to correct the misuse or failure within 10 days after notification from Franchisor;
- (i) Franchisee has (i) failed to obtain an Approved Location within 4 months after the date of this Agreement or (ii) failed, after diligent pursuit, to open and operate an All American Pets Resorts Business at the Approved Location within 18 months of executing this Agreement; or (iii) Franchisee or its Designated Business Manager has failed to complete the training program to the satisfaction of Franchisor; or
- (j) Franchisee sells, transfers or otherwise assigns the All American Pet Resorts[®] Business, an interest in the All American Pet Resorts[®] Business or Franchisee entity, this Agreement, All American Pet Resorts[®] or a substantial portion of the assets of the All American Pet Resorts[®] Business owned by Franchisee without obtaining Franchisor's prior written approval and complying with the all provisions of Section 16;
- (k) Franchisee submits on 2 or more occasions during the Operating Term or any Interim Period a report, financial statement, tax return, schedule or other information or supporting record which understates its Gross Revenue by more than 2%, unless Franchisee demonstrates that such understatement resulted from inadvertent error;
- (l) Franchisee fails, or refuses, to submit any report, financial statement, tax return, schedule or other information or supporting records required herein, or submits such reports more than 5 days late on 2 or more occasions during the Operating Term or any Interim Period unless due to circumstances beyond the control of Franchisee;
- (m) Franchisee sells or offers for sale any unauthorized merchandise, product or service, engages in any unauthorized business or practice or sells any unauthorized product or service under the Marks or under a name or mark which is confusingly similar to the Marks;
- (n) Franchisee contests in any court or proceeding the validity of or Franchisor's ownership of the Marks or copyrighted materials, however, Franchisee shall not be in breach of this provision in the event Franchisee is compelled by a court of competent jurisdiction to provide testimony, or the like, or any other type of documentation relating to an unaffiliated third party contest or proceeding challenging the validity of Franchisor's ownership of the Marks or copyrighted materials;

- (o) If Franchisee is a corporation, limited liability company, partnership or other business entity, any action is taken which purports to merge, consolidate, dissolve or liquidate such entity without Franchisor's prior written consent;
- (p) Franchisee receives from Franchisor during the Operating Term and any Interim Period 3 or more notices of default regardless whether such notices of default relate to the same or different defaults, or whether such defaults have been remedied by Franchisee; or
- (q) Any misrepresentation under Section 2.9 or any violation of Anti-Terrorism Laws by Franchisee, its Designated Business Manager, its owners, agents or employees.
- 18.2 Franchisor shall have the right, at its option, to (i) suspend performance of certain or all of its services to Franchisee during the time period Franchisee is in default of this Agreement; or (ii) terminate this Agreement (subject to any state laws to the contrary, where state law shall prevail), effective upon 30 days written notice to Franchisee, if Franchisee breaches any other provision of this Agreement and fails to cure the default during such 30 day period. In that event, this Agreement will terminate without further notice to Franchisee, effective upon expiration of the 30 day period. Defaults shall include, but not be limited to, the following:
 - (a) Franchisee fails to maintain the then-current operating procedures and standards established by Franchisor as set forth herein or in the Operations Manual or otherwise communicated to Franchisee;
 - (b) Franchisee fails, refuses or neglects to obtain Franchisor's prior written approval or consent as required by this Agreement;
 - (c) Franchisee fails or refuses to comply with the then-current requirements of the Operations Manual;
 - (d) Franchisee, or any partnership, joint venture, limited liability company, corporation or other business entity in which Franchisee has a controlling equity interest, defaults under any term of the Lease of the Pet Resort or any other premises used by Franchisee to operate the All American Pet Resorts[®] Business, any other franchise agreement with Franchisor or any other agreement material to the All American Pet Resorts[®] Business and such default is not cured within the time specified in such Lease, other franchise agreement or other agreement;
 - (e) Franchisee fails, refuses or neglects to submit a statement of monthly revenues accompanying the Royalty Fee or Individual Advertising Expense or any other report required under the Agreement when due;
 - (f) Franchisee fails, refuses or neglects to accurately report Gross Revenues, sales information or other information required by Franchisor to be reported; or
 - (g) Franchisee fails to comply with any other provision of this Agreement or any specification, standard or operating procedure prescribed by Franchisor and does not

correct such failure within 10 days (or 30 days if this is the first non-compliance or breach) after written notice from Franchisor (which shall describe the action that Franchisee must take) is delivered to Franchisee.

- 18.3 Notwithstanding the foregoing, if the breach is curable, but is of a nature which cannot be reasonably cured within such 30 day period and Franchisee has commenced and is continuing to make good faith efforts to cure the breach during such 30 day period, Franchisee shall be given an additional reasonable period of time to cure the same, but in no event longer than 30 additional days.
- 18.4 A termination of this Agreement by Franchisee shall be deemed to be a termination without cause and a breach hereof, by Franchisee. Franchisee agrees that it shall not, on grounds of an alleged nonperformance by Franchisor of any of its obligations or any other reason, withhold payment of any amount due to Franchisor whatsoever or set off amounts owed to Franchisor under this Agreement, against any monies owed to Franchisee, which right of set off is hereby expressly waived by Franchisee.
- 18.5 No endorsement or statement on any check or payment of any sum less than the full sum due to Franchisor shall be construed as an acknowledgment of payment in full or an accord and satisfaction, and Franchisor may accept and cash such check or payment without prejudice to its right to recover the balance due or pursue any other remedy provided herein or by law. Franchisor may apply any payments made by Franchisee against any past due indebtedness of Franchisee as Franchisor may see fit. Franchisor may set off against any payment due to Franchisee hereunder any outstanding debts of Franchisee to Franchisor, and may, at Franchisor's option, pay Franchisee's trade creditors out of any sum otherwise due to Franchisee.
- 18.6 Franchisee agrees to pay within 5 days of the effective date of termination or expiration of the Franchise all amounts owed to Franchisor, the landlord of the Pet Resort or other premises used in the All American Pet Resorts[®] Business (if applicable) and Franchisee's trade and other creditors which are then unpaid.
- 18.7 All royalty and advertising contributions, all amounts due for goods purchased by Franchisee from time to time from Franchisor or its Affiliates and any other amounts owed to Franchisor or its Affiliates by Franchisee pursuant to this Agreement or any other agreement shall bear interest after the due date at the rate of 18% per annum or the highest rate permitted by law, whichever is lower, both before and after default, with interest on overdue interest at the aforesaid rate. The acceptance of any interest payment shall not be construed as a waiver by Franchisor of its rights in respect of the default giving rise to such payment and shall be without prejudice to Franchisor's right to terminate this Agreement in respect of such default.
- 18.8 Should Franchisee, or any partnership or joint venture or corporation in which Franchisee has a controlling equity interest, be a franchisee pursuant to another Franchise Agreement with Franchisor, respecting another franchised All American Pet Resorts[®] Business using the Marks, or any other agreement with Franchisor, a default under this Agreement shall constitute a default under such other Franchise Agreement and vice versa, with like remedies available to Franchisor. Should such other Franchise Agreement cease to be valid, binding and in full force and effect for any reason, then Franchisor may, at its option, terminate this Agreement

and this Agreement shall be forthwith surrendered by Franchisee and terminated, and likewise should this Agreement cease to be valid binding and in full force and effect for any, reason, Franchisor may at its option terminate the other Franchise Agreement and the other Franchise Agreement shall be forthwith surrendered and terminated. In the event that there is more than one Franchisee, or if Franchisee should consist of more than one legal entity, Franchisee's liability hereunder shall be both joint and several. A breach hereof by one such entity or Franchisee shall be deemed to be a breach by both or all.

- 18.9 Franchisee agrees that upon termination or expiration of this Agreement, it shall take the following action:
 - (a) Immediately discontinue the use of all Marks, signs, structures, forms of advertising, telephone listings, facsimile numbers, e-mail addresses, the Operations Manual, and all materials, Products and Services of any kind which are identified or associated with the System and return all these materials and Products to Franchisor;
 - (b) Immediately turn over to Franchisor all materials, including the Operations Manual, customer lists, records, files, instructions, brochures, advertising materials, agreements, Confidential Information, Trade Secrets and any and all other materials provided by Franchisor to Franchisee or created by a third party for Franchisee relating to the operation of the All American Pet Resorts[®] Business (all of which are acknowledged to be Franchisor's property). Under no circumstances shall Franchisee retain any printed or electronic copies of the Operations Manual, Confidential Information or Trade Secrets or portions thereof upon expiration or termination of this Agreement;
 - Franchisee hereby acknowledges that all telephone numbers, facsimile numbers and Internet addresses used in the operation of the All American Pet Resorts® Business constitute assets of Franchisor, and upon termination or expiration of this Agreement, Franchisee shall take such action within 5 days to cancel or assign to Franchisor or its designee as determined by Franchisor, all Franchisee's right, title and interest in and to Franchisee's telephone numbers, facsimile numbers and Internet addresses and shall notify the telephone company and all listing agencies of the termination or expiration of Franchisee's right to use any telephone number, facsimile number, and Internet and e-mail addresses, and any regular, classified or other telephone directory listing associated with the Marks and to authorize a transfer of same to or at the direction of Franchisor. Franchisee acknowledges as between Franchisor and Franchisee, Franchisor has the sole rights to, and interest in, all telephone numbers, facsimile numbers, directory listings and Internet addresses used by Franchisee to promote the All American Pet Resorts® Business and/or associated with the Marks. Franchisee hereby irrevocably appoints Franchisor, with full power of substitution, as its true and lawful attorney-in-fact, which appointment is coupled with an interest, to execute such directions and authorizations as may be necessary or prudent to accomplish the foregoing. Attachment E evidences such appointment;
 - (d) Make no representation nor state that Franchisee is in any way approved, endorsed or licensed by Franchisor or associated or identified with Franchisor or the System in any manner;

- (e) Immediately take all steps necessary to amend or terminate any registration or filing of any d/b/a or business name or fictitious name or any other registration or filing containing the Marks so as to delete the Marks and all references to anything associated with the System;
 - (f) Provide Franchisor the option to purchase as set forth in Section 17; and
- (g) Comply with the provisions of Sections 11.1(c), 11.1(d) and 15 and the Non-Disclosure and Non-Competition Agreement, including without limitation the non-competition obligations therein.
- 18.10 If, within 30 days after termination or expiration of this Agreement by Franchisor, Franchisee fails to remove all displays of the Marks from the All American Pet Resorts[®] Business, which are identified or associated with the System, Franchisor may enter the All American Pet Resorts[®] Business to effect removal. In this event, Franchisor will not be charged with trespass nor be accountable or required to pay for any displays or materials.
- 18.11 If, within 30 days after termination or expiration of this Agreement Franchisee has not taken all steps necessary to amend or terminate any registration or filing of any business name or d/b/a or any other registration or filing containing the Marks, Franchisee hereby irrevocably appoints Franchisor as Franchisee's true and lawful attorney for Franchisee, and in Franchisee's name, place and stead and on Franchisee's behalf, to take action as may be necessary to amend or terminate all registrations and filings, this appointment being coupled with an interest to enable Franchisor to protect the System.
- 18.12 Termination or expiration of this Agreement shall not affect, modify or discharge any claims, rights, causes of action or remedies which Franchisor may have against Franchisee, or that Franchisee may have against Franchisor, whether such claims or rights arise before or after termination or expiration.
- 18.13 All obligations of the parties hereto which expressly or by their nature survive the expiration or termination of this Agreement shall continue in full force and effect notwithstanding such expiration or termination. In particular, but without limiting the generality of the foregoing, the provisions of Sections 11, 13, 15 and 17, hereof shall survive termination or expiration of this Agreement.
- 18.14 In the event that this Agreement expires or is terminated for any reason whatsoever and Franchisor is the lender under any loan agreement ("Loan") or the holder of any promissory note ("Note") or the holder of any personal property, security interest, chattel mortgage, debenture or mortgage of any nature whatsoever ("Security Interest") from Franchisee concerning assets used at any time by Franchisee in the All American Pet Resorts® Business or which are situated on the All American Pet Resorts® Business premises, such Loan, Note or Security Interest shall, upon the effective date of termination or expiration, immediately become fully due and payable as to all principal and interest so loaned and secured.
- 18.15 If any applicable and binding law or rule of any jurisdiction requires a greater prior notice of the termination of this Agreement than is required hereunder, the prior notice or other

action required by such law or rule shall be substituted for the notice requirements hereof. Such modifications to this Agreement shall be effective only in such jurisdiction and shall be enforced as originally made and entered into in all other jurisdictions.

- 18.16 In the event of termination of the Agreement for any reason whatsoever the parties shall accept the default remedies contained herein as full and final satisfaction of all claims. The parties waive, to the extent permitted by law, any claim against the other for punitive or exemplary damages; except for such punitive or exemplary damages for violation of the Lanham Act, trademark infringement or dilution, unauthorized dissemination of the Confidential Information or Trade Secrets or arising under the indemnification set out in Section 12.
- 18.17 The rights of the parties hereto are cumulative and no exercise or enforcement by a party of any right or remedy hereunder shall preclude the exercise or enforcement by that party of any other right or remedy herein contained, or to which it is entitled by law.
- 18.18 Nothing herein shall prevent Franchisor or Franchisee from seeking injunctive relief to prevent irreparable harm, in addition to all other remedies. If it is necessary for Franchisor to seek preliminary or permanent injunctive relief, Franchisor may do so without a bond.
- 18.19 The parties acknowledge that in the event that the terms of this Agreement regarding termination or expiration are inconsistent with applicable state or federal law, such law shall govern Franchisee's rights regarding termination or expiration of this Agreement.

19. NOTICES.

Any notice of default under this Agreement shall be delivered personally, by courier, or via e-mail (with copy via overnight courier) to the appropriate location as designated below. Any other notice, request, demand, approval, consent or other communication which the parties hereto may be required or permitted to be given hereunder shall be in writing and may be given to the party for whom it is intended by personal delivery, facsimile transmission or delivering it to such party by mailing it by prepaid registered mail, or by recognized overnight delivery or courier services, in the case of Franchisor to:

Notices to Franchisor: ALL AMERICAN PET RESORTS, LLC

41850 West Eleven Mile Road, Suite 202,

Novi, MI 48375

Attention: Chief Operating Officer

AllAmericanPetResorts.com

With a Copy to: Earsa R. Jackson, Esq. (which shall not constitute notice)

Clark Hill PLC

901 Main Street, Ste. 6000

Dallas, Texas 75202

ejackson@clarkhill.com

Notices to Franchisee:	

Any such notice or other document delivered personally or by facsimile transmission shall be deemed to have been received by and given to the addressee on the day of delivery and any such other notice or other document mailed as aforesaid, shall be deemed to have been received by and given to the addressee on the 3rd business day following the date of mailing, and any delivery made by recognized overnight courier services shall be deemed to be delivered the next business day. Any party may at any time give notice in writing to any other party of any change of address.

20. **DISPUTE RESOLUTION.**

- 20.1 The parties will first attempt to resolve any dispute relating to or arising out of this Agreement by negotiation. Franchisor may provide a procedure for internal dispute resolution as set forth in the Operations Manual, and the parties acknowledge and agree that Franchisor may revise this procedure periodically.
- 20.2 To protect from violations that would cause immediate loss and damages, Franchisor and Franchisee shall each have the right to seek from a state or federal court located in Oakland County, Michigan, as more fully set forth in Section 21.1:
 - (a) injunctive relief and any related incidental damages;
 - (b) an action for disputes or claims related to or based on the Marks;
 - (c) enforcement of a covenant not to compete; and
 - (d) issues related to the disclosure of or misuse of Confidential Information or Trade Secrets.
- 20.3 The parties hereby agree that any suit, action or proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Franchise Agreement or the transactions contemplated hereby, whether in contract, tort or otherwise, shall be determined and shall be resolved by binding arbitration (and not by litigation, except with respect to the enforcement of an arbitrator's decision) in accordance with the following:
 - (a) Such arbitration shall be conducted before a single Qualified Arbitrator in the Novi, Michigan area in accordance with then Commercial Arbitration Rules of the American Arbitration Association (or any successor organization) ("<u>AAA</u>") for expedited procedures ("<u>Expedited Procedures</u>"). The term "<u>Qualified Arbitrator</u>" shall mean a person who has at least ten (10) years' experience in a calling reasonably connected with the subject matter of the arbitration and shall be a disinterested and impartial person of recognized competence.

- (b) The party seeking arbitration shall deliver written notice thereof (such notice, an "<u>Arbitration Notice</u>") to the other party (which notice, if so delivered, shall also be deemed to constitute notice thereof to the Company) of such party's desire and election to enter into arbitration to resolve the dispute in question; which notice shall include a brief statement of the nature of the dispute and the relief being sought.
- 20.4 Within thirty (30) days after the giving of an Arbitration Notice, the parties shall attempt to select a Qualified Arbitrator to resolve the dispute described in the Arbitration Notice. If the parties have not resolved the dispute in question or agreed on a single Qualified Arbitrator within said thirty (30) day period, then any disputing party (including the party who delivered the Arbitration Notice) may apply to the local office of the AAA to appoint a Qualified Arbitrator in accordance with the Expedited Procedures to decide the dispute raised in the Arbitration Notice or if the AAA shall not then exist or shall fail, refuse or be unable to act such that the Qualified Arbitrator is not so appointed by the AAA within thirty (30) days after application therefore, then any disputing party (including the party who delivered the Arbitration Notice) may apply to the federal court located in Oakland County, Michigan for the appointment of the Qualified Arbitrator.

The arbitration costs may be charged to the losing party or allocated between the parties as may be determined by the Qualified Arbitrator; provided if no such allocation is made by the Qualified Arbitrator, the arbitration costs shall be charged to the losing party.

- 20.5 Parties to arbitration under this Agreement shall not include, by consolidation, joinder or in any other manner, any person other than Franchisee and any person in privity with or claiming through, in the right of or on behalf of Franchisee or Franchisor, unless both parties consent in writing. To the extent permitted by applicable law, no issue of fact or law shall be given preclusive or collateral estoppel effect in any arbitration hereunder, except to the extent such issue may have been determined in another proceeding between Franchisor and Franchisee or any person in privity with or claiming through, in the right of or on behalf of Franchisee or Franchisor.
- 20.6 The parties agree that any arbitration arising out of a dispute relating to this Agreement is only a matter between Franchisor and Franchisee and no other franchisees. Franchisee agrees not to join or attempt to join other franchisees or licensees.
- 20.7 Except for payments owed by one party to the other, and unless prohibited by applicable law, any legal action or arbitration proceeding brought or instituted with respect to any dispute arising from or related to this Agreement or with respect to any breach of the terms of this Agreement must be brought or instituted within a period of two years from the date of discovery of the conduct or event that forms the basis of the legal action or proceeding.

21. MISCELLANEOUS.

21.1 Except to the extent governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections 1051 et seq.) or other applicable federal law, this Agreement shall be interpreted under the laws of the State of Michigan, and any dispute between the parties shall be governed by and determined in accordance with the substantive laws of the State of Michigan, which laws shall prevail in the event of any conflict of law; provided, however, the parties expressly agree that this Agreement is not intended to confer on any Franchisee that is not a

resident of the State of Michigan the benefit of any Michigan law providing specific protection to franchisees residing or operating in the State of Michigan. Franchisee and Franchisor have negotiated regarding a forum in which to resolve any disputes which may arise between them and have agreed to select a forum in order to promote stability in their relationship. Therefore, if a claim is asserted in any legal proceeding involving Franchisee, its officers or directors and Franchisor, its officers, directors, shareholders, members, employees or Affiliates both parties agree that the exclusive venue for disputes between them shall be in the State of Michigan and each waive any objection either may have to the personal jurisdiction of or venue in the State of Michigan. Franchisee irrevocably submits to the jurisdiction of such courts and waives any objection Franchisee may have to either the jurisdiction or venue in such court.

- 21.2 All provisions of this Agreement are severable and this Agreement shall be interpreted and enforced as if all completely invalid or unenforceable provisions were not contained herein; all partially valid and enforceable provisions shall be enforced to the extent that they are valid and enforceable.
- 21.3 If either party institutes a legal proceeding, including court proceedings or arbitration, and prevails entirely or in part in any action at law or in equity against the other party based entirely or in part on the terms of this Agreement, the prevailing party shall be entitled to recover from the losing party, in addition to any judgment, reasonable attorneys' fees and expenses, court costs and all of the prevailing party's expenses in connection with any action at law or equity, as well as the prevailing party's pre-suit efforts to enforce this Agreement.
- 21.4 No failure, forbearance, neglect or delay of any kind on the part of Franchisor in connection with the enforcement or exercise of any rights under this Agreement shall affect or diminish Franchisor's right to strictly enforce and take full benefit of each provision of this Agreement at any time, whether at law for damages, in equity for injunctive relief or specific performance, or otherwise. No custom, usage or practice with regard to this Agreement by Franchisee or Franchisor's other franchisees shall preclude the strict enforcement of this Agreement in accordance with its literal terms. No waiver by Franchisor of performance of any provision of this Agreement shall constitute or be implied as a waiver of Franchisor's right to enforce that provision at any future time. No interpretation, change, termination or waiver of any provision of this Agreement, and no consent or approval under this Agreement, shall be binding upon Franchisee or Franchisor or effective unless in writing signed by Franchisee and Franchisor's CEO, President or Chief Operating Officer, except that a waiver need be signed only by the party waiving.
- 21.5 This Agreement, together with the Operations Manual, any written related agreements, all Exhibits, Attachments, and the State Addenda attached to the Disclosure Document as Exhibit F, constitutes the entire understanding and agreement between Franchisee and Franchisor and supersedes all prior understandings, whether oral or written, pertaining to this Agreement, the License, the System or All American Pet Resorts® Business. Nothing in this or in any related agreement, however, is intended to disclaim the representations made in the Franchise Disclosure Document Franchisee received.
- 21.6 The headings of the sections hereof are for convenience only and do not define, limit or construe the contents of the sections of such Sections or other Sections. The term

"Franchisee" as used herein is applicable to one or more persons, a corporation, limited liability company, a partnership or other business entity, as the case may be, and the singular usage (where applicable) includes the plural and the masculine and neuter usages (where applicable) include the other and the feminine. The term Lease shall include a sublease, and a renewal or extension of a lease or sublease. Time shall be of the essence of this Agreement and of every part thereof.

- 21.7 When calculating the date upon which or the time within which any act is to be done pursuant to this Agreement, the date which is the reference date in calculating such period shall be excluded; if the last day of such period is a non-business day, the period in question shall end on the next business day. Time shall be of the essence of this Agreement and of every part thereof.
- 21.8 Neither party hereto shall be liable for any loss or damage due to any delay in the due performance of the terms hereof (except for the payment of money) by reason of strikes, lockouts and other labor relations, fires, riots, wars, embargoes and civil commotion, or acts of God ("Force Majeure Event"). Any such delay shall extend performance only so long as such event is in progress except such Force Majeure Event will not affect or change Franchisee's obligation to pay Royalty Fees and Brand Development Fees when due. Notwithstanding the foregoing, if there is a Force Majeure Event, Franchisor, may elect to waive the Royalty Fees and Brand Development Fees during the period of delay caused by the Force Majeure Event or such shorter period.
- 21.9 Franchisee shall execute and deliver such further instruments, contracts, forms and other documents, and shall perform such further acts, as may be necessary or desirable, to carry out, complete and perform all terms, covenants and obligations herein contained. Franchisee hereby irrevocably appoints Franchisor as his attorney, and hereby empowers it to execute such instruments regarding the Marks for and in Franchisee's name in order to give full effect to Sections 11, 13, 16, and 18 of this Agreement. Franchisee hereby declares that the power of attorney herein granted may be exercised during any subsequent legal incapacity on its part.
- 21.10 This Agreement shall be binding upon, and subject to Section 16 hereof, shall inure to the benefit of, Franchisee's successors and permitted assigns.
- 21.11 This Agreement may only be modified or amended by a written document executed by Franchisee and Franchisor. Notwithstanding anything herein to the contrary, Franchisor shall have the right unilaterally to reduce the scope of any covenants of Franchisee contained in this Agreement upon notice to Franchisee, whereupon Franchisee shall comply therewith as so modified.
- 21.12 From time to time, Franchisor shall have the right to delegate the performance of any portion or all of its obligations and duties hereunder to third parties, whether the same are agents of Franchisor or independent contractors which Franchisor has contracted with to provide such services. Franchisee agrees in advance to any such delegation by Franchisor of any portion or all of its obligations and duties hereunder.

22. ACKNOWLEDGEMENT.

BEFORE SIGNING THIS AGREEMENT, FRANCHISEE SHOULD READ IT CAREFULLY WITH THE ASSISTANCE OF LEGAL COUNSEL.

FRANCHISEE ACKNOWLEDGES THAT:

- 1. NO STATEMENT, REPRESENTATION OR OTHER ACT, EVENT OR COMMUNICATION, EXCEPT AS SET FORTH IN THIS DOCUMENT, AND IN ANY DISCLOSURE DOCUMENT SUPPLIED TO FRANCHISEE, IS BINDING ON FRANCHISOR IN CONNECTION WITH THE SUBJECT MATTER OF THIS AGREEMENT; AND
- 2. FRANCHISEE HAD A COMPLETE COPY OF THIS AGREEMENT, WITH ALL BLANKS FILLED IN, IN ITS POSSESSION FOR A PERIOD OF TIME NOT LESS THAN 5 FULL BUSINESS DAYS, DURING WHICH TIME FRANCHISEE HAD THE OPPORTUNITY TO SUBMIT SAME FOR PROFESSIONAL REVIEW AND ADVICE OF FRANCHISEE'S CHOOSING PRIOR TO FREELY EXECUTING THIS AGREEMENT. FRANCHISEE ACKNOWLEDGES THAT IT HAS HAD AMPLE TIME AND OPPORTUNITY TO INVESTIGATE FRANCHISOR'S ALL AMERICAN PET RESORTS® BUSINESS AND TO CONSULT WITH LEGAL AND FINANCIAL ADVISORS OF ITS CHOICE; AND
- 3. FRANCHISEE HAS CONDUCTED AN INDEPENDENT INVESTIGATION OF THE SYSTEM AND RECOGNIZES THAT THE ALL AMERICAN PET RESORTS® BUSINESS VENTURE CONTEMPLATED BY THIS AGREEMENT AND ITS SUCCESS INVOLVES SUBSTANTIAL BUSINESS RISK AND WILL BE LARGELY DEPENDENT UPON THE ABILITY OF FRANCHISEE AS AN INDEPENDENT BUSINESS PERSON AND ITS ACTIVE PARTICIPATION IN THE DAILY AFFAIRS OF THE ALL AMERICAN PET RESORTS® BUSINESS. FRANCHISEE HEREBY ASSUMES THE RESPONSIBILITY FOR ITS SUCCESS OR FAILURE OF THE ALL AMERICAN PET RESORTS® BUSINESS VENTURE; AND
- 4. FRANCHISOR HAS NOT PROVIDED ANY STATEMENT, REPRESENTATION OR OTHER ACT, EVENT OR COMMUNICATION OF ACTUAL, AVERAGE, PROJECTED, FORECASTED OR POTENTIAL PURCHASES, SALE, COST, EARNINGS, INCOME OR PROFITS TO FRANCHISEE OUTSIDE OF ITEM 19 OF THE FRANCHISE DISCLOSURE DOCUMENT; AND
- 5. FRANCHISOR EXPRESSLY DISCLAIMS THE MAKING OF, AND FRANCHISEE ACKNOWLEDGES THAT IT HAS NOT RECEIVED, ANY ASSURANCE, WARRANTY OR GUARANTEE, EXPRESSED OR IMPLIED, AS TO THE POTENTIAL VOLUME, PROFITS, EARNINGS OR SUCCESS OF THE ALL AMERICAN PET RESORTS® BUSINESS VENTURE CONTEMPLATED BY THIS AGREEMENT.

This entire Agreement, including corrections, changes, and all attachments and addenda, will only be binding upon Franchisor when executed or initialed by Franchisor's authorized representative.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first above set forth.

ALL AMERICAN PET RESORTS, LLC By: ______ Title: _____ FRANCHISEE: _____ Date: _____ Individually OR: (if a corporation or partnership) Company Name By: _____

Title:

Date: _____

STATE OF) COUNTY OF)			
Be it remembered, that on this da subscriber, a Notary Public in and			
Resorts, LLC, a limited liability company. (A but which executed the foregoing instrument, and for and on behalf of said business entity, acknowledge and that the signing and execution of said instruments such officer, and the free and voluntary act and purposes in said instrument mentioned).	siness entity himself or had the signing ent is his or h	whose naderself and and executed and execute	me is subscribed to and l as such officer and for ution of said instrument, d voluntary act and deed
IN TESTIMONY WHEREOF, I have he notarial seal on the day and year last aforesaid.	ereunto subsc	cribed my	name and affixed my
My Commission Expires:			
	Notary Pub	olic	
STATE OF) COUNTY OF)			
Be it remembered, that on this day subscriber, a Notary Public in and for said State, 1 to me personally known and known to be the same instrument, and acknowledged the execution theree	personally ape person who	peared se name is	s signed to the foregoing
IN TESTIMONY WHEREOF, I have he notarial seal on the day and year last aforesaid.	ereunto subsc	cribed my	name and affixed my
My Commission Expires:			
	Notary Pub	olic	

STATE OF)	
COUNTY OF)	
subscriber, a Notary Public in and for president/general partner/managing	day of, 20, before me, the said State, personally appeared, the member of, the business entity whose name is subscribed to and which
executed the foregoing instrument, and behalf of said business entity, acknowled	d for himself or herself and as such officer and for and or edged the signing and execution of said instrument, and that
2 2	ament is his or her free and voluntary act and deed as such and deed of said business entity, for the uses and purposes
IN TESTIMONY WHEREOF notarial seal on the day and year last af	, I have hereunto subscribed my name and affixed my foresaid.
My Commission Expires:	
	Notary Public

ATTACHMENT A TO FRANCHISE AGREEMENT

INITIAL FRANCHISE FEE

Initial Franchise Fee – Check one:

	First All American Pet Resorts® Business \$69,500 paid by Franchisee to Franchisor w	vith the execution of this Agree	\$69,500 ement.
	Second All American Pet Resorts® Busines \$50,000 paid by Franchisee to Franchisor w		\$50,000 ement.
FRAN	ICHISOR:	FRANCHISEE:	
ALL AMERICAN PET RESORTS, LLC			
By:		Ву:	
Title:		Title:	

ATTACHMENT B TO FRANCHISE AGREEMENT

TERRITORY

The Territory set forth in Section 3.1 of the Agreement shall be:		
		_
FRANCHISOR:	FRANCHISEE:	
ALL AMERICAN PET RESORTS, LLC		
By:	By:	
Title:	Title:	

ATTACHMENT C TO FRANCHISE AGREEMENT

GUARANTY AND ASSUMPTION OF FRANCHISEE'S OBLIGATIONS

In consideration of, and as an inducement to, the execution of the Franchise Agreement	nt
executed between ("Franchisee") and All American Po	et
Resorts LLC ("Franchisor") on, ("Agreement") each of the	ıe
undersigned hereby personally and unconditionally:	
1. Guarantees to Franchisor and its successors and assigns, for the Operating Term including any Interim Period thereof, that Franchisee shall punctually pay and perform each an every undertaking, agreement and covenant set forth in the Agreement; and	
2. Agrees to be personally bound by and personally liable for the breach of, each an every provision in the Agreement, including but not limited to the terms of Section 15.	ıd
Each of the undersigned waives the following:	
1. Acceptance and notice of acceptance by Franchisor of the foregoing undertaking:	;
2. Notice of demand for payment of any indebtedness or nonperformance of an obligations hereby guaranteed;	ıy
3. Protest and notice of default to any party with respect to the indebtedness of nonperformance of any obligations hereby guaranteed;	or
4. Any right he or she may have to require that any action be brought against Franchisee or any other person as a condition of liability; and	st
5. Any and all other notices and legal or equitable defenses to which he or she may be entitled.	Эe
Each of the undersigned consents and agrees that:	
6. His or her direct and immediate liability under this guaranty shall be joint an several;	ıd
7. He or she shall render any payment or performance required under the Agreement	nt

8. Such liability shall not be contingent or conditioned upon pursuit by Franchisor of any remedies against Franchisee or any other person; and

9. Such liability shall not be diminished, relieved or otherwise affected by any extension of time, credit or other indulgence which Franchisor may from time to time grant to Franchisee or to any other person, including without limitation the acceptance of any partial payment or performance, or the compromise or release of any claims, none of which shall in any

upon demand if Franchisee fails or refuses punctually to do so;

way modify or amend this guaranty, which shall be continuing and irrevocable during the Operating Term, including any Interim Period thereof.

IN WITNESS WHEREOF, each of the undersigned has affixed his or her signature effective on the same day and year as the Agreement was executed.

GUARA	ANTOR(S)	
AFFILI	ATED LESSOR GUARANTOR(S)
-		
-		
-		
-		

ACKNOWLEDGMENT

Franchisee, and its shareholders, members, or partners, as applicable, jointly and severally acknowledge that they have carefully read this Agreement and all other related documents to be executed concurrently or in conjunction with the execution hereof, that they have obtained the advice of counsel in connection with entering into this Agreement, that they understand the nature of this Agreement, and that they intend to comply herewith and be bound hereby.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the first date set forth above.

ACCEPTED on this day of	, 20
FRANCHISOR:	
ALL AMERICAN PET RESORTS, LLC	
By:	
Its	
FRANCHISEE:	
By:	
Its	
an Individual	

AFFILIATED LESSOR GUARANTOR(S):	
By:	
Its	

ATTACHMENT D TO FRANCHISE AGREEMENT

STATEMENT OF OWNERSHIP

Franchisee:			
Trade Name (if diffe	rent from above):	:	
		Form of Ownership (Check One)	
Individual	Partnership	Corporation	_ Limited Liability Company
			h partner* showing percentage owned, hich the partnership was formed.
	, and list the na	mes and addresses	ration, the names and addresses of each of every shareholder* showing what
	and list the name	s and addresses of	date of formation, the name and address every member* and the percentage of
If a Trust, Indinterest holders.	lividual Retiremen	nt Account, or other	retirement vehicle identify all beneficial
of desired Designated Manager must succe duties. If Franchised immediately notify I	d Business Managessfully complete desires to substi	ger. Franchisee acknown Franchisor's initial itute a new Designation; and the propositions.	ess Manager, identify name and address towledges that any Designated Business I training prior to assuming his or her ed Business Manager, Franchisee must sed new Designated Business Manager ior to assuming his or her duties.
			-

Franchisee acknowledges that this Statement of Ownership applies to the All American Pet Resorts® Business authorized under the Franchise Agreement.

Use additional sheets if necessary. Any and all c	hanges to the above information must be reported
to Franchisor in writing, including any change	in interest in ownership of any member. Such
changes are subject to Franchisor's written a	pproval in accordance with Section 16 of the
Franchise Agreement.	
Date	Name

^{*}If any partner, shareholder or member is a partnership or a business entity, include any individual owners of such partnership or business entity. Franchisor may require additional disclosures if any portion of ownership is held by another entity.

ATTACHMENT D-1 TO FRANCHISE AGREEMENT

STATEMENT OF OWNERSHIP AFFILIATED LESSOR

Franchisee:
Trade Name (if different from above):
Form of Ownership (Check One)
Individual Partnership Corporation Limited Liability Company
If a Partnership, provide name and address of each partner* showing percentage owned, whether active in management, and indicate the state in which the partnership was formed.
If a Corporation, give the state and date of incorporation, the names and addresses of each officer and director, and list the names and addresses of every shareholder* showing what percentage of stock is owned by each.
If a Limited Liability Company, give the state and date of formation, the name and address of the manager(s), and list the names and addresses of every member* and the percentage of membership interest held by each member.
If a Trust, Individual Retirement Account, or other retirement vehicle identify all beneficial interest holders.
Franchisee acknowledges that this Statement of Ownership applies to the All American Pet Resorts® Business authorized under the Franchise Agreement.
Use additional sheets if necessary. Any and all changes to the above information must be reported to Franchisor in writing, including any change in interest in ownership of any member. Such changes are subject to Franchisor's written approval in accordance with Section 16 of the Franchise Agreement.
Date Name

*If any partner, shareholder or member is a partnership or a business entity, include any individual owners of such partnership or business entity. Franchisor may require additional disclosures if any portion of ownership is held by another entity.

ATTACHMENT E TO FRANCHISE AGREEMENT

BY AND BETWEEN ALL AMERICAN PET RESORTS, LLC

	AND ("FRANCHISEE")					
EFT AUTHORIZATION AGREEMENT (DIRECT DEBITS)						
LLC ("Company") to initiate debit entries	positor") hereby authorizes All American Pet Resorts, es and/or credit correction entries to the undersigned's dicated below and the depository designated below uant to Company's instructions.					
Depository	Branch					
Address	City, State, Zip Code					
Bank Transit/ABA Number	Account Number					
written notification from Company and Dein such time and in such manner as to afform If an erroneous debit entry is initiated to have the amount of such entry credited to calendar days following the date on which a written notice pertaining to such entry Depositor shall have sent to Depository a entry was in error and requesting Deposite	Il force and effect until Depository has received joint epositor of the Depositor's termination of such authority and Depository a reasonable opportunity on which to act. Depositor's account, Depositor shall have the right to be such account by Depository, if (a) within fifteen (15) in Depository sent to Depositor a statement of account or or (b) 45 days after posting, whichever occurs first, a written notice identifying such entry, stating that such ory to credit the amount thereof to such account. These for may have under federal and state banking laws.					
Depositor	Depository					
By:						
Date:	Date:					

ATTACHMENT F TO FRANCHISE AGREEMENT

COLLATERAL ASSIGNMENT OF TELEPHONE NUMBERS AND TELEPHONE LISTINGS AND INTERNET ADDRESSES

THIS ASSIGNMENT is entered into this day of	, 20	<u>,</u> in
accordance with the terms of the All American Pet Resorts® Franchise Agreement	"Franc	hise
Agreement") between ("Franchisee")	and	All
American Pet Resorts, LLC ("Franchisor"), executed concurrently with this Assignr	nent, ui	nder
which Franchisor granted Franchisee the right to own and operate an All American P	et Reso	orts®
Business ("Franchise Business") located at		_•

FOR VALUE RECEIVED, Franchisee hereby assigns to Franchisor (1) those certain telephone numbers and regular, classified or other telephone directory listings (collectively, the "Telephone Numbers and Listings") and (2) those certain Internet website addresses ("URLs") associated with Franchisor's trade and service marks and used from time to time in connection with the operation of the Franchise Business at the address provided above. This Assignment is for collateral purposes only and, except as specified herein, Franchisor shall have no liability or obligation of any kind whatsoever arising from or in connection with this Assignment, unless Franchisor shall notify the telephone company and/or the listing agencies with which Franchisee has placed telephone directory listings (all such entities are collectively referred to herein as "Telephone Company") and/or Franchisee's Internet service provider ("ISP") to effectuate the assignment pursuant to the terms hereof.

Upon termination or expiration of the Franchise Agreement (without the extension of Franchisee's rights to operate the Franchise Business), Franchisor shall have the right and is hereby empowered to effectuate the assignment of the Telephone Numbers and Listings and the URLs, and, in such event, Franchisee shall have no further right, title or interest in the Telephone Numbers and Listings and the URLs, and shall remain liable to the Telephone Company and the ISP for all past due fees owing to the Telephone Company and the ISP on or before the effective date of the assignment hereunder.

Franchisee agrees and acknowledges that as between Franchisor and Franchisee, upon termination or expiration of the Franchise Agreement, Franchisor shall have the sole right to, and interest in, the Telephone Numbers and Listings and the URLs, and Franchisee irrevocably appoints Franchisor as Franchisee's true and lawful attorney-in-fact, which appointment is coupled with an interest, to direct the Telephone Company and the ISP to assign same to Franchisor, and execute such documents and take such actions as may be necessary to effectuate the assignment. Upon such event, Franchisee shall immediately notify the Telephone Company and the ISP to assign the Telephone Numbers and Listings and the URLs to Franchisor. If Franchisee fails to promptly direct the Telephone Company and the ISP to assign the Telephone Company and the ISP to effectuate the assignment contemplated hereunder to Franchisor. The parties agree that the Telephone Company and the ISP may accept Franchisor's written direction, the Franchise Agreement or this Assignment as conclusive proof of Franchisor's exclusive rights in and to the Telephone Numbers and Listings and the URLs upon such termination or expiration

and that such assignment shall be made automatically and effective immediately upon Telephone Company's and ISP's receipt of such notice from Franchisor or Franchisee. The parties further agree that if the Telephone Company or the ISP requires that the parties execute the Telephone Company's or the ISP's assignment forms or other documentation at the time of termination or expiration of the Franchise Agreement, Franchisor's execution of such forms or documentation on behalf of Franchisee shall effectuate Franchisee's consent and agreement to the assignment. The parties agree that at any time after the date hereof they will perform such acts and execute and deliver such documents as may be necessary to assist in or accomplish the assignment described herein upon termination or expiration of the Franchise Agreement.

ASSIGNEE:	ASSIGNOR:
ALL AMERICAN PET RESORTS, LLC	
By:	Ву:
Its:	Its:

ATTACHMENT G TO FRANCHISE AGREEMENT

LEASE ADDENDUM

This	Addendum to Lease, dated		, 20	, is e	entered into b	y and
between	("Less	sor"),				
("Lessee"), a	nd All American Pet Resorts, LLC ("Francl	hisor").			
A.	The parties hereto have entered into, 20, and pertaining to the premise				,	
("Lease").	-					

- B. Lessor acknowledges that Lessee intends to operate an All American Pet Resorts[®] franchise from the leased premises ("Premises") pursuant to a Franchise Agreement ("Franchise Agreement") with All American Pet Resorts, LLC ("Franchisor") under the name "All American Pet Resorts[®]" or other name designated by Franchisor (herein referred to as "Franchised Business").
- C. The parties now desire to amend the Lease in accordance with the terms and conditions contained herein.

NOW, THEREFORE, it is hereby mutually covenanted and agreed between Lessor, Lessee, and Franchisor as follows:

- 1. <u>Remodeling and Decor.</u> Lessor agrees that Lessee shall have the right to remodel, equip, paint and decorate the interior of the Premises and to display the proprietary marks and signs on the interior and exterior of the Premises as Lessee is reasonably required to do pursuant to the Franchise Agreement and any successor Franchise Agreement under which Lessee may operate a Franchised Business on the Premises.
- 2. Rental Rate Increases. Lessor shall not increase the amount of payments required to be made by Lessee under the Lease without the prior written consent of both Lessee and Franchisor.
- 3. <u>Assignment</u>. Lessee shall have the right to assign all of its right, title and interest in the Lease to Franchisor or Franchisor's parent, subsidiary, or affiliate, (including another franchisee) at any time during the term of the Lease, including any extensions or renewals thereof, without first obtaining Lessor's consent in accordance with the Collateral Assignment of Lease attached hereto as Attachment G-1. However, no assignment shall be effective until such time as Franchisor or its designated affiliate gives Lessor written notice of its acceptance of the assignment, and nothing contained herein or in any other document shall constitute Franchisor or its designated subsidiary or affiliate a party to the Lease, or guarantor thereof, and shall not create any liability or obligation of Franchisor or Franchisor's parent unless and until the Lease is assigned to, and accepted in writing by, Franchisor or Franchisor's parent, subsidiary or affiliate. In the event of any assignment, Lessee shall remain liable under the terms of the Lease. Franchisor shall

have the right to reassign the Lease to another franchisee without the Landlord's consent in accordance with Section 4(a).

4. Default and Notice.

- (a) In the event there is a default or violation by Lessee under the terms of the Lease, Lessor shall give Lessee and Franchisor written notice of the default or violation within a reasonable time after Lessor receives knowledge of its occurrence. If Lessor gives Lessee a default notice, Lessor shall contemporaneously give Franchisor a copy of the notice. Franchisor shall have the right, but not the obligation, to cure the default. Franchisor will notify Lessor whether it intends to cure the default and take an automatic assignment of Lessee's interest as provided in Paragraph 4(a). Franchisor will have an additional fifteen (15) days from the expiration of Lessee's cure period in which it may exercise the option, but it is not obligated to cure the default or violation.
- (b) All notices to Franchisor shall be sent by registered or certified mail, postage prepaid, or by a recognized overnight courier or delivery services to the following address:

All American Pet Resorts, LLC 41850 West Eleven Mile Road, Suite 202 Novi, MI 48375 Attention: President

Franchisor may change its address for receiving notices by giving Lessor written notice of the new address. Lessor agrees that it will notify both Lessee and Franchisor of any change in Lessor's mailing address to which notices should be sent.

(c) Following Franchisor's approval of the Lease, Lessee agrees not to terminate, or in any way alter or amend the same during the Operating Term of the Franchise Agreement or any Interim Period thereof without Franchisor's prior written consent, which consent shall be granted or denied as Franchisor deems appropriate, and any attempted termination, alteration or amendment shall be null and void and have no effect as to Franchisor's interests thereunder; and a clause to the effect shall be included in the Lease.

5. Termination, Expiration or Non-renewal.

- (a) Upon Lessee's default and failure to cure the default within the applicable cure period, if under either the Lease or the Franchise Agreement or Lessee's failure to exercise its renewal options under the Lease, Franchisor will, at its option, have the right, but not the obligation, to take an automatic assignment of Lessee's interest and at any time thereafter to reassign the Lease to a new franchisee without Landlord's consent and to be fully released from any and all liability to Landlord upon the reassignment, provided Franchisee agrees to assume Lessee's obligations and the Lease
- (b) Upon the expiration or termination of either the Lease or the Franchise Agreement, Landlord will cooperate with and assist Franchisor in securing possession of the Premises and if Franchisor does not elect to take an assignment of the Lessee's interest, Lessor

will allow Franchisor to enter the Premises, without being guilty of trespass and without incurring any liability to Lessor, to remove all signs, awnings, and all other items identifying the Premises as a Franchised Business and to make other modifications (such as repainting) as are reasonably necessary to protect the All American Pet Resorts® marks and system, and to distinguish the Premises from a Franchised Business. In the event Franchisor exercises its option to purchase assets of Lessee, Lessor shall permit Franchisor to remove all the assets being purchased by Franchisor.

6. <u>Consideration; No Liability</u>.

- (a) Lessor hereby acknowledges that the provisions of this Addendum to Lease are required pursuant to the Franchise Agreement under which Lessee plans to operate its business and Lessee would not lease the Premises without this Addendum. Lessor also hereby consents to the Collateral Assignment of Lease from Lessee to Franchisor as evidenced by Attachment G-1.
- (b) Lessor further acknowledges that Lessee is not an agent or employee of Franchisor and Lessee has no authority or power to act for, or to create any liability on behalf of, or to in any way bind Franchisor or any affiliate of Franchisor, and that Lessor has entered into this Addendum to Lease with full understanding that it creates no duties, obligations or liabilities of or against Franchisor or any affiliate of Franchisor.
- 7. <u>Sales Reports</u>. If requested by Franchisor, Lessor will provide Franchisor with whatever information Lessor has regarding Lessee's sales from its Franchised Business.
- 8. <u>Amendments</u>. No amendment or variation of the terms of the Lease or this Addendum to the Lease shall be valid unless made in writing and signed by the parties hereto.
- 9. <u>Reaffirmation of Lease</u>. Except as amended or modified herein, all of the terms, conditions and covenants of the Lease shall remain in full force and effect and are incorporated herein by reference and made a part of this Agreement as though copies herein in full.
- 10. <u>Beneficiary</u>. Lessor and Lessee expressly agree that Franchisor is a third-party beneficiary of this Addendum.

IN TESTIMONY WHEREOF, witness the signatures of the parties hereto as of the day, month and year first written above.

LESSOR:	LESSEE:
By:	By:
Title:	Title:

FRANCHI	SOR: ALL AM	ERICAN PET I	RESORTS, LLC
			<u> </u>
Ву:			
Title:			

ATTACHMENT G-1 TO FRANCHISE AGREEMENT

COLLATERAL ASSIGNMENT OF LEASE

FO	R VA	LUE RECEIV	ED, as of	the	day of		, 20	0 ("Effe	ctive
Date"), the	unde	rsigned,					("Assignor")	, hereby ass	igns,
transfers a	nd set	s over unto All	America	n Pet Resor	ts, LLC	C ("Assi	gnee") all of	Assignor's r	ight,
title and in	terest	as tenant, in, to	and unde	er that certain	in lease	, a copy	of which is a	ttached here	to as
Exhibit	A	("Lease")	with	respect	to	the	premises	located	at
			·	This Colla	teral A	<u>ssignme</u>	ent of Lease ("	Assignment	<u>t")</u> is
for collate	ral pu	rposes only an	d except	as specified	d herei	n, Assig	gnee shall hav	e no liabili	ty or
obligation	of an	y kind whatso	ever arisi	ng from or	in cor	nection	with this As	signment u	nless
Assignee s	hall ta	ike possession of	of the pre	mises demi	sed by	the Leas	se pursuant to	the terms he	ereof
and shall a	ssume	the obligations	s of Assig	gnor thereur	ider.				

Assignor represents and warrants to Assignee that it has full power and authority to so assign the Lease and its interest therein and that Assignor has not previously, and is not obligated to, assign or transfer any of its interest in the Lease or the premises demised thereby.

Upon a default by Assignor under the Lease or under that certain franchise agreement for an All American Pet Resorts® Franchised Business between Assignee and Assignor ("Franchise Agreement"), or in the event of a default by Assignor under any document or instrument securing the Franchise Agreement, Assignee shall have the right and is hereby empowered to take possession of the premises demised by the Lease, expel Assignor therefrom, and, in the event, Assignor shall have no further right, title or interest in the Lease.

Assignor agrees it will not suffer or permit any surrender, termination, amendment or modification of the Lease without the prior written consent of Assignee. Through the Operating Term of the Franchise Agreement and any Interim Period thereof, Assignor agrees that it shall elect and exercise all options to extend the term of or renew the Lease not less than thirty (30) days before the last day that said option must be exercised, unless Assignee otherwise agrees in writing. Upon failure of Assignee to otherwise agree in writing, and upon failure of Assignor to so elect to extend or renew the Lease as stated herein, Assignor hereby irrevocably appoints Assignee as its true and lawful attorney-in-fact, which appointment is coupled with an interest, to exercise the extension or renewal options in the name, place and stead of Assignor for the sole purpose of effecting the extension or renewal.

[Signatures on following page]

IN WITNESS WHEREOF, Assignor and Assignee have signed this Collateral Assignment of Lease as of the Effective Date first above written.

ASSIGNOR:
By:
Its:
ASSIGNEE:
ALL AMERICAN PET RESORTS, LLC, a Michigan limited liability company
By: Its:

EXHIBIT A

LEASE With Respect to Premises Located at:

._____

(To Be Attached)

ADDENDUM RELATING TO

ALL AMERICAN PET RESORTS, LLC

FRANCHISE AGREEMENT

THIS ADDENDUM (Addendum) is made and entered into onALL AMERICAN PET RESORTS, LLC, a Michigan limited liability comp West Eleven Mile Road, Suite 202, Novi, MI 48375 ("Franchisor"), located at	any, located at 41850 and ("Franchisee")
Recitals. Franchisor and Franchisee entered into a Franchise Agreement 20, ("Franchise Agreement"). The Franchisee agreed among other things to a franchise located at Franchi a lender a loan ("Loan) in which funding is provided with the assistance of the Business Administration ("SBA"). SBA requires the execution of this Addend obtaining the SBA assisted financing.	o operate and maintain see has obtained from e United States Small
NOW, THEREFORE, in consideration of the mutual promises belovaluable considerations in hand paid by each of the parties to the others, the rof which the parties acknowledge, the parties agree as follows:	_
 The Franchise Agreement is in full force and effect, and Franchi notice of default to Franchisee under the Franchise Agreement on the date hereof. 	
 Franchisor will not unreasonably withhold, delay or condition proposed transfer or assignment by Franchisee which requires under Section 16.4 of the Franchise Agreement. 	
• If the franchisor must operate the business under Section Agreement, Franchisor will operate the business for 90 necessary for up to one year, and the Franchisor will periodic with the franchisee or its heirs.	days, renewable as

SBA no longer has any interest in the Loan.

This Addendum automatically terminates on the earliest to occur of the following: (i) a Termination occurs under the Franchise Agreement; (ii) the Loan is paid; or (iii)

IN WITNESS WHEREOF, the parties hereto have duly signed and executed this Addendum as of the day and year first above written.

FRANCHISOR:	FRANCHISEE:
ALL AMERICAN PET RESORTS, LLC	
By:	By:
Print Name:	Print Name:
Title:	Title:



EXHIBIT C ALL AMERICAN PET RESORTS, LLC LIST OF CURRENT FRANCHISEES AND AFFILIATE-OWNED OUTLETS

EXHIBIT C

LIST OF FRANCHISEES

As of the date of this Disclosure Document, we have signed franchise agreements with the following persons.

CURRENT OPEN FRANCHISES:

<u>Florida</u>

Amy Moore All American Pet Resorts® Punta Gorda 6101 Duncan Road. Punta Gorda, FL 33950 Phone: 941-916-9655 www.aaprpuntagorda.com

Tim Towles
Richard Kresge
Kathy Kresge
All American Pet Resorts® Fort Myers
14573 Global Parkway
Fort Myers, FL 33913
Phone: 239-939–3647
www.aaprfortmyers.com

Michigan

Sarah Ignash David Ignash All American Pet Resorts[®] Lakeshore 20286 Cornillie Drive Roseville, Michigan 48066 Phone: 586-285-5100 www.aaprlakeshore.com

Rick Ruhland All American Pet Resorts® Canton 7320 North Haggerty Road Canton, MI 48187 Phone: 734-455-3647

www.aaprcanton.com

Sarah Ignash David Ignash All American Pet Resorts® Rochester Hills* 2736 Product Drive Rochester Hills, MI 48307

Phone: 248-299-3647

www.aaprrochesterhills.com

Amanda Luther Breana Luther All American Pet Resorts® Shelby Township 6240 26 Mile Road Shelby Township, MI 48316 Phone: 586-231-7297

Phone: 586-231-7297 www.aaprshelbytwp.com

www.aaprroyaloak.com

All American Pet Resorts® Royal Oak (affiliate owned) Art & Yolanda Rimbold Ann Marie Peskowski 4227 Delemere Blvd. Royal Oak, MI 48073 Phone: 248-554-0100

New Jersey

Glenn Shapiro All American Pet Resorts® North Brunswick 330 Thalia Street North Brunswick, NJ 08902 Phone: 732-709-7387

www.aaprnorthbrunswick.com

North Carolina

Paul Norman Bassler and Jennifer Jones Bassler Big Dog House Incorporated d/b/a All American Pet Resorts New Bern 9409 Huey Blvd. Waxhaw, NC 28173 jennyjonesbassler@gmail.com

Andy & Bethany Brehm¹
All American Pet Resorts[®] Greensboro
1046 NC-68
Oak Ridge NC 27310
Phone: 713 542 6830

Phone: 713-542-6830 www.aaprgreensboro.com

David & Nicki Webb² All American Pet Resorts[®] Asheville 1450 Sand Hill Rd Buncombe, NC 28715 Phone: 318-773-7389

Phone: 318-773-7389 www.aaprasheville.com

Allen & Linda Parris³
All American Pet Resorts[®] Marvin
TBD
Marvin, NC

Phone: 704-773-4358 www.aaprmarvin.com

Texas

James & Keith Ezell All American Pet Resorts[®] Dallas 2737 Mockingbird Lane Dallas, TX 75235 Phone: 469-335-0400

Fax: 469-335-0401 www.aaprdallas.com

Ernesto Flores⁴
All American Pet Resorts[®] Leander
11830 Old 2243 W
Leander, TX 78641
Phone: 786-782-5726
www.aaprleander.com

Note:

¹Andy & Bethany Brehm Projected Opening: 4Q2024

²David & Nicki Webb Projected Opening: 4Q2024

³Allen & Linda Parris Projected Opening: 2Q2025

⁴Ernesto Flores

Projected Opening: 1Q2025

FRANCHISEES WHO LEFT THE SYSTEM DURING OUR LAST FISCAL YEAR:

* Chris Masson All American Pet Resorts® Rochester Hills 2736 Product Drive Rochester Hills, MI 48307 Phone: 248-299-3647 www.aaprrochesterhills.com

(franchisee transferred location first quarter 2024 to Sarah Ignash and David Ignash)

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.



EXHIBIT D ALL AMERICAN PET RESORTS, LLC TRADEMARK-SPECIFIC FRANCHISEE ORGANIZATIONS

TRADEMARK-SPECIFIC FRANCHISEE ORGANIZATIONS

The following is the name, address, telephone number, email address, and Web address (to the extent known) of each trademark-specific franchisee organization that has been created, sponsored, or endorsed by us, and the relationship between the organization and us:

NONE

The following independent franchisee organizations have asked to be included in this Disclosure Document:

NONE



EXHIBIT E ALL AMERICAN PET RESORTS, LLC LIST OF STATE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

EXHIBIT E

LIST OF STATE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

STATE	STATE ADMINISTRATOR	AGENT FOR
		SERVICE OF PROCESS
CALIFORNIA	California Department of Business Oversight	California Commissioner of
	One Sansome Street, Suite 600	Business Oversight
	San Francisco, CA 94104	320 West 4th Street, Suite 750
	415-972-8559	Los Angeles 90013-2344
	1-866-275-2677	1-866-275-2677
CONNECTICUT	Securities and Business Investment Division	Connecticut Banking Commissioner
	Connecticut Department of Banking	Same Address
	260 Constitution Plaza	
	Hartford, CT 06103	
	860-240-8230	
FLORIDA	Department of Agriculture & Consumer	Same
	Services	
	Division of Consumer Services	
	Mayo Building, Second Floor	
	Tallahassee, FL 32399-0800	
	850-245-6000	
GEORGIA	Office of Consumer Affairs	Same
GLORGIA	2 Martin Luther King Drive, S.E.	Same
	Plaza Level, East Tower	
	Atlanta, GA 30334	
	404-656-3790	
HAWAII	Department of Commerce and Consumer	Commissioner of Securities of the State
	Affairs	of Hawaii
	Business Registration Division	Dept. of Commerce and Consumer
	Commissioner of Securities	Affairs
	335 Merchant Street, Room 203	Securities Compliance Branch
	Honolulu, HI 96813	335 Merchant Street, Room 203
	808-586-2722	Honolulu, HI 96813
ILLINOIS	Franchise Division	Illinois Attorney General
	Office of the Attorney General	Same Address
	500 South Second Street	Same Address
	Springfield, IL 62706	
INDIANA	217-782-4465 Securities Commissioner	Indiana Secretary of State
INDIANA	Indiana Securities Division	201 State House
	302 West Washington Street, Room E 111	200 West Washington Street
	Indianapolis, IN 46204	Indianapolis, IN 46204
		mulanapons, nv 40204
IOWA	317-232-6681	Como
IUWA	Iowa Securities Bureau Second Floor	Same
	Lucas State Office Building	
	Des Moines, IA 50319	
	515-281-4441	

STATE	STATE ADMINISTRATOR	AGENT FOR
		SERVICE OF PROCESS
KENTUCKY	Kentucky Attorney General's Office	Same
	Consumer Protection Division	
	1024 Capitol Center Drive	
	Frankfort, KY 40602	
	502-696-5389	
LOUISIANA	Department of Urban & Community Affairs	Same
	Consumer Protection Office	
	301 Main Street, 6th Floor	
	One America Place	
	Baton Rouge, LA 70801	
	504-342-7013 (gen. info.) 504-342-7900	
MAINE	Department of Business Regulations	Same
	State House - Station 35	
	Augusta, ME 04333	
	207-298-3671	
MARYLAND	Office of the Attorney General	Maryland Securities Commissioner
	Securities Division	Same Address
	200 St. Paul Place	
	Baltimore, MD 21202	
	410-576-6360	
MICHIGAN	Michigan Department of Attorney General	Michigan Department of Commerce
	Consumer Protection Division	Corporations and Securities Bureau
	Antitrust and Franchise Unit	Same Address
	525 W. Ottawa Street	
	G. Mennen Williams Building, 1st Floor	
	Lansing, MI 48913	
	517-373-7117	
MINNESOTA	Minnesota Department of Commerce	Minnesota Commissioner of Commerce
	85 7 th Place East, Suite 500	Same Address
	St. Paul, MN 55101	
	651-539-1500	
NEBRASKA	Department of Banking and Finance	Same
	Bureau of Securities/Financial Institutions	
	Division	
	1526 K Street, Suite 300	
	Lincoln, NE 68508-2732	
	P.O. Box 95006	
	Lincoln, Nebraska 68509-5006	
	Tele: 402-471-2171	
NEW HAMPSHIRE	Attorney General	Same
	Consumer Protection and Antitrust Bureau	
	State House Annex	
	Concord, NH 03301	
	603-271-3641	

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
NEW YORK	New York State Department of Law	Secretary of State of New York
	Investor Protection Bureau	99 Washington Avenue
	28 Liberty Street, 21st Floor	Albany, New York 12231
	New York, NY 10005	
	212-416-8236	
NORTH CAROLINA	Secretary of State's Office/Securities Division	Secretary of State
	2 South Salisbury Street	Secretary of State's Office
	Raleigh, NC 27601	Same Address
	919-733-3924	
NORTH DAKOTA	North Dakota Securities Department	North Dakota Securities Commissioner
	600 East Boulevard Avenue	Same Address
	State Capitol, Fifth Floor	
	Bismarck, ND 58505-0510	
	701-328-4712; Fax: 701-328-0140	
ОНЮ	Attorney General	Same
	Consumer Fraud & Crime Section	
	State Office Tower	
	30 East Broad Street, 15th Floor	
	Columbus, OH 43215	
	614-466-8831 or 800-282-0515	
OKLAHOMA	Oklahoma Securities Commission	Same
	2915 Lincoln Blvd.	
	Oklahoma City, OK 73105	
OPEGON	405-521-2451	5.
OREGON	Department of Insurance and Finance	Director
	Corporate Securities Section	Department of Insurance and Finance
	Labor and Industries Building	Same Address
	Salem, OR 96310	
RHODE ISLAND	503-378-4387	Diseases Dhe de Island Denoutre out of
KHUDE ISLAND	Rhode Island Department of Business	Director, Rhode Island Department of
	Regulation Securities Division	Business Regulation Same address
		Same address
	John O. Pastore Center – Building 69-1 1511 Pontiac Avenue	
	Cranston, RI 02920	
	401-222-3048	
SOUTH CAROLINA	Secretary of State	Same
SOUTH CAROLINA	P.O. Box 11350	Same
	Columbia, SC 29211	
	803-734-2166	
SOUTH DAKOTA	South Dakota Department of Labor and	Director of the South Dakota
	Regulation	Division of Insurance, Securities
	Division of Insurance	Regulation
	Securities Regulation	Same Address
	124 S. Euclid Avenue, Suite 104	Tiddless
	Pierre, SD 57501	
	1	
	605-773-3563	

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
TEXAS	Secretary of State	Same
IEAAS	Statutory Documents Section	Same
	P.O. Box 12887	
	Austin, TX 78711-2887	
TITATI	512-475-1769	g
UTAH	Utah Department of Commerce	Same
	Consumer Protection Division	
	160 East 300 South (P.O. Box 45804)	
	Salt Lake City, UT 84145-0804	
	TELE: 801-530-6601	
	FAX:801-530-6001	
VIRGINIA	State Corporation Commission	Clerk of the State Corporation
	Division of Securities and Retail Franchising	Commission
	Tyler Building, 9th Floor	Tyler Building, 1st Floor
	1300 E. Main Street	1300 E. Main Street
	Richmond, VA 23219	Richmond, VA 23219
	804-371-9733	804-371-9051
WASHINGTON	Department of Financial Institutions	Director, Dept. of Financial Institutions
	Securities Division	Securities Division
	150 Israel Rd S.W.	150 Israel Rd. S.W.
	Tumwater, WA 98501	Tumwater, WA 98501
	360-902-8762	· ·
WISCONSIN	Wisconsin Dept. of Financial Institutions	Wisconsin Commissioner of Securities
	Division of Securities	Same Address
	345 W. Washington Avenue, 4th Floor	
	Madison, WI 53703	
	608-266-8557	



EXHIBIT F ALL AMERICAN PET RESORTS, LLC STATE-SPECIFIC ADDENDA

<u>CALIFORNIA ADDENDUM TO</u> FRANCHISE DISCLOSURE DOCUMENT

- 1. THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT.
- 2. Item 2 is amended by the addition of the following language:
 - Neither All American Pet Resorts, LLC nor any person identified in Item 2 of the Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Action of 1934, 15 U.S.C.A.78a, et seq., suspending or expelling such persons from membership in such association or exchange.
- 3. California Business and Professions Code Sections 20000 through 20043 provide rights to the franchise concerning termination, transfer or non-renewal of a franchise. If the franchise agreement contains a provision that is inconsistent with the law, the law will control.
- 4. The franchise agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law (11 U.S.C.A. Section 101 and following).
- 5. The franchise agreement requires binding arbitration. The arbitration will occur in the city of our then-current headquarters (currently, Novi, Michigan) with the costs being borne by you if All American Pet Resorts, LLC prevails. Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedures Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside the State of California.
- 6. Section 31125 of the Franchise Investment Law requires us to give to you a disclosure document approved by the Commissioner of Business Oversight before we ask you to consider a material modification of your franchise agreement.
- 7. You must sign a general release of claims if you renew or transfer your franchise. California Corporations Code Sections 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code Sections 31000 through 31516.) Business and Professions Code Section 200010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 20000 through 20043).
- 8. The franchise agreement contains a covenant not to compete, which extends beyond the termination of the franchise. This provision may not be enforceable under California law.
- 9. The franchise agreement contains a liquidated damages clause. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable.
- 10. OUR WEBSITE HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF BUSINESS OVERSIGHT. ANY COMPLAINTS

CONCERNING THE CONTENT OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF BUSINESS OVERSIGHT AT www.DBO.ca.gov.

ADDENDUM TO THE FRANCHISE AGREEMENT REQUIRED FOR CALIFORNIA FRANCHISEES

This Addendum pertains to franchises sold in the State of California and is for the purpose of complying with California statutes and regulations. Notwithstanding anything which may be contained in the body of the Franchise Agreement to the contrary, the Agreement is amended to include the following:

- 1. Section 15 of the Franchise Agreement contains a covenant not to compete which extends beyond the term of the franchise. This provision may not be enforceable under California law.
- 2. Any capitalized terms that are not defined in this Addendum will have the meaning given them in the Franchise Agreement.
- 3. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

FRANCHISOR: All American Pet Resorts, LLC	FRANCHISEE:
By:	Ву:
Printed Name:	Printed Name:
Title:	Title:
Date:	Date:

MARYLAND ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

<u>Item 1, Additional Disclosures</u>. The following statements are added to Item 1:

The general release required as a condition of renewal shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

Item 17, Additional Disclosures. The following statements are added to Item 17:

The general release required as a condition of assignment/transfer shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

The "Summary" column of Item 17(H) of the Disclosure Document, pertaining to "Cause defined – defaults that cannot be cured" is supplemented to state that any provision in the Franchise Agreement which terminates the franchise upon the bankruptcy of the Franchisee may not be enforceable under federal bankruptcy law.

Franchisee may sue in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law, provided that the Franchise Agreement may provide for arbitration in a forum outside of Maryland.

Any claims arising under the Maryland Franchise Registration and Disclosure Law may be brought within 3 years after the grant of the franchise.

<u>Exhibit H, Statement of Franchisee</u>. The following statement is added to the Statement of Franchisee:

All representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

ADDENDUM TO THE FRANCHISE AGREEMENT REQUIRED FOR MARYLAND FRANCHISEES

		nent ("Franchise Agreement") dated Pet Resorts, LLC ("Franchisor") and ered into simultaneously with the execution	
of the l	Franchise Agreement.	•	
1.	The provisions of this Addendum form an integral part of, and are incorporated into the Franchise Agreement. This Addendum is being executed because: (a) the offer or sale of the franchise to Franchisee was made in the State of Maryland; (b) Franchisee is a resident of the State of Maryland; and/or (c) the Franchised Business will be located or operated in the State of Maryland.		
2.	The following sentence is added to the end of Sections 2 and 22:		
	All representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.		
3.	The following sentence is added to the end of Sections 4.2(c) and 16.7(c):		
	The general release required as a condition of renewal or transfer shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law.		
4.	The following sentence is added to the end of Sections 20.2, 20.3 and 21.1:		
	A franchisee may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.		
5.	Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.		
6.	Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.		
	NCHISOR: .merican Pet Resorts, LLC	FRANCHISEE:	
By:		By:	
Printe	ed Name:	Printed Name:	
Title:	tle: Title:		
Date:			

NEW YORK ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

NEW YORK

1. The following information is added to the cover page of the Franchise Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SERVICES OR INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE OR PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CAN NOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS THAT ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is to be added at the end of Item 3:

With the exception of what is stated above, the following applies to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal, or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature, or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10-year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.

- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation, or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.
- 3. The following is added to the end of the "Summary" sections of Item 17(c), titled "Requirements for a franchisee to renew or extend," and Item 17(m), entitled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687(4) and 687(5) be satisfied.

- 4. The following language replaces the "Summary" section of Item 17(d), titled "**Termination by franchisee**": You may terminate the agreement on any grounds available by law.
- 5. The following is added to the end of the "Summary" sections of Item 17(v), titled "Choice of forum," and Item 17(w), titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or the franchisee by Article 33 of the General Business Law of the State of New York

- 6. Franchise Questionnaires and Acknowledgements--No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 7. Receipts--Any sale made must be in compliance with § 683(8) of the Franchise Sale Act (N.Y. Gen. Bus. L. § 680 *et seq.*), which describes the time period a Franchise Disclosure Document (offering prospectus) must be provided to a prospective franchise before a sale may be made. New York law requires a franchisor to provide the Franchise Disclosure Document at the earlier of the first personal meeting, ten (10) business days before the execution of the franchise or other agreement, or the payment of any consideration that relates to the franchise relationship.

<u>VIRGINIA ADDENDUM TO</u> FRANCHISE DISCLOSURE DOCUMENT

ITEM 17.h of the Franchise Disclosure Document is modified to read as follows:

Under Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the development agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.



EXHIBIT G ALL AMERICAN PET RESORTS, LLC NON-DISCLOSURE AND NON-COMPETITION AGREEMENT

EXHIBIT G

NON-DISCLOSURE AND NON-COMPETITION AGREEMENT

This Non-Disclosure and N	Non-Competition Agreement ("Agreement") is made and
entered into this day of	, 20 by and between All American Pet
Resorts, LLC, a Michigan limited	liability company ("Company"), located at our principal
business address is 41850 West	Eleven Mile Road, Suite 202, Novi, MI, 48375 and
	_ ("Associate"), who resides or has a principal place of
address at	·

RECITALS

- A. The Company is engaged in the business of selling franchises for the operation of a business offering professional pet boarding, daycare and grooming services, and the sale of related products ("Franchise Business"). The Franchise Business is operated under the Company's trademark "All American Pet Resorts®" and other service marks, trademarks, logo types, designs, and other commercial symbols (collectively "Marks").
- B. The Company has developed methods for establishing, operating and promoting Franchise Businesses under the Company's distinctive business format, plans, methods, data, processes, supply systems, marketing systems, formulas, techniques, designs, layouts, operating procedures, Marks and information and know-how of the Company ("Confidential Information" and "Trade Secrets") and such Confidential Information and Trade Secrets as may be further developed from time to time by the Company.
- C. The Company and its affiliates have established substantial goodwill and an excellent reputation with respect to the quality of its System, which goodwill and reputation have been and will continue to be of major benefit to the Company.
- D. Associate desires to become involved with the Company or a franchisee of the Company identified in Section 1(g) (a "Franchisee") in the capacity of an officer, partner, director, agent, manager, employee, Designated Business Manager or as a beneficial owner of the Franchise Business, or is an immediate family member of a principal owning an interest in the Franchise Business, and will become privileged as to certain Confidential Information and Trade Secrets. Associate may or may not have signed the Franchise Agreement or Guaranty and Assumption of Franchisee's Obligations form.
- E. Associate and the Company have reached an understanding with regard to nondisclosure by Associate of Confidential Information and Trade Secrets and with respect to noncompetition by Associate with the Company and other franchisees of the Company. Associate agrees to the terms of this Agreement as partial consideration for the Company's willingness to allow Associate to engage in a business relationship with Company or a Franchisee using the Company's Confidential Information and Trade Secrets.

NOW THEREFORE, in consideration of the above-noted Recitals, the mutual promises contained below and other good and valuable consideration, the receipt and sufficiency of which are acknowledged, Associate and the Company, intending legally to be bound, agree as follows:

1. <u>Definitions</u>.

- (a) "Affiliated Lessor" will mean any entity that controls, is controlled by, or is under common control with, Franchisee or any Franchisee's Owner(s) or in which Franchisee Owner(s) have an equity interest that leases premises to Franchisee or any person or entity for the operation of the All American Pet Resorts[®] Business.
- (b) "Associate" will mean the individual or entity described on page 1 of this Agreement and the Associate's managers, officers, beneficial owners, directors, employees, partners, members, principals and immediate family members. Associate shall also include Affiliated Lessor.
- (c) "Designated Business Manager" means an individual who will have day-to-day management of the Business after successful completion of our initial training, exercise on-site supervision, personally participate in the direct operation of the Business, attend required conferences and meetings; and has authority to make business decisions in Franchisee's absence, unavailability, incapacity or death.
- (d) "Competitive Business" as used in this Agreement means any business operating in competition with or similar to the Franchise Business; provided, however, Associate will not be prohibited from owning not more than a total of 5% of the stock of any company which is subject to the reporting requirements of the U.S. Securities and Exchange Act of 1934.
- (e) "Confidential Information" will mean without limitation, all knowledge, know-how, standards, formulas, methods and procedures related to the establishment and operation of the Franchise Business and includes all records pertaining to customers, suppliers, and other service providers of, and/or related in any way to, the Franchise Business including, without limitation, all databases (whether in print, electronic or other form), all names, addresses, phone numbers, e-mail addresses, customer purchase records, mail lists, manuals, promotional and marketing materials, marketing strategies and any other data and information which the Company or its affiliates designates as confidential including all information contained in the Company's Operations Manual, which may be provided as one or more separate manuals, written instructional guides, CD Rom, or other communications from the Company or its affiliates, which may be changed or supplemented from time to time.

(f)	"Franchise	Agreement"	will	mean	the	franchise	agreement	between
Company and	d			d	ated		<i>a</i>	as may be
amended or re	enewed from	time to time.						
(g)	"Franchisee	will mean						, who
has entered in	nto a Franchis	e Agreement.						

- (h) "Territory" will have the meaning defined in the Franchise Agreement.
- (i) "Term" will have the meaning defined in the Franchise Agreement.
- (j) "Trade Secret(s)" will mean information, including a formula, pattern, compilation, program, device, method, technique or process related to the Franchise Business that both derives independent economic value, actual or potential, from not being generally known to, and not being readily ascertainable by proper means by other persons who can obtain economic value from its disclosure or use and is the subject of efforts that are reasonable under the circumstances to maintain its secrecy.
- 2. <u>Confidential Information and Trade Secrets</u>. Associate and the Company acknowledge that the Confidential Information and Trade Secrets which are developed and utilized in connection with the operation of the Franchise Business is unique and the exclusive property of the Company or its affiliates. Associate acknowledges that any unauthorized disclosure or use of the Confidential Information and Trade Secrets would be wrongful and would cause irreparable injury and harm to the Company or its affiliates. Associate further acknowledges that the Company or its affiliates has expended a great amount of effort and money in obtaining and developing the Confidential Information and Trade Secrets, that the Company or its affiliates has taken numerous precautions to guard the secrecy of the Confidential Information and Trade Secrets, and that it would be very costly for competitors to acquire or duplicate the Confidential Information and Trade Secrets.
- 3. Nondisclosure of Confidential Information and Trade Secrets. During the Term and any renewal Term of the Franchise Agreement and for a period of three (3) years after the expiration or termination of the Franchise Agreement (unless such information is a Trade Secret in which case the requirements in this Section 3 will remain in place for as long as such information constitutes a Trade Secret), Associate must not at any time, publish, disclose, divulge or in any manner communicate to any person, firm, corporation, association, partnership or any other entity whatsoever or use, directly or indirectly, for its own benefit or for the benefit of any person, firm, corporation or other entity other than for the use of the Company or the Franchise Business, any of the Confidential Information or Trade Secrets of the Company or its affiliates.
- 4. Exceptions to Disclosing Confidential Information. Except as noted above, the restrictions on the disclosure and use of the Confidential Information will not apply to the following: (a) information that was in the public domain prior to being communicated to the Associate through no fault of the Associate; (b) information that entered the public domain after it was communicated to the Associate through no fault of the Associate; (c) information that was in the Associate's possession free of any obligation of confidence at the time it was communicated to the Associate; or (d) the disclosure of the Confidential Information in judicial or administrative proceedings to the extent that the Franchisee is legally compelled to disclose the information, if the Franchisee has notified the Franchisor before disclosure and used the Franchisee's best efforts, and afforded the Franchisor the opportunity, to obtain an appropriate protective order or other assurance satisfactory to the Franchisor of confidential treatment for the information required to be so disclosed.

- 5. <u>Non-competition Covenant</u>. Associate acknowledges that the Company must be protected against the potential for unfair competition by Associate's use of the Confidential Information and Trade Secrets in direct competition with the Company. Associate further acknowledges that the Confidential Information and Trade Secrets would not have been divulged to the Associate absent the Associate's agreement to strictly comply with the provisions of this Agreement. Associate therefore agrees that other than the Franchise Business licensed under the Franchise Agreement, Associate, will not during the Term and renewal Term of the Franchise Agreement:
 - (a) have any direct or indirect interest as a disclosed or beneficial owner in a Competitive Business;
 - (b) perform services as a manager, officer, beneficial owner, director, principal, employee, partner, member, consultant, representative, agent or otherwise for a Competitive Business; or
 - (c) divert or attempt to divert any business related to, or any customer or account of the Franchise Business, the Company's business, the business of any affiliate of the Company or any other franchisee's business, by direct inducement or otherwise, or divert or attempt to divert the employment of any employee of the Company or another franchisee licensed by Company, to any Competitive Business by any direct inducement or otherwise.
- 6. Post-Termination Covenant Not to Compete. Upon termination or expiration of the Franchise Agreement for any reason, Associate agrees that, for a period of three (3) years commencing on the effective date of termination or expiration of the Franchise Agreement, Associate will not have any direct or indirect interest (through any immediate family member of Associate or its beneficial owners or otherwise) as a disclosed or beneficial owner, investor, partner, director, officer, manager, employee, consultant, representative or agent or in any other capacity in any Competitive Business, located or operating: (a) in the Territory or any other franchisee's territory; (b) within 100 miles of the Territory or any other franchisee's Territory; or (3) within 100 miles of any Company or Company's affiliate owned Franchise Business.

The restrictions of this Section 6 will not be applicable to the ownership of shares of a class of securities listed on a stock exchange or traded on the over-the-counter market that represent 5% or less of the number of shares of that class of securities issued and outstanding.

The parties have attempted in this Agreement to limit the Associate's right to compete only to the extent necessary to protect the Company from unfair competition. The parties hereby expressly agree that if the scope of enforceability of the provision of Sections 5 and 6 are disputed at any time by the Franchisee, a court or arbitrator, as the case may be, may modify Sections 5 and 6 to the extent that it deems necessary to make such provisions enforceable under applicable law. THE ASSOCIATE EXPRESSLY ACKNOWLEDGES THAT THE ASSOCIATE POSSESSES SKILLS AND ABILITIES OF A GENERAL NATURE AND HAS OTHER OPPORTUNITIES TO EXPLOIT SUCH SKILLS. CONSEQUENTLY, ENFORCEMENT OF THE COVENANTS NOTED ABOVE WILL NOT DEPRIVE ASSOCIATE OF THE ABILITY TO EARN A LIVING.

- 7. <u>Injunction</u>. Associate hereby acknowledges and agrees that in the event of any breach or threatened breach of this Agreement, the Company will be authorized and entitled to seek, from any court of competent jurisdiction, preliminary and permanent injunctive relief in addition to any other rights or remedies to which the Company may be entitled. Associate agrees that the Company may obtain such injunctive relief, without posting a bond or bonds. Associate's sole remedy, in the event of the entry of such injunctive relief, will be dissolution of such injunctive relief, if warranted, upon a hearing duly had; provided, however, that all claims for damages by reason of the wrongful issuance of any such injunction are hereby expressly waived by Associate. In any litigation, arbitration or other proceeding concerning the entry of any requested injunction against Associate, Associate, for value, voluntarily waives such defenses as Associate might otherwise have under the law of the jurisdiction in which the matter is being litigated, arbitrated or otherwise relating to any claimed "prior breach" on the part of the Company; it being specifically understood and agreed between the parties that no action or lack of action on the part of the Company will entitle or permit the Associate to disclose any such Confidential Information and Trade Secrets in any circumstances.
- 8. <u>Effect of Waiver</u>. The waiver by Associate or the Company of a breach of any provision of this Agreement will not operate or be construed as a waiver of any subsequent breach thereof.
- 9. <u>Binding Effect</u>. This Agreement will be binding upon and inure to the benefit of Associate and the Company and their respective heirs, executors, representatives, successors and assigns.
- 10. <u>Entire Agreement</u>. This instrument contains the entire agreement of Associate and the Company relating to the matters noted in this Agreement. It may not be changed verbally, but only by an agreement in writing, signed by the party against whom enforcement of any waiver, change, modification, extension or discharge is sought.
- 11. <u>Governing Law</u>. This instrument will be governed by and construed under the laws of the State of Michigan.
- 12. <u>Jurisdiction and Venue</u>. In the event of a breach or threatened breach by Associate of this Agreement, Associate hereby irrevocably submits to the jurisdiction of the state and federal courts of Novi, Oakland County, Michigan, and irrevocably agrees that venue for any action or proceeding will be in the state and federal courts of the city of Novi and Oakland County, Michigan. Both parties waive any objection to the jurisdiction of these courts or to venue in the state and federal courts of the city of Novi and Oakland County, Michigan. In the event that the laws of the state where the Associate resides prohibit the aforesaid designation of jurisdiction and venue, then such other state's laws will control.
- 13. <u>Severability</u>. If any provision of this Agreement will be held, declared or pronounced void, voidable, invalid, unenforceable or inoperative for any reason, by any court of competent jurisdiction, government authority or otherwise, such holding, declaration or pronouncement will not affect adversely any other provisions of this Agreement which will otherwise remain in full force and effect.

14. Attorneys' Fees. In any action at law or in equity to enforce any of the provisions or rights under this Agreement, the unsuccessful party in such litigation, as determined by the court in a final judgment or decree, will pay the successful party or parties all costs, expenses and reasonable attorneys' fees incurred by such party or parties (including without limitation such costs, expenses and fees on any appeals), and if such successful party will recover judgment in any such action or proceeding, such costs, expenses and attorneys' fees will be included as part of such judgment.

SIGNATURES APPEAR ON THE FOLLOWING PAGE

IN WITNESS WHEREOF, the parties have signed this Agreement on the date first above written.

COMPANY:	ASSOCIATE:
ALL AMERICAN PET RESORTS, LLC	
By:	By:
Title:	Print Name:
Date:	Date:
AFFILIATED LESSOR:	ASSOCIATE:
By:	By:
Title:	Print Name:
Date:	Date:



EXHIBIT G-1 ALL AMERICAN PET RESORTS, LLC NON-DISCLOSURE AND NON-COMPETITION AGREEMENT

(Designated Business Manager)

EXHIBIT G-1

NON-DISCLOSURE AND NON-COMPETITION AGREEMENT (Designated Business Manager)

This Non-Disclosure and I	Non-Competition Agreement ("Agreement") is made and
entered into this day of	, 20 by and between All American Pet
Resorts, LLC, a Michigan limited	I liability company ("Company"), located at our principal
business address is 41850 West	Eleven Mile Road, Suite 202, Novi, MI, 48375 and
	_ ("Associate"), who resides or has a principal place of
address at	
	DECIMALO

RECITALS

- A. The Company is engaged in the business of selling franchises for the operation of a business offering professional pet boarding, daycare and grooming services, and the sale of related products ("Franchise Business"). The Franchise Business is operated under the Company's trademark "All American Pet Resorts®" and other service marks, trademarks, logo types, designs, and other commercial symbols (collectively "Marks").
- B. The Company has developed methods for establishing, operating and promoting Franchise Businesses under the Company's distinctive business format, plans, methods, data, processes, supply systems, marketing systems, formulas, techniques, designs, layouts, operating procedures, Marks and information and know-how of the Company ("Confidential Information" and "Trade Secrets") and such Confidential Information and Trade Secrets as may be further developed from time to time by the Company.
- C. The Company and its affiliates have established substantial goodwill and an excellent reputation with respect to the quality of its System, which goodwill and reputation have been and will continue to be of major benefit to the Company.
- D. Associate desires to become a Designated Business Manager of a franchisee of the Company identified in Section 1(g) (a "**Franchisee**") and will become privileged as to certain Confidential Information and Trade Secrets. Associate may or may not have signed the Franchise Agreement or Guaranty and Assumption of Franchisee's Obligations form.
- E. Associate and the Company have reached an understanding with regard to nondisclosure by Associate of Confidential Information and Trade Secrets and with respect to noncompetition by Associate with the Company and other franchisees of the Company. Associate agrees to the terms of this Agreement as partial consideration for the Company's willingness to allow Associate to engage in a business relationship with Company or a Franchisee using the Company's Confidential Information and Trade Secrets.

NOW THEREFORE, in consideration of the above-noted Recitals, the mutual promises contained below and other good and valuable consideration, the receipt and sufficiency of which are acknowledged, Associate and the Company, intending legally to be bound, agree as follows:

1. Definitions.

- (a) "Affiliated Lessor" will mean any entity that controls, is controlled by, or is under common control with, Franchisee or any Franchisee's Owner(s) or in which Franchisee Owner(s) have an equity interest that leases premises to Franchisee or any person or entity for the operation of the All American Pet Resorts[®] Business.
- (b) "Associate" will mean the individual or entity described on page 1 of this Agreement and the Associate's managers, officers, beneficial owners, directors, employees, partners, members, principals and immediate family members. Associate shall also include Affiliated Lessor.
- (c) "Designated Business Manager" means an individual who will have day-to-day management of the Business after successful completion of our initial training, exercise on-site supervision, personally participate in the direct operation of the Business, attend required conferences and meetings; and has authority to make business decisions in Franchisee's absence, unavailability, incapacity or death.
- (d) "Competitive Business" as used in this Agreement means any business operating in competition with or similar to the Franchise Business; provided, however, Associate will not be prohibited from owning not more than a total of 5% of the stock of any company which is subject to the reporting requirements of the U.S. Securities and Exchange Act of 1934.
- (e) "Confidential Information" will mean without limitation, all knowledge, know-how, standards, formulas, methods and procedures related to the establishment and operation of the Franchise Business and includes all records pertaining to customers, suppliers, and other service providers of, and/or related in any way to, the Franchise Business including, without limitation, all databases (whether in print, electronic or other form), all names, addresses, phone numbers, e-mail addresses, customer purchase records, mail lists, manuals, promotional and marketing materials, marketing strategies and any other data and information which the Company or its affiliates designates as confidential including all information contained in the Company's Operations Manual, which may be provided as one or more separate manuals, written instructional guides, CD Rom, or other communications from the Company or its affiliates, which may be changed or supplemented from time to time.

(f)	"Franchise	Agreement"	will	mean	the	franchise	agreement	between
Company and	1			d	ated		8	as may be
amended or re	enewed from	time to time.						-
(g)		e" will mean						, who
has entered in	ito a Franchis	e Agreement.						

- (h) "Territory" will have the meaning defined in the Franchise Agreement.
- (i) "Term" will have the meaning defined in the Franchise Agreement.

- (j) "Trade Secret(s)" will mean information, including a formula, pattern, compilation, program, device, method, technique or process related to the Franchise Business that both derives independent economic value, actual or potential, from not being generally known to, and not being readily ascertainable by proper means by other persons who can obtain economic value from its disclosure or use and is the subject of efforts that are reasonable under the circumstances to maintain its secrecy.
- 2. <u>Confidential Information and Trade Secrets</u>. Associate and the Company acknowledge that the Confidential Information and Trade Secrets which are developed and utilized in connection with the operation of the Franchise Business is unique and the exclusive property of the Company or its affiliates. Associate acknowledges that any unauthorized disclosure or use of the Confidential Information and Trade Secrets would be wrongful and would cause irreparable injury and harm to the Company or its affiliates. Associate further acknowledges that the Company or its affiliates has expended a great amount of effort and money in obtaining and developing the Confidential Information and Trade Secrets, that the Company or its affiliates has taken numerous precautions to guard the secrecy of the Confidential Information and Trade Secrets, and that it would be very costly for competitors to acquire or duplicate the Confidential Information and Trade Secrets.
- 3. <u>Nondisclosure of Confidential Information and Trade Secrets</u>. During the Term and any renewal Term of the Franchise Agreement and for a period of three (3) years after the expiration or termination of the Franchise Agreement (unless such information is a Trade Secret in which case the requirements in this Section 3 will remain in place for as long as such information constitutes a Trade Secret), Associate must not at any time, publish, disclose, divulge or in any manner communicate to any person, firm, corporation, association, partnership or any other entity whatsoever or use, directly or indirectly, for its own benefit or for the benefit of any person, firm, corporation or other entity other than for the use of the Company or the Franchise Business, any of the Confidential Information or Trade Secrets of the Company or its affiliates.
- 4. Exceptions to Disclosing Confidential Information. Except as noted above, the restrictions on the disclosure and use of the Confidential Information will not apply to the following: (a) information that was in the public domain prior to being communicated to the Associate through no fault of the Associate; (b) information that entered the public domain after it was communicated to the Associate through no fault of the Associate; (c) information that was in the Associate's possession free of any obligation of confidence at the time it was communicated to the Associate; or (d) the disclosure of the Confidential Information in judicial or administrative proceedings to the extent that the Franchisee is legally compelled to disclose the information, if the Franchisee has notified the Franchisor before disclosure and used the Franchisee's best efforts, and afforded the Franchisor the opportunity, to obtain an appropriate protective order or other assurance satisfactory to the Franchisor of confidential treatment for the information required to be so disclosed.
- 5. <u>Non-competition Covenant</u>. Associate acknowledges that the Company must be protected against the potential for unfair competition by Associate's use of the Confidential Information and Trade Secrets in direct competition with the Company. Associate further acknowledges that the Confidential Information and Trade Secrets would not have been divulged to the Associate absent the Associate's agreement to strictly comply with the provisions of this

Agreement. Associate therefore agrees that other than the Franchise Business licensed under the Franchise Agreement, Associate, will not during the Term and renewal Term of the Franchise Agreement:

- (a) have any direct or indirect interest as a disclosed or beneficial owner in a Competitive Business;
- (b) perform services as a manager, officer, beneficial owner, director, principal, employee, partner, member, consultant, representative, agent or otherwise for a Competitive Business; or
- (c) divert or attempt to divert any business related to, or any customer or account of the Franchise Business, the Company's business, the business of any affiliate of the Company or any other franchisee's business, by direct inducement or otherwise, or divert or attempt to divert the employment of any employee of the Company or another franchisee licensed by Company, to any Competitive Business by any direct inducement or otherwise.
- 6. Post-Termination Covenant Not to Compete. Upon termination or expiration of the Franchise Agreement for any reason, Associate agrees that, for a period of three (3) years commencing on the effective date of termination or expiration of the Franchise Agreement, Associate will not have any direct or indirect interest (through any immediate family member of Associate or its beneficial owners or otherwise) as a disclosed or beneficial owner, investor, partner, director, officer, manager, employee, consultant, representative or agent or in any other capacity in any Competitive Business, located or operating: (a) in the Territory or any other franchisee's territory; (b) within 100 miles of the Territory or any other franchisee's Territory; or (3) within 100 miles of any Company or Company's affiliate owned Franchise Business. This Section 6 in no way prohibits Associate from being employed by a Competitive Business so long as Associate observes its obligations under this Agreement regarding Confidential Information and Trade Secrets.

The restrictions of this Section 6 will not be applicable to the ownership of shares of a class of securities listed on a stock exchange or traded on the over-the-counter market that represent 5% or less of the number of shares of that class of securities issued and outstanding.

The parties have attempted in this Agreement to limit the Associate's right to compete only to the extent necessary to protect the Company from unfair competition. The parties hereby expressly agree that if the scope of enforceability of the provision of Sections 5 and 6 are disputed at any time by the Franchisee or Associate, a court or arbitrator, as the case may be, may modify Sections 5 and 6 to the extent that it deems necessary to make such provisions enforceable under applicable law. THE ASSOCIATE EXPRESSLY ACKNOWLEDGES THAT THE ASSOCIATE POSSESSES SKILLS AND ABILITIES OF A GENERAL NATURE AND HAS OTHER OPPORTUNITIES TO EXPLOIT SUCH SKILLS. CONSEQUENTLY, ENFORCEMENT OF THE COVENANTS NOTED ABOVE WILL NOT DEPRIVE ASSOCIATE OF THE ABILITY TO EARN A LIVING.

- 7. <u>Injunction</u>. Associate hereby acknowledges and agrees that in the event of any breach or threatened breach of this Agreement, the Company will be authorized and entitled to seek, from any court of competent jurisdiction, preliminary and permanent injunctive relief in addition to any other rights or remedies to which the Company may be entitled. Associate agrees that the Company may obtain such injunctive relief, without posting a bond or bonds. Associate's sole remedy, in the event of the entry of such injunctive relief, will be dissolution of such injunctive relief, if warranted, upon a hearing duly had; provided, however, that all claims for damages by reason of the wrongful issuance of any such injunction are hereby expressly waived by Associate. In any litigation, arbitration or other proceeding concerning the entry of any requested injunction against Associate, Associate, for value, voluntarily waives such defenses as Associate might otherwise have under the law of the jurisdiction in which the matter is being litigated, arbitrated or otherwise relating to any claimed "prior breach" on the part of the Company; it being specifically understood and agreed between the parties that no action or lack of action on the part of the Company will entitle or permit the Associate to disclose any such Confidential Information and Trade Secrets in any circumstances.
- 8. <u>Effect of Waiver</u>. The waiver by Associate or the Company of a breach of any provision of this Agreement will not operate or be construed as a waiver of any subsequent breach thereof.
- 9. <u>Binding Effect</u>. This Agreement will be binding upon and inure to the benefit of Associate and the Company and their respective heirs, executors, representatives, successors and assigns.
- 10. <u>Entire Agreement</u>. This instrument contains the entire agreement of Associate and the Company relating to the matters noted in this Agreement. It may not be changed verbally, but only by an agreement in writing, signed by the party against whom enforcement of any waiver, change, modification, extension or discharge is sought.
- 11. <u>Governing Law</u>. This instrument will be governed by and construed under the laws of the State of Michigan.
- 12. <u>Jurisdiction and Venue</u>. In the event of a breach or threatened breach by Associate of this Agreement, Associate hereby irrevocably submits to the jurisdiction of the state and federal courts of Novi, Oakland County, Michigan, and irrevocably agrees that venue for any action or proceeding will be in the state and federal courts of the city of Novi and Oakland County, Michigan. Both parties waive any objection to the jurisdiction of these courts or to venue in the state and federal courts of the city of Novi and Oakland County, Michigan. In the event that the laws of the state where the Associate resides prohibit the aforesaid designation of jurisdiction and venue, then such other state's laws will control.
- 13. <u>Severability</u>. If any provision of this Agreement will be held, declared or pronounced void, voidable, invalid, unenforceable or inoperative for any reason, by any court of competent jurisdiction, government authority or otherwise, such holding, declaration or pronouncement will not affect adversely any other provisions of this Agreement which will otherwise remain in full force and effect.

14. Attorneys' Fees. In any action at law or in equity to enforce any of the provisions or rights under this Agreement, the unsuccessful party in such litigation, as determined by the court in a final judgment or decree, will pay the successful party or parties all costs, expenses and reasonable attorneys' fees incurred by such party or parties (including without limitation such costs, expenses and fees on any appeals), and if such successful party will recover judgment in any such action or proceeding, such costs, expenses and attorneys' fees will be included as part of such judgment.

SIGNATURES APPEAR ON THE FOLLOWING PAGE

IN WITNESS WHEREOF, the parties have signed this Agreement on the date first above written.

COMPANY:	ASSOCIATE:
ALL AMERICAN PET RESORTS, LLC	
By:	By:
Title:	Print Name:
Date:	Date:
AFFILIATED LESSOR:	ASSOCIATE:
By:	By:
Title:	Print Name:
Date:	Date:



EXHIBIT H ALL AMERICAN PET RESORTS, LLC STATEMENT OF FRANCHISEE

EXHIBIT H

STATEMENT OF FRANCHISEE

[Note: Dates and Answers Must be Completed in the Prospective Franchisee's Own Handwriting]

In order to make sure that no misunderstanding exists between you, the Franchisee, and us, All American Pet Resorts, LLC (also called "All American Pet Resorts®", the "Franchisor" or "we") and to make sure that no violations of law might have occurred, and understanding that we are relying on the statements you make in this document, you assure us as follows:

	Date	Initials	
1.		, 20	The date on which I received a Franchise Disclosure Document regarding the All American Pet Resorts® Business.
2.		, 20	The date of my first face to face meeting with Marketing Representative to discuss a possible purchase of an All American Pet Resorts® Business.
3.		, 20	The date on which I received a completed copy (other than signatures) of the Franchise Agreement which I later signed.
4.		, 20	The date on which I signed the Franchise Agreement.
5.		, 20	The earliest date on which I delivered cash, check or other consideration to the Marketing Representative or an officer of Franchisor.
B.	Representations.		
or with location protection the	standings, "side agreem th me with respect to a on, operational, market ted territory or otherwi	ents," options, right- ny matter (including ting or administrativese), nor have I relied the Disclosure Docu	es, agreements, commitments, representations, of-first-refusal or otherwise have been made to but not limited to advertising, marketing, site re assistance, exclusive rights or exclusive or in any way on same, except as expressly noted ment or an attached written Addendum signed ollows:

(If none, you should write NONE in your own handwriting and initial.)

2. No oral, written, visual or other promises, agreements, commitments, representation, understandings, "side agreements" or otherwise which expanded upon or were inconsistent with the Disclosure Document or the Franchise Agreement or any attached written addendum signed by me and an officer of All American Pet Resorts[®], were made to me by any person or entity, nor have I relied in any way on same, except as follows:

(If none, you should write NONE in your own handwriting and initial.)

3. No oral, written, visual or other claim or representation (including but not limited to charts, tables, spreadsheets or mathematical calculations to demonstrate actual or possible results based on a combination of variables, such as multiples of price and quantity to reflect gross sales, or otherwise,) which stated or suggested a specific level or range of actual or potential sales, income, expenses, profits, cash flow, tax effects or otherwise (or from which such items might be ascertained) from the All American Pet Resorts[®] Business was made to me by any person or entity, nor have I relied in any way on same, except as follows:

(If none, you should write NONE in your own handwriting and initial.)

4. No contingency, prerequisite, reservation or otherwise exists with respect to any matter (including but not limited to my obtaining financing, or my fully performing any of my obligations), nor have I relied in any way on same, except as expressly noted in the Franchise Agreement or any attached written Addendum signed by All American Pet Resorts® and me: _____

- (If none, you should write NONE in your own handwriting and initial.)
- 5. The individuals signing for me constitute all of the executive officers, partners, shareholders, investors and/or principals. Each of such individuals has reviewed the Franchise Disclosure Document and all exhibits and carefully read, discussed, understands and agrees to the Franchise Agreement, each attached written Addendum and any personal guaranties.
- 6. I have had an opportunity to consult with an independent professional advisor, such as an attorney or accountant, prior to signing any binding documents or paying any sums, and All American Pet Resorts® has strongly recommended that I obtain such independent advice. I have also been strongly advised by All American Pet Resorts® to discuss my proposed purchase of All American Pet Resorts Business with any existing All American Pet Resorts® franchisees prior to signing any binding documents or paying any sums and All American Pet Resorts® has supplied me with a list of all existing franchisees if any exist.
- 7. I understand that a) entry into any business venture necessarily involves some unavoidable risk or loss or failure; b) while the purchase of a franchise may improve the chances for success, the purchase of an All American Pet Resorts[®] Business or any other franchise is a speculative investment; c) investment beyond that outlined in the Disclosure Document may be required to succeed; d) there exists no guaranty against possible loss or failure in this or any other business;

and e) the most important factors in the success of any All American Pet Resorts[®] Business, including the one to be operated by me, are my personal business skills, which include marketing, sales, and management, and require sound judgment and extremely hard work.

8. I understand that the occurrence of an economic crisis and financial situation, whether nationally, regionally, or locally, could have a negative impact on the pet care industry, the All American Pet Resorts® System and my All American Pet Resorts® Business and that the economic situation may worsen.

If there are any matters inconsistent with the statements in this document or if anyone has suggested that you sign this document without all of its statements being true, correct and complete, immediately inform All American Pet Resorts® (Phone: (248) 449-2949) and our president.

You understand and agree that we do not furnish, or authorize our salespersons, brokers or others to furnish any oral or written information concerning actual or potential sales, costs, income, expenses, profits, cash flow, tax effects or otherwise (or information from which such items might be ascertained), from franchise or non-franchised units, that no such results can be assured or estimated, and that actual results will vary from unit to unit.

You understand and agree to all provisions notes above and represent and warrant that all of the above statements are true, correct and complete.

PROSPECTIVE FRANCHISEE:	MARKETING REPRESENTATIVE:			
Date:	Date:			
REVIEWED BY FRANCHISOR:				
ALL AMERICAN PET RESORTS, LLC				
D.				
By:				
Its:				
Date:				



EXHIBIT I ALL AMERICAN PET RESORTS, LLC OPERATIONS MANUAL TABLE OF CONTENTS

Operations Manual

All American Pet Resorts Table of Contents

Section 1: Design Requirements	
1.1 Design Assistance	
1.1.1 Included Services	
1.1.2 Optional Services and Costs to Franchisee	.3
1.1.3 Kennels – Recommended	
1.1.4 Square Footage Allocation & Non-Revenue Space Design Recommendations	.4
Section 2: Business Management	
2.1 Brand Compliance	
2.2 Franchise Brand and Standards Review	.4
2.3 Revenue Reporting	.5
2.3.1 Required Reporting	.5
2.3.1.1 Monthly Requirements	.5
2.3.1.2 Semi-annual Requirements	.5
2.3.1.3 Franchisor – Operations Software Access	.5
2.3.1.4 Optional	
2.3.1.5 Royalty Payments	
2.3.1.6 Internal Dispute Resolution	.5
2.3.1.6.1 Type of Dispute	.5
2.4 Gingr Software	.6
2.4.1 Gingr Implementation	.6
2.4.2 Gingr Set-Up	.6
2.4.2.1 Gingr Payments – Credit Card Processing	.6
2.4.2.2 Naming Standards – Suites, Additional Services, and Grooming	.7
2.4.2.3 Service Pricing	.8
2.4.2.4 Maximizing Revenue Techniques	.9
2.4.2.4.1 Beginning Stage Techniques	.9
2.4.2.4.2 Established Resort Recommendations	.9
2.4.2.5 Gingr Training	
2.4.2.6 Gingr Reports	.10
2.4.2.7 Gingr Payment Monthly Statement	
2.4.2.8 Pet Suite Card	
2.4.2.9 Customer Spend Report	
2.4.2.10 Estimated Revenue Report	.20
2.4.2.11 Service by Date Report	
2.4.2.12 Birthday Report	
2.4.2.13 Custom Animal Icon Report	
2.4.3 Fiscal Planning	
2.4.3.1 Banking Relationship	
2.4.3.2 Budgeting and Cash Flow Management	
2.4.3.2.1 Management and Finance	
2.4.3.2.2 Approved Chart of Accounts	
2.4.3.3 Profit & Loss Statement (P&L)	
2.4.3.4 Balance Sheet	
2.4.3.5 Budget and Financial Planning	.26
2.4.3.6 Tax Planning	
2.4.3.7 Financing and Refinancing	.27
2.4.3.8 Payroll	
2.4.3.8.1 Payroll Taxes	
2.4.3.9 Sales Tax	
2.4.3.10 Regulations	
2.5 Business Management Task Lists	.28

	2.5.1 Daily	
	2.5.2 Weekly	
	2.5.3 Monthly	
	2.5.4 Quarterly	
	2.5.5 Semi-Annual	
	2.5.6 Annual	
	2.5.6.1 Insurance Cost Variables	
	2.5.6.2 Insurance Providers	
	2.5.6.3 All American Pet Resorts Requirements	
	2.5.7 Software Records Backup	
	6 Sample Franchise Brand and Standards Review	
	ection 3: Resort Administration	
	1 Staff Management	
	3.1.1 Recruitment	
	3.1.2 Scheduling	
	3.1.3 Staff Roles and Responsibilities	
	3.1.3.1 Resort Manager	./
	3.1.3.2 Assistant Manager	_
	3.1.3.2 Lead Supervisor	
	3.1.3.3 Kennel Crossover	
	3.1.3.4 Kennel Staff	
	3.1.3.5 Kennel Supervisor	
	3.1.3.6 Desk Manager	
	3.1.3.7 Night Staff	
	3.1.3.8 Facility Specialist	
	3.1.3.9 Assistant Manager	
	3.1.4 Salary Ranges	
	3.1.4.1 Bonuses: Holidays and Annual	
2	2 Basic Resort Setup	
	3.2.1 Setting Resort Policies	
	3.2.2 Hours of Operation	
	3.2.3 Client Contracts	
	3.2.3.1 Boarding/Daycare Terms and Conditions Contract	
	3.2.3.2 Group Play Terms and Conditions/Waiver	
3	3 Customer Relationships	
	3.3.1 Tours	
	3.3.1.1 Observation Area	
	3.3.1.2 Tour Process	
	3.3.2 Telephone Etiquette	
	3.3.2.1 Initial Information Call	
	3.3.2.2 Selling the VISIT	
	3.3.2.3 General Telephone Tips	
	3.3.2.4 On Hold	
	3.3.3 Securing Customer Data	.18
	3.3.4 Handling Customer Concerns with Pet Care	
	3.3.4.1 Common Complaints	.18
	3.3.4.1.1 Fees	
	3.3.4.1.2 Hours	.18
	3.3.4.1.3 Perceived Negligence	
	3.3.4.2 Handling Complaints	.19
	3.3.4.2.1 Handling Common Complaint Scenarios	.19
	4 Admitting Pets	.20
	3.4.1 Pre-Entry Evaluation	
	3.4.2 Required Vaccinations	
	3.4.2.1 Vaccinations are not Current	.21

3.4.3 Returning Pets	
3.4.3.1 Body Checks	
3.4.4 Pick-Up and Drop-Off	
3.4.5 Holiday Boarding/Deposits	
3.4.6 Bedding	
3.4.7 Toys	
3.4.8 Food and Treats from Home	
3.4.9 Medication	
3.4.10 Crates and Carriers	
3.4.11 Unaltered Pets	
3.4.11.1 Breeding Females	
3.4.11.2 Unaltered Males	
3.4.12 Aggressive Pets	
3.4.13 Group Play	
3.5 Add-On Services and Pricing	
3.5.2 Multiple Pet Discounts	
3.5.3 All-Inclusive Packages	
3.5.4 Training Programs	
3.5.4.1 Private Play	
3.5.4.2 Puppy Program/Puppy Care	
3.5.4.3 Geriatric Program/Geriatric Care	
3.5.5 Medications	
3.5.6 Bathing, Nail Trimming, and Anal Gland Expression	
3.5.7 Cancellation Policy	
3.5.8 Cat Boarding	
3.5.9 Retail	
3.6 Maintenance	
3.7 Communication with Clients	
3.7.1 Pet Injury/Pet Illness	
3.7.2 Pet Behavior	
3.7.2.1 Too Aggressive	
3.7.2.2 Pet Passes Evaluation but is not Suitable for Group Play	
3.7.2.3 Dog Fight/Injury	
3.7.3 Forms of Communication	
3.7.3.1 Written	
3.7.3.2 Oral	
3.7.3.3 Visual	
3.7.4 Long Stay Communication	
3.8 Job Descriptions	
3.8.1 Desk and Kennel – Sample Job Responsibilities	31
3.8.2 Resort Manager - Sample Job Responsibilities	31
3.8.3 Resort Assistant Manager - Sample Job Responsibilities	
3.8.4 Desk Manager - Sample Job Responsibilities	
3.8.5 Desk Supervisor - Sample Job Responsibilities	44
3.8.6 Kennel Supervisor - Sample Job Responsibilities	47
3.8.7 Night Shift Personnel - Sample Job Responsibilities	
3.8.8 Facility Specialist - Sample Job Responsibilities	
3.8.9 Kennel Personnel - Sample Job Responsibilities	
3.8.10 Groomer - Sample Job Responsibilities	
3.9 Sample Application	
3.10 Pet Incident Report	57
Section 4: Resort Operations	
4.1 Basic Pet Care	

4.1.1 Pre-Entry Evaluation	
4.1.2 Required Vaccination and Treatment Schedules	7
4.1.3 Suite Set Up	8
4.1.4 Feeding Protocol	
4.1.5 Standard Operating Procedures (SOP) Feeding	9
4.1.6 Large Dog Feeding Concerns	
4.1.7 Administering Medication	
4.1.7.1 Requirements for Administering Medication	
4.1.7.2 Techniques for Medication Administration	
4.1.7.2.1 Pills	
4.1.7.2.2 Eyedrops	
4.1.7.2.3 Eardrops	
4.1.7.2.4 Nose Drops	
4.1.7.2.5 Liquid Medicine	
4.1.7.2.6 Insulin	
4.1.7.3 Best Practice for Administering Medication	
4.1.8 Safely Moving Pets	
4.1.9 Walks, Group Play, and Daycare	
4.1.9.1 Managing Rough Behavior	
4.1.10 Capturing an Escapee	
4.1.11 Monitoring Pet Behavior	
4.1.12 Managing Difficult Pets	
4.1.12.1 Aggression	
4.1.12.2 Excessive Barking	
4.1.12.3 Shyness, Nervousness, or Stress	
4.1.13 Common Situations with Our Guests	
4.1.13.1 Dog is Not Eating	
4.1.13.2 Pets Preventing Others from Eating	
4.1.13.3 Diarrhea	
4.1.13.4 Diabetic Pets	
4.1.13.5 Obese Pets	
4.2 Kennel Operations	
4.2.1 Suite Cleaning	
4.2.1.1 Approved Cleaning Products	
4.2.2 Daily Cleaning	
4.2.3 Cleaning Upon Pet Departure	
4.2.3.1 Tools Needed	
4.2.3.1 Procedure to Properly Clean Suite	
4.2.4 Grooming Room Cleaning	
4.2.4.1 After Grooming	
4.2.4.2 Nightly Cleaning	
4.2.4.3 Weekly Cleaning	
4.2.5 Common Areas and Lobby	
4.2.5.1 Daily	
4.2.5.2 Weekly	
4.2.5.3 Quarterly	
4.2.6 Outdoor Play Areas	
4.2.6.1 Daily	25
4.2.6.2 Nightly	
4.2.7 Dish Washing Protocol	
4.2.8 Customer Restrooms	
4.2.8.1 Daily	
4.2.8.2 Weekly	
4.2.11 Water Features – Pools and Fountains	
4.2.11.1 Daily	
4.2.12 Resort Equipment	27

4.2.12.1 Indoor/Outdoor Play Equipment	
4.2.12.2 Washer and Dryer	27
4.2.12.3 Poop Scoopers	28
4.2.12.4 Poop Dump	
4.2.12.5 Foamers	
4.2.13 Pet Food	
4.2.14 Resort Safety and Emergency Procedures	
4.2.14.1 Fire Prevention and Evacuation	
4.2.14.2 Occupational Safety and Health Administration	
4.2.14.3 First Aid Guidelines	
4.2.14.4 Material Safety Data Sheets	
4.2.14.5 Staff Safety Training	
4.3 Daycare/Group Play Setup and Operating Procedures	
4.3.3 Commonly Asked Questions	
4.4 Injury and Illness	3±
4.4.1 First Aid	
4.4.1.1 Bite or Puncture Wound	
4.4.1.2 Injury Communication	
4.4.1.3 Symptoms to Watch	
4.4.2 Communicating with a Veterinarian	50 77
4.4.3 Common Illnesses	
4.4.3.1 Digestive Problems	
4.4.3.2 Constipation	
4.4.3.3 Lameness	
4.4.3.4 Lacerations.	
4.4.3.5 Skin Problems	
4.4.3.5.1 Hot Spots or Lick Sores	
4.4.3.5.2 Pressure Sores	
4.4.3.6 Bladder or Urinary Tract Infections	
4.4.3.7 Coughing or Bordetella (Canine Cough)	
4.4.3.7.1 Caring for Pet with Canine Cough	∓⊥ 11
4.4.3.7.2 Educating Customers	
4.4.4 Less Common Illnesses	
4.4.4.1 Gastric Dilation and Volvulus (Bloat, GDV)	
4.4.4.2 Canine Influenza (H3N8 and H3N2)	
4.4.4.2.1 Pet Care	
4.4.4.2.2 Resort Care	
4.4.4.3 Hypothermia and Frostbite	_
4.4.4.4 Heat Stroke	
4.4.4.5 Parasites (Worms)	
4.4.4.5.1 Roundworms	
4.4.5 Temperature and Fever	
4.4.6 Choking	
4.4.6.1 Pet Care	
4.4.7 Bleeding	
4.4.8 Allergic Reaction	
4.4.9 Rescue Breathing	
4.4.10 Puppy Warts	
4.4.11 Handling the Death of a Pet While in Resort	
4.5 Advanced Pet Care	
4.5.1 Introduction	
4.5.2 Puppy Care	
4.5.2.1 Feeding	
4.5.2.2 Walks and Group Play	
4.5.3 Geriatric Dogs	49

4.5.3.1 Geriatric Care	
4.5.3.2 Group Play	
4.5.3.3 Marketing	
4.5.4 Diabetic Dogs	
4.5.4.2 Resort Care	
4.6 Grooming	
4.6.1 Spa Services	
4.6.2 Trimming Nails	
4.6.2.1 Equipment	
4.6.2.2 Technique	
4.6.2.3 First Aid	
4.6.3 Brushing Teeth	
4.6.4 Cleaning Ears4.6.5 Expressing Anal Glands Externally	
4.6.6 Bathing	
4.6.7 Drying	
4.7 Cats	
4.7.1 Cattery Requirements	
4.7.2 Admitting Cats	
4.7.3 Condominium Cleaning	
4.7.4 Condominiums	
4.7.5 Security	
4.7.6 Food and Feeding Procedures	
4.7.7 Medication 4.7.7.1 Drops	
4.7.7.2 Pills	
4.7.7.3 Liquids	
4.7.8 Standard Operating Procedures	
4.7.9 Monitoring Behavior	
4.7.9.1 Spraying	
4.7.9.2 Food and Water Intake	
4.7.9.3 Litterbox	
4.7.9.4 Cat Flu	
4.7.9.6 Bathing Cats	
4.7.9.7 Revenue-generating Ideas for Cats	
4.7.9.8 Handling Death of a Cat	
4.8 Best Practices to Incorporate into Your Standard Operating Procedures	
4.8.1 Best Practice #1 Standardize the Check-In Process	
4.8.2 Best Practice #2 Pet Handling	
4.8.3 Best Practice #3 Standardize the Departure Process	
4.8.4 Best Practice #4 Boarding and Daycare	
4.8.5 Best Practice #5 Lobby Area Procedures	
4.8.6.1 Tips for Managing Your Staff	
Section 5: Marketing	
5.1 All American Pet Resorts Difference	
5.2 Marketing	
5.2.1 What is Marketing?	
5.3 Marketing Goals	.5
5.3.1 Adding Your Marketing Goals to Your Marketing Plan	
5.4 Customer Acquisition	
5.5 Customer Conversion	
5.6 Customer Retention	
5.7 Where Do You Start?	.ŏ

5.7.1 Marketing Plan	9
5.7.1.1 Tracking	
5.7.2 Advertising	10
5.7.2.1 Print Advertising	10
5.7.2.2 Digital Advertising	10
5.7.3 National Marketing	
5.7.4 Digital Advertising/Social Media Guidelines	11
5.7.4.1 What is Social Media?	12
5.7.4.2 Identifying Your Relationship with All American Pet Resorts	12
5.7.4.3 Best Practices	
5.7.4.4 Responding in Social Media	
5.7.4.5 Unacceptable Uses	
5.7.4.6 Digital Marketing Channels	
5.7.4.7 Using Social Media in Ongoing Local Marketing	
5.7.4.8 Social Media Boosting (Instagram, Facebook)	
5.7.4.9 Engagement	
5.7.4.10 Strategies During Low Occupancy	
5.7.5 Local Marketing	
5.7.5.1 Identify Your Target Market	
5.7.5.2 Competitive Market Analysis	
5.7.6 Required Launch Program for New Pet Resort Grand Opening	
5.7.6.1 Grand Opening: Announcing You are Open for Business	18
5.7.6.2 Brand Standards	
5.7.6.3 Planning	
5.7.6.4 Your Grand Opening Timeline	
5.7.6.5 Strategy for Launch Program	
5.7.6.6 Grand Opening Open House	
5.8 Mascot	
5.8.1 General Instructions	
5.8.2 Mascot Guidelines for Performers/Employees	
5.8.2.1 Dressing the Performer	
5.8.2.2 Performance Tips	
5.8.2.3 Emotions	
5.8.3 Gestures/Waves	
5.8.4 Walking Around	
5.8.4.1 Working with Kids	
5.8.5 Mascot Commandments	
5.8.6 Minor Crisis Situations	
5.8.7 Be Prepared for the Day	
5.8.8 Storing the Costume	
5.8.9 Mascot Care and Cleaning	
5.9 Ongoing Local Marketing	
5.9.1 Build Relationships with Local Veterinarians	
5.9.2 Using Your Customer Relationship Management (CRM) System	
5.9.2.1 Key Times for Boarding	
5.9.3 Direct Mail	
5.9.4 Flags and Signage	
5.9.5 Use of the All American Pet Resorts Trademarks	
5.9.5.1 Additional Marketing Ideas	
5.9.6 Promotional Materials	
5.9.7 Nurturing Customers	
5.10 Marketing Terms	
Section 6: Approved Suppliers and Equipment	
6.1 Recommended Supplies	
6.1.1 Grooming	
6.1.2 Recommended Facility Supplies	4

6.1.3 Recommended General Resort Supplies	8
6.2 List of Equipment Suppliers	9



EXHIBIT J ALL AMERICAN PET RESORTS, LLC APPLICATION AND CONFIDENTIALITY AGREEMENT

EXHIBIT J

ALL AMERICAN PET RESORTS, LLC

APPLICATION AND CONFIDENTIALITY AGREEMENT ("APPLICATION")

This Application is entered into this	day of	, 20, by and	between
ALL AMERICAN PET RESORTS, LLC ("We	" or " Us "), and _	("You" or
"Your"). You hereby acknowledge and agree a	s follows:		

- 1. You have made application or expressed a desire to purchase an All American Pet Resorts® Franchise ("Franchise") and you have had an opportunity to review our disclosure documentation that we provided to you more than 10 business days prior to the date of this Application ("Disclosure Documentation"), indicating your *bona fide* intent to enter into a franchise agreement ("Franchise Agreement") with us on such terms and conditions as are generally contained in our standard form of Franchise Agreement.
- 2. If you wish to cancel this Application, you must return all materials provided by us, and sign a release in a form reasonably prescribed by us.
- 3. You understand that, prior to executing the Franchise Agreement, we or our subsidiaries or affiliates or our franchisees ("Franchisees") may disclose or otherwise impart to you information and material pertaining to us, our methods of operation, techniques, know-how, promotion, and publicity, including information which we consider to be a trade secret (collectively the "System Information"). You acknowledge that the System Information is of a proprietary and confidential nature constituting our trade secrets and having immeasurable value. Accordingly, in recognition of the provisions noted above and in consideration thereof, you agree to hold in confidence and keep secret the System Information that is disclosed or made known to you in the course of your discussions with us, the Franchisees, and our subsidiaries and affiliates, and you will not impart or make known any of the same or anything relating to the same to any person, firm, or corporation unless we authorize you to do so in writing or unless otherwise required by law.
- 4. You agree that you will retain and receive all the System Information for use only in connection with the business to be operated under the All American Pet Resorts[®] Franchise applied for and will not use, directly or indirectly, the System Information for any other purpose without our prior written consent.
- 5. Upon termination of this Application for any reason whatsoever, you will promptly return the System Information to us, together with any copies thereof. Upon our request, you must sign a document in the form reasonably prescribed by us stating that all of the System Information has been returned and that any further System Information in your possession will be returned if discovered.
- 6. You acknowledge that consumer credit reports containing personal information are being or may be referred to in connection with this Application for an All American Pet Resorts[®] Franchise and you hereby consent to the disclosure of credit or personal information to any credit

reporting agency or to any person with whom you have or propose to have financial relations. You agree to provide us with a copy of your social security number, or if you are a business entity, your employer identification number, or proof of citizenship, in order to enable us to verify your registered name for the purposes of properly conducting any such inquiries.

If you are a business entity, you agree to provide us with a copy of the articles of

Dated at	, this day of	, 20
WITNESS:		
	Name of Applicant	
	By:	
	Its:	
All American Pet Resor		
o consider you as a candidate f	ts, LLC hereby acknowledges receipt of an All American Pet Resorts® Busin	of the application and agreness.
o consider you as a candidate f	ts, LLC hereby acknowledges receipt of	of the application and agreeness
o consider you as a candidate f	ts, LLC hereby acknowledges receipt of for an All American Pet Resorts® Busing day of, this day of	of the application and agreeness



EXHIBIT K ALL AMERICAN PET RESORTS, LLC FORM GENERAL RELEASE OF CLAIMS

RELEASE OF CLAIMS

Resorts,	nd in consider LLC (" Fran laims (" Agree	chisoı	r") and					,	
			RE	CITALS					
A.	Franchisor	and	Franchisee	entered	into	a	Franchise	Agreement	dated

- B. [NOTE: Describe the circumstances relating to the release.]
- C. Subject to and as addressed with greater specificity in the terms and conditions set forth below, Franchisor and Franchisee now desire to settle any and all disputes that may exist between them relating to the Franchise Agreement.

AGREEMENTS

- 1. Consideration. [NOTE: Describe the consideration paid.]
- 2-3. [NOTE: Detail other terms and conditions of the release.]
- Release of Claims by Franchisor. [NOTE: Only included in the event of a termination or transfer.] Except as provided below and in consideration of, and only upon full _ to Franchisor, and the other terms and conditions of this Agreement, the payment of \$ receipt and sufficiency of which is hereby acknowledged, Franchisor, for itself and for each of its affiliated corporations, subsidiaries, divisions, insurers, indemnitors, attorneys, successors, and assigns, together with all of its past and present directors, officers, employees, attorneys, agents, assigns and representatives does hereby release and forever discharge Franchisee and each of his heirs, executors, successors, and assigns of and from any and all actions, suits, proceedings, claims (including, but not limited to, claims for attorney's fees), complaints, judgments, executions, whether liquidated or unliquidated, known or unknown, asserted or unasserted, absolute or contingent, accrued or not accrued, disclosed or undisclosed, related to the Franchise Agreement. This release does not release Franchisee for any post-termination obligations, including any indemnification, confidentiality or covenant not to compete obligations contained in the Franchise Agreement. Further, this release does not release Franchisee from any obligations he may have under this Agreement.
- 5. Release of Claims by Franchisee. In consideration of the other terms and conditions of this Agreement, the receipt and sufficiency of which is hereby acknowledged, Franchisee, for himself and for each of his heirs, executors, administrators, insurers, attorneys, agents, representatives, successors, and assigns, does hereby release and forever discharge Franchisor and each of its respective affiliated corporations, subsidiaries, divisions, insurers, indemnitors, attorneys, successors, and assigns, together with all of their past and present directors, officers, employees, attorneys, agents, assigns and representatives in their capacities as such, of and from any and all actions, suits, proceedings, claims (including, but not limited to, claims for attorney's

fees), complaints, charges, judgments, executions, whether liquidated or unliquidated, known or unknown, asserted or unasserted, absolute or contingent, accrued or not accrued, related to the Franchise Agreement.

- 6. Reservation of Claims Against Non-Settling Parties. Franchisor and Franchisee expressly reserve their right and claims against any non-settling persons, firms, corporations, or other entities for whatever portion or percentage their damages are found to be attributable to the wrongful conduct of said non-settling parties.
- 7. Entire Agreement. This Agreement constitutes the entire agreement between the parties relative to the subject matter contained herein, and all prior understandings, representations and agreements made by and between the parties relative to the contents contained in this Agreement are merged into this Agreement.
- <u>Voluntary Nature of Agreement</u>. The parties acknowledge and agree that they have entered into this Agreement voluntarily and without any coercion. The parties further represent that they have had the opportunity to consult with an attorney of their own choice, that they have read the terms of this Agreement, and that they fully understand and voluntarily accept the terms.
- Governing Law and Jurisdiction. This Agreement will be construed and enforced in accordance with the law of the state of _____.
- Attorneys' Fees. All rights and remedies under this Agreement will be cumulative and none will exclude any other right or remedy allowed by law. In the event of a breach of this Agreement that requires one of the parties to enforce the terms and conditions of this Agreement, the non-prevailing party will pay the prevailing party's attorneys' fees and costs incurred by reason of the breach.

Dated:	, 20	ALL AMERICAN PET RESORTS, LLC
		By
		Its
Dated:	, 20	FRANCHISEE:
		By

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

State	Effective Date
Florida	July 10, 2023
Indiana	May 15, 2023
Michigan	August 11, 2023
North Carolina	July 6, 2021*
Texas	May 15, 2006*
Virginia	May 24, 2023
New York	Pending

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

*Denotes One-Time Filing, perpetual registration



EXHIBIT L ALL AMERICAN PET RESORTS, LLC RECEIPT

RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If All American Pet Resorts, LLC offers you a franchise, All American Pet Resorts, LLC must provide this disclosure document to you 14 calendar days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale or grant.

Iowa, New York, Oklahoma and Rhode Island require that All American Pet Resorts, LLC gives you this disclosure document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship.

Michigan and Oregon require that All American Pet Resorts, LLC gives you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If All American Pet Resorts, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580, and the appropriate state agency identified on Exhibit E.

The franchisor is All American Pet Resorts, LLC located at 41850 West Eleven Mile Road, Suite 202, Novi, MI, 48375. Its telephone number is (248) 513-3006.

Issuance Date: April 30, 2024

The name, principal business address and telephone number of each franchise seller offering the franchise:

Name	Address	Phone Number		
Arthur Rimbold, 418 48375	350 West E	Eleven Mile Road,	Suite 202, Novi, MI,	(248) 513-3006
Stephan Dimitroff, 4 48375	1850 West	Eleven Mile Road,	Suite 202, Novi, MI,	(248) 513-3006

All American Pet Resorts® authorizes the respective state agencies identified on Exhibit E to receive service of process for it in the particular state.

I received a disclosure document dated April 30, 2024 that included the following Exhibits: (A) Financial Statements; (B) Franchise Agreement; (C) List of Franchisees; (D) Trademark-Specific Franchisee Organizations; (E) List of State Administrators and Agents for Service of Process; (F) State-Specific Addenda to Disclosure Document and Franchise Agreement; (G) Nondisclosure and Non-Competition Agreement; (H) Statement of Franchisee; (I) Operations Manual Table of Contents; (J) Application and Confidentiality Agreement; (K) Form General Release of Claims; and (L) Receipt.

Date	Signature	Printed Name
Date	Signature	Printed Name
	Keep this copy for y	our records.

RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If All American Pet Resorts, LLC offers you a franchise, All American Pet Resorts, LLC must provide this disclosure document to you 14 calendar days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale or grant.

Iowa, New York, Oklahoma and Rhode Island require that All American Pet Resorts, LLC gives you this disclosure document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship.

Michigan and Oregon require that All American Pet Resorts, LLC gives you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If All American Pet Resorts, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580, and the appropriate state agency identified on Exhibit E.

The franchisor is All American Pet Resorts, LLC located at 41850 West Eleven Mile Road, Suite 202, Novi, MI 48375. Its telephone number is (248) 513-3006.

Issuance Date: April 30, 2024

The name, principal business address and telephone number of each franchise seller offering the franchise:

_	Name	Address	Phone Number		
	.00,0	350 West E	Eleven Mile Road,	Suite 202, Novi, MI,	(248) 513-3006
	Stephan Dimitroff, 4 48375	1850 West	Eleven Mile Road,	, Suite 202, Novi, MI,	(248) 513-3006

All American Pet Resorts[®] authorizes the respective state agencies identified on Exhibit E to receive service of process for it in the particular state.

I received a disclosure document dated April 30, 2024 that included the following Exhibits: (A) Financial Statements; (B) Franchise Agreement; (C) List of Franchisees; (D) Trademark-Specific Franchisee Organizations; (E) List of State Administrators and Agents for Service of Process; (F) State-Specific Addenda to Disclosure Document and Franchise Agreement; (G) Nondisclosure and Non-Competition Agreement; (H) Statement of Franchisee; (I) Operations Manual Table of Contents; (J) Application and Confidentiality Agreement; (K) Form General Release of Claims; and (L) Receipt.

Date	Signature	Printed Name
AAPR FDD 2024	L-3	

Date Signature Printed Name

Please sign this copy of the receipt, date your signature, and return it to our Chief Operating Officer, Stephan Dimitroff, at our principal business address is 41850 West Eleven Mile Road, Suite 202, Novi, MI 48375.